AudioCodes One Voice Operations Center

EMS, SEM and IP Phone Manager

Version 7.2





Та	hl	<u> </u>	f C	Or	Ite	nts
I a				0	π	1113

1	Over	view	17
Pr	e-insta	allation Information	19
2	Mana	aged VoIP Equipment	21
3	Hard	ware and Software Specifications	23
	3.1	EMS Server and Client Requirements	23
	3.2	Bandwidth Requirements	24
		3.2.1 EMS Bandwidth Requirements	24
	2.2	3.2.2 SEM Bandwidth Requirements	24
	3.3	Performance and Data Storage	26
	3.4	Microsoft Lync Monitoring SQL Server Prerequisites	
4	EMS	Software Deliverables	29
	4.1	Dedicated Hardware Installation – DVDs 1-4	29
	4.2	VMware – DVD 5	30
	4.3	Hyper-V – DVD 5	30
EN	IS Ser	ver Installation	31
5	Test	ng Installation Requirements -Dedicated Hardware	33
	51	Hardware Requirements	33
	0.1	5.1.1 Testing Hardware Requirements on the Linux Platform	33
6	Insta	Iling the EMS Server on Dedicated Hardware	35
	61	ISO Files Verification	35
	0.1	6.1.1 Windows	
		6.1.2 Linux	36
	6.2	Installing the EMS Server on the Linux Platform	37
		6.2.1 DVD1: Linux CentOS 5.9.	37
		6.2.2 DVD2: Oracle DB Installation 6.2.3 DVD3: FMS Server Application Installation	40
	6.3	EMS Server Users	45
7	Insta	Iling the EMS on Virtual Server Platform	47
	7.1	Installing the EMS Server on the VMware Platform	47
		7.1.1 Installing the VMware Virtual Machine	47
		7.1.2 Configuring the Virtual Machine Hardware Settings	54
		 7.1.3 Connecting EMS Server to Network	56
		7.1.4.1 Site Requirements	57
		7.1.4.2 Cluster Host Node Failure	60
	7.2	Installing the EMS Server on Microsoft Hyper-V Platform	60
		7.2.1 Installing the Microsoft Hyper-V Virtual Machine7.2.2 Configuring the Virtual Machine Hardware Settings	61 65

	7.2.2.1 Expanding Disk Capacity	67
	7.2.3 Changing MAC Addresses from Dynamic to Static	71
	7.2.5 Configuring EMS Virtual Machines in a Microsoft Hyper-V Cluster	73
	7.2.5.1 Site Requirements	73
	7.2.5.2 Add the EMS VM in Failover Cluster Manager	74
	7.2.5.3 Cluster Host Node Failure	
EN	IS Server Upgrade	79
8	Upgrading the EMS Server on Dedicated Hardware	81
	8.1 Upgrading the EMS Server-DVD	81
	8.2 Upgrading the EMS Server-ISO File	84
9	Upgrading the EMS Server on the VMware Platform	85
	9.1 Step 1: Setting Up the VMware vSphere Web Client	85
	9.1.1 Setting up Upgrade Using VMware Remote Console Application (VN	IRC)88
	9.1.2 Setting up Upgrade Using VMware Server Host	90
	9.2 Step 2: Running the VMware Upgrade	91
FN	AS Server Machine Backup and Restore	95
4.0		07
10	EMS Server Backup	97
11	EMS Server Restore	99
FN	AS Server Manager	101
12	Getting Started	103
		400
	12.1.1 Connecting to the EMS Server Manager	103 107
13	Viewing Process Statuses	109
14	Viewing General Information	111
15	Collecting Logs	113
16		
	Application Maintenance	115
	Application Maintenance 16.1 Start /Stop the Application	115 115
	Application Maintenance	115 115 116
	Application Maintenance 16.1 Start /Stop the Application 16.2 Web Servers 16.2.1 Apache and Tomcat Server Processes	115 115 116 116
	Application Maintenance 16.1 Start /Stop the Application 16.2 Web Servers 16.2.1 Apache and Tomcat Server Processes 16.2.2 HTTP/HTTPS Services	115 115 116 116 117
	Application Maintenance 16.1 Start /Stop the Application. 16.2 Web Servers 16.2.1 Apache and Tomcat Server Processes	115 115 116 116 117 118
	Application Maintenance 16.1 Start /Stop the Application. 16.2 Web Servers. 16.2.1 Apache and Tomcat Server Processes 16.2.2 HTTP/HTTPS Services. 16.2.2.1 JAWS IP Configuration 16.3 Change Schedule Backup Time	115 115 116 116 117 118 118
	Application Maintenance 16.1 Start /Stop the Application. 16.2 Web Servers 16.2.1 Apache and Tomcat Server Processes 16.2.2 HTTP/HTTPS Services. 16.2.2.1 JAWS IP Configuration 16.3 Change Schedule Backup Time 16.4 Restore	115 115 116 116 117 118 118 118 118
	Application Maintenance 16.1 Start /Stop the Application. 16.2 Web Servers. 16.2.1 Apache and Tomcat Server Processes 16.2.2 HTTP/HTTPS Services. 16.2.2.1 JAWS IP Configuration 16.3 Change Schedule Backup Time 16.4 Restore 16.5 License	115 115 116 116 117 118 118 118 118 119

	16.6	Shutdo	own the EMS Server Machine	121
	16.7	Reboo	t the EMS Server Machine	121
17	Netw	ork Cor	nfiguration	123
	47.4	0		400
	17.1	Server	IP Address	123
	17.2	Ethern	et Interfaces	125
		17.2.1	EMS Client Login on all EMS Server Network Interfaces	125
		17.2.2	Add Interface	126
		17.2.3	Remove Interface	127
	47.0	17.2.4	Modify Interface	127
	17.3	Etnern	et Redundancy	128
		17.3.1	Add Redundant Interface	129
		17.3.2	Remove Ethernet Redundancy	131
	47.4	DNC C		132
	17.4		nem	133
	17.5	NAT	,	134
	17.6	Static	Routes	134
	17.7	SNMP	Agent	135
	17.8	Server	SNMPv3 Engine ID	135
18	Date	and Tin	ne Settings	137
	10 1	NTD		127
	10.1	NIF	Stanning and Starting the NTD Sanjar	120
		10.1.1	Restrict Access to NTP Clients	139
	10.2	Timoza	ano Sottingo	120
	10.2		nd Time	139
	10.3	Date a	na Time	140
19	Secu	rity		141
	19.1	EMS U	ser	142
	19.2	SSH		143
		19.2.1	SSH Log Level	143
		19.2.2	SSH Banner	144
		19.2.3	SSH on Ethernet Interfaces	145
			19.2.3.1 Add SSH to All Ethernet Interfaces	.145
			19.2.3.2 Add SSH to Ethernet Interface	.146
		1924	Fnable/Disable SSH Password Authentication	. 140 1 4 7
		19.2.5	Enable SSH IgnoreUserKnownHosts Parameter	147
		19.2.6	SSH Allowed Hosts	148
			19.2.6.1 Allow ALL Hosts	.148
			19.2.6.2 Deny ALL Hosts	.149
			19.2.6.3 Add Hosts to Allowed Hosts	.149
	10 2		seword	151
	19.5		ers Passwords	152
	10.4		Conoral Password Sattings	152
		19.4.1	Operating System Users Security Extensions	153
		10.1.2		.00

	19.5	File Integrity Checker	155
	19.6	Software Integrity Checker (AIDE) and Pre-linking	155
	19.7	USB Storage	156
	19.8	Network Options	156
	19.9	Auditd Options	157
	19.1	0HTTPS/SSL/TLS Security	158
		19.10.1 HTTPS Authentication	159
		19.10.2 Enable SEM Client Secured Connection	160
		19.10.3 Enable IP Phone Manager Client JAWS and NBIF Secured	160
		19 10 4 Server Certificates Update	160
		19.10.5 SEM - AudioCodes Devices Communication	168
20	Diadu	nostics	169
	200 4		400
	20.1	Server Syslog Configuration	169
	20.2	Devices Systog Configuration	171
	20.3	Devices Debug Configuration	172
НА	(Hiah	Availability)	173
24	Cotti	ng Storted with HA (High Avgilghility)	475
21	Getti	ng Started with HA (High Availability)	175
	21.1	EMS HA Pre-requisites	176
	21.2	EMS HA Data Synchronization	177
		21.2.1 Replicate EMS Server Manager Actions	177
	21.3	EMS Server Manager	177
	21.4	EMS Client	178
	21.5	EMS Server Upgrade	178
		21.5.1 Upgrading to Version 7.2.1000	178
	21.6	EMS Server Restore	180
22	EMS	HA Configuration	181
	22.1	Primary Server HA Installation in Global IP Model	182
	22.2	Primary Server HA Installation in Geo HA Model	184
		22.2.1 Ping Nodes	186
	22.3	Secondary Server HA Installation	187
	22.4	HA Status	188
		22.4.1 Advanced Status View	190
	22.5	EMS Server Manual Switchover	191
	22.6	EMS HA Uninstall	1 92
Co	nfigur	ing the Firewall and Installing the EMS Client	195

23	Conf	iguring	g the Firewall	197
24	Insta	lling th	e EMS Client	207
	24.1	Runni	ing the EMS Client on a PC or Laptop	
	24.2	Initial	Login	
	24.3	Instal	ling and Running the EMS Client on a PC using Ja	va Web Start
		(JAW	S)	
Ар	pendi	x		213
Α	Site I	Prepara	ation	215
В	EMS	Applic	ation Acceptance Tests	217
	B.1	Introd	luction	
	B.2	Confi	guration	
		B.2.1	Client Installation	217
		B.2.2	Server Installation	
		B.2.3	Add Auxiliary File	
		B.2.4	Add Media Gateway	218
		B.2.5	Entity Profile – Digital CPE Devices	219
		B.2.6	Entity Profile – Analog CPE Devices	220
	B.3	Faults	5	221
		B.3.1	Alarm Receiver	221
		B.3.2	Delete Alarms	221
		B.3.3	Acknowledge Alarm	221
		B.3.4	Forwarding Alarms	222
	B.4	Secur	'ity	223
		B.4.1	Users List	223
		B.4.2	Non Repetitive Passwords	223
		B.4.3	Removing Operator	224
		B.4.4	Journal Activity	224
	B.5	Utilitie	es	225
		B.5.1	MG Search	
		B.5.2	Online Help	226
		B.5.3	Backup and Recovery	226

AudioCodes

С	Conf Ser	iguring vers	RAID-0 for AudioC	odes EMS on HP ProLiant DL360p G	en8 227
	C.1	Prere	uisites		
	C.2	Hardy	are Preparation		227
	C 3	Confi	uring RAID-0		228
D	Mana	aging C	usters		
	D 1	Migra	ng EMS Virtual Ma	chinos in a VMwaro Clustor	225
	D.1 D.2	Movin	TIG LING VII LUAI MA	chines in a vieware cluster	233
	D.2			Jer-v Cluster	
Е	Cust	om X.5	9 Certificates- Sup	plementary Procedures	241
	E.1	Load	eyStore File to PC	Client	241
	E.2	Clean	p Temporary Files	on EMS Server	
	E.3	Clean	p Java Web Start (Client Files	
	E.4	Updat	the Java Security	Level on PC	243
	E.5	Instal	ng Custom Certific	ates on AudioCodes Devices	245
		E.5.1	Sinale-Sian On to the	AudioCodes Device	245
		E.5.2	Enterprise Gateways	and SBC Devices	246
			E.5.2.1 Step 1: Gene	rate a Certificate Signing Request (CSR)	246
			E.5.2.2 Step 2: Rece	ive the New Certificates from the CA	247
			E.5.2.3 Step 3: Upda	te Device with New Certificate	
			E.5.2.4 Step 4: Upda	te Device's Trusted Certificate Store	
			E.5.2.5 Step 5: Conti	gure HTTPS Parameters on the Device	
		E 5 2	MD 1xx Dovices	Device to Apply the New Configuration	
		E.5.5	E 5 3 1 Stop 1: Cono	rate a Cortificate Signing Pequest (CSP)	203
			E 5 3 2 Sten 2: Rece	ive the New Certificates from the CA	254
			E.5.3.3 Step 3: Upda	te Device with New Certificate	
			E.5.3.4 Step 4: Upda	te Device's Trusted Certificate Store	255
			E.5.3.5 Step 5: Confi	gure HTTPS Parameters on Device	257
			E.5.3.6 Step 6: Rese	Device to Apply the New Configuration	257
F	Tran	sferring	Files		259
G	Verif	ying ar	d Converting Certif	icates	261
н	Self-	Signed	Certificates		
	H.1	Intern	t Explorer		
	H.2	Using	Mozilla Firefox		264
	H.3	Chror	е		265

List of Figures

Figure 5.1: Linux Testing Dequirements	24
Figure 5-1: Linux Testing Requirements	
Figure 6-2: Linux CentOS Installation	37
Figure 6-3: CentOS 5	
Figure 6-4: Linux CentOS Installation Complete	
Figure 6-5: Linux CentOS Network Configuration	39
Figure 6-6: Oracle DB Installation (Linux)	40
Figure 6-7: Oracle DB Installation - License Agreement (Linux)	41
Figure 6-8: Oracle DB Installation (Linux) (cont)	41
Figure 6-9: Oracle DB Installation (Linux) (cont)	41
Figure 6-10: EMS Server Application Installation (Linux)	42
Figure 6-11: EMS Server Application Installation (Linux) – License Agreement	43
Figure 6-12: EMS Server Application Installation (Linux) (cont)	43
Figure 6-13: EMS Server Application Installation (Linux) - Java Installation	44
Figure 7-1: Vmware vSphere Web Client	48
Figure 7-2: Hosts and Clusters	48
Figure 7-3: Deploy OVE Template Option	49
Figure 7-4: Client Integration Plug-in	49
Figure 7-5: Browse to OVE Package	50
Figure 7-6: OVF Template Details Screen	51
Figure 7-7: Virtual Machine Name and Location Screen	51
Figure 7-8: Destination Storage Screen	
Figure 7-9:: Setup Networking Screen	53
Figure 7-10: Ready to Complete Screen	
Figure 7-11: Deployment Progress Screen	54
Figure 7-12: Edit Settings option	
Figure 7-13: CPU, Memory and Hard Disk Settings	
Figure 7-14: Recent Tasks	56
Figure 7-15: Power On	
Figure 7-16: Storage Adapters	57
Figure 7-17:Turn On vSphere HA	57
Figure 7-18: Activate HA on each Cluster Node	58
Figure 7-19: Networking	58
Figure 7-20: Switch Properties	59
Figure 7-210: Installing the EMS server on Hyper-V – Hyper-V Manager	61
Figure 7-221: Installing EMS server on Hyper-V – Import Virtual Machine Wizard	62
Figure 7-232: Installing EMS server on Hyper-V – Locate Folder	62
Figure 7-243: Installing EMS server on Hyper-V – Choose Import Type	63
Figure 7-254: Installing EMS server on Hyper-V – Choose Destination	63
Figure 7-265: Installing EMS server on Hyper-V – Choose Storage Folders	64
Figure 7-276: File Copy Progress Bar	64
Figure 7-287: Adjusting VM for EMS server – Settings - Memory	65
Figure 7-298: Adjusting VM for EMS server - Settings - Processor	66
Figure 7-300: Expanding Disk Capacity	67

Figure 7-32: Edit Virtual Hard Disk Wizard-Choose Action	3)
Figure 7-33: Edit Virtual Hard Disk Wizard-Expand Virtual Hard Disk)
Figure 7-34: Edit Virtual Hard Disk Wizard-Completion)
Figure 7-35: Advanced Features - Network Adapter – Static MAC Address	
Figure 7-36: Power On Virtual Machine	2
Figure 7-37: Connect to EMS Server Console	
Figure 7-38: Hyper-V-Failover Cluster Manager Nodes	5
Figure 7-39: Configure Role	Ļ
Figure 7-40: Choose Virtual Machine	-
Figure 7-41: Confirm Virtual Machine	;
Figure 7-42: Virtual Machine Successfully Added	;
Figure 8-1: EMS Server Upgrade (Linux)82	
Figure 8-2: EMS Server Upgrade (Linux) – License Agreement	
Figure 8-3: EMS Server Application Upgrade (Linux) - Java Installation83	5
Figure 8-4: EMS Server Upgrade (Linux) Complete	5
Figure 8-5: EMS Server Upgrade (Linux)	ŀ.
Figure 9-1: VMware vSphere Web Client	;
Figure 9-2: Hosts and Clusters	;
Figure 9-3: Edit Settings Option	•
Figure 9-4: Connection Options	•
Figure 9-5: Help Link to Launch Remote Console	5
Figure 9-6: Remote Console Application)
Figure 9-7: Virtual Machine Settings)
Figure 9-8: Connect to Host CD Device/ Datastore ISO file90)
Figure 9-9: CD/DVD Drive - Connected Status)
Figure 9-10: EMS Server Installation Script	
Figure 9-11: EMS Server Upgrade (Linux) – License Agreement	
Figure 9-12: EMS Server Application Upgrade (Linux) - Java Installation	2
Figure 9-13: EMS Server Upgrade (Linux) Complete93	}
Figure 12-1: EMS Server Manager Menu104	
Figure 13-1: Application Status)
Figure 14-1: General Information111	
Figure 14-2: General Information	
Figure 15-1: EMS Server Manager – Collect Logs113	}
Figure 15-2: TAR File Location114	
Figure 16-1: Application Maintenance115)
Figure 16-2: Start or Stop the EMS Server116	;
Figure 16-3: - Web Servers116	;
Figure 16-4: JAWS IP Configuration118	;
Figure 16-5: License Manager)
Figure 17-1: Network Configuration	5
Figure 17-2: EMS Server Manager – Change Server's IP Address124	
Figure 17-3: IP Configuration Complete	
Figure 17-4: EMS Server: Triple Ethernet Interfaces	i
Figure 17-5: EMS Server Manager – Configure Ethernet Interfaces	i

Figure 17-6: Physical Ethernet Interfaces Redundancy	.128
Figure 17-7: Ethernet Redundancy Configuration	.129
Figure 17-8: Add Redundant Interface (Linux)	.130
Figure 17-9: Ethernet Redundancy Interface to Disable	.131
Figure 17-10: Modify Redundant Interface (Linux)	.132
Figure 17-11: DNS Setup	.133
Figure 17-12: Routing Table and Menu	.134
Figure 17-13: EMS Server Manager – Configure SNMPv3 Engine ID	.136
Figure 17-14: SNMPv3 Engine ID Configuration – Complete Configuration	.136
Figure 18-1: EMS Server Manager - Change System Time & Date	.137
Figure 18-2: EMS Server Manager - Configure NTP	.138
Figure 18-3: Change System Time and Date Prompt	.140
Figure 19-1: Security Settings	.141
Figure 19-2: SSH Configuration	.143
Figure 19-3: SSH Log Level Manager	.144
Figure 19-4: SSH Banner Manager	.144
Figure 19-5: Configure SSH on Ethernet Interfaces	.145
Figure 19-6: Disable Password Authentication	.147
Figure 19-7: SSH IgnoreUserKnowHosts Parameter - Confirm	.147
Figure 19-8: Configure SSH Allowed Hosts	.148
Figure 19-9: Add Host/Subnet to Allowed Hosts	.149
Figure 19-10: Add Host/Subnet to Allowed Hosts-Configured Host	.150
Figure 19-11: EMS Server Manager – Change DB Password	.151
Figure 19-12: OS Passwords Settings with Security Extensions	.154
Figure 19-13: Maximum Active SSH Sessions.	.154
Figure 19-14: Software Integrity Checker (AIDE) and Pre-linking	.155
Figure 19-15: USB Storage	.156
Figure 19-16: Network Options	.157
Figure 19-17: Auditd Options	.157
Figure 19-18: OVOC Maximum Security Implementation	.158
19-19: HTTPS Authentication	.159
19-20:Server Certificate Updates	.161
Figure 19-21: Generate Server Private Key	.161
Figure 19-22: Server Private Key Generated	.162
Figure 19-23: Generating a Server Certificate Signing Request (CSR)	.162
Figure 19-24: Transfer CSR File to PC	.163
Figure 19-25: Installed Server Certificate	.165
Figure 19-26: Installed Root Certificate	.165
Figure 19-27: Update Client Java Web Start Certificates	.166
Figure 19-28: Enter Java Web Start Passphrase	.166
Figure 19-29: Java Files Updating	.167
Figure 19-30: SEM - AudioCodes Device Communication	.168
- Figure 20-1: Diagnostics	.169
Figure 20-2: Syslog Configuration	.170
Figure 20-3: Forward Messages to an External Server	.170
Figure 22-1: EMS Server Manager - HA Configuration	.181

Figure 22-2: Primary HA Server Menu	182
Figure 22-3: Primary HA Server Sub-menu	182
Figure 22-4: HA Configuration Display	183
Figure 22-5: HA Server Configured as Primary Server - Confirmation	183
Figure 22-6: Primary HA Server Menu	184
Figure 22-7: Primary HA Server Sub-menu	184
Figure 22-8: HA Configuration Display	185
Figure 22-9: HA Server Configured as Primary Server - Confirmation	185
Figure 22-10: Primary HA Server IP	187
Figure 22-11: Secondary HA Server Configuration	187
Figure 22-12: EMS HA Status - Example Display	188
Figure 22-13: Advanced Status View	190
Figure 22-14: Manual Switchover	191
Figure 22-15: Switchover Status	191
Figure 22-16: Status after Switchover	192
Figure 22-17: Uninstall EMS HA Status Display	193
Figure 23-1: Firewall Configuration Schema	204
Figure 24-1: EMS Client Installation-Run as Administrator	207
Figure 24-2: EMS Client Installation File-Windows 8 Properties	208
Figure 24-3: EMS Client Installation File-Compatibility Tab	209
Figure 24-4: Running EMS Client-Run as Administrator	210
Figure A-1: Save MGs Tree Command	215
Figure B-1: Alarm Receiver	221
Figure B-2: Destination Rule Configuration	222
Figure B-3: Users List	223
Figure B-4: Actions Journal	224
Figure B-5: Media Gateway Search	225
Figure C-1: Hardware Preparation	227
Figure C-2: HP Array Configuration Utility (ACU)	228
Figure C-3: RAID-Latest Firmware Versions	229
Figure C-4: Actions Menu	229
Figure C-5: Clear Configuration	230
Figure C-6: Summary Screen	230
Figure C-7: Main Screen	231
Figure C-8: Logical Drive	231
Figure C-9: Summary Screen	232
Figure C-10: Set Bootable Logical Drive/Volume	232
Figure C-11: Set Bootable Logical Drive/Volume	233
Figure C-12: Exit Application	233
Figure C-13: Power Button	234
Figure C-14: Reboot Button	234
Figure D-1: Migration	235
Figure D-2: Change Host	235
Figure D-3: Target Host for Migration	236
Figure D-4: Migration Process Started	236
Figure D-5: Hyper-V Live Migration	237

Figure D-6: Move Virtual Machine	238
Figure D-7: Hyper-V Migration Process Started	239
Figure E-8: Java KeyStore	241
Figure E-9: Java Control Panel (Version 7.2)	243
Figure E-10: Java Control Panel (Version 8.0)	244
Figure E-11: Exception Site List	245
Figure E-12: Context Certificates	246
Figure E-13: Certificate Signing Request Group	247
Figure E-14: Upload Certificate Files from your Computer Group	248
Figure E-15: Importing Certificate into Trusted Certificates Store	249
Figure E-16: TLS Contexts: Edit Record	251
Figure E-17: Device Reset	252
Figure E-18: Certificate Signing Request Group	253
Figure E-19: Maintenance Actions Page	257
Figure H-1: Continue to Website	
Figure H-2: Mozilla Firefox Settings	264
Figure H-3: Chrome Browser Settings	265

List of Tables

Table 3-1: EMS- Minimal Platform Requirements	23
Table 3-2: SEM Bandwidth Requirements	24
Table 3-3: Performance and Data Storage	26
Table 7-1: Virtual Machine Configuration	54
Table 7-2: Virtual Machine Configuration	65
Table 13-1: Application Statuses	109
Table 23-1: Firewall Configuration Rules	197
Table 23-2: OAM Flows: NOC ↔ Device/ IP Phone/ SBA/ EMS	205
Table 23-3: OAM Flows: Device/ IP Phone/ SBA/ EMS↔ NOC	205
Table B-1: Acceptance Test – Client Installation	217
Table B-2: Acceptance Test – Server Installation	218
Table B-3: Acceptance Test – Add Auxiliary File	218
Table B-4: Acceptance Test – Add MG	218
Table B-5: Acceptance Test – Entity Profile: Digital CPE Devices	219
Table B-6: Acceptance Test – Analog CPE Devices	220
Table B-7: Acceptance Test – Alarm Receiver	221
Table B-8: Acceptance Test – Delete Alarms	221
Table B-9: Acceptance Test – Acknowledge Alarm	221
Table B-10: Acceptance Test – Forwarding Alarms	222
Table B-11: Acceptance Test – Add an Operator	223
Table B-12: Acceptance Test – Non Repetitive Passwords	223
Table B-13: Acceptance Test – Removing Operator	224
Table B-14: Acceptance Test – Journal Activity	224
Table B-15: Acceptance Test – MG Search	226
Table B-16: Acceptance Test – Online Help	226
Table B-17: Acceptance Test – Backup and Recovery	226

Notice

Information contained in this document is believed to be accurate and reliable at the time of printing. However, due to ongoing product improvements and revisions, AudioCodes cannot guarantee accuracy of printed material after the Date Published nor can it accept responsibility for errors or omissions. Updates to this document and other documents can be viewed by registered customers at http://www.audiocodes.com/downloads.

This document is subject to change without notice.

Date Published: May-11-2017

WEEE EU Directive

Pursuant to the WEEE EU Directive, electronic and electrical waste must not be disposed of with unsorted waste. Please contact your local recycling authority for disposal of this product."

Customer Support

Customer technical support and service are generally provided by AudioCodes' Distributors, Partners, and Resellers from whom the product was purchased. For technical support for products purchased directly from AudioCodes, or for customers subscribed to AudioCodes Customer Technical Support (ACTS), contact support@audiocodes.com.

Documentation Feedback

AudioCodes continually strives to produce high quality documentation. If you have any comments (suggestions or errors) regarding this document, please fill out the Documentation Feedback form on our Web site at http://www.audiocodes.com/downloads.

Related Documentation

Manual Name
Mediant 500 MSBR User's Manual
Mediant 500L MSBR User's Manual
Mediant 500 E-SBC User's Manual
Mediant 500L E-SBC User's Manual
Mediant 800B Gateway and E-SBC User's Manual
Mediant 800B MSBR User's Manual
Mediant 1000B Gateway and E-SBC User's Manual
Mediant 1000B MSBR User's Manual
Mediant 2600 SBC User's Manual
Mediant 3000 User's Manual
Mediant 4000 SBC User's Manual
Mediant 9000 SBC User's Manual
Mediant Software SBC User's Manual
Element Management System (EMS) Server Installation, Operation and Maintenance Manual
Element Management System (EMS) Product Description
Element Management System (EMS) OAM Integration Guide
Element Management System (EMS) User's Manual
SEM User's Manual
IP Phone Management Server Administrator's Manual
IP Phone Manager Express Administrator's Manual
AudioCodes One Voice Operations Center Product Description
OVOC Security Guidelines
Element Management System (EMS) Online Help
Mediant 500 Gateway and E-SBC Performance Monitors and Alarms Guide
Mediant 800 Gateway and E-SBC Mediant Software SBC CloudBond 365 and CCE Alarms Guide
Mediant 1000B Gateway and E-SBC Performance Monitors and Alarms Guide
Mediant 2600-4000-9000-SW SBC Series Performance Monitors and Alarms Guide
Mediant 3000 with TP-6310 Performance Monitoring and Alarms Guide
Mediant 3000 with TP-8410 Performance Monitoring and Alarms Guide

Mediant MSBR Series Performance Monitors and Alarms Guide

IOM Manual

1 Overview

The EMS provides customers with the capability to easily and rapidly provision, deploy and manage AudioCodes devices.

Provisioning, deploying and managing these devices with the EMS are performed from a centralized management station in a user-friendly Graphic User Interface (GUI).

This document describes the installation of the EMS server and its components.

It is intended for anyone responsible for installing and maintaining AudioCodes' EMS server and the EMS server database.



This page is intentionally left blank.

Part I

Pre-installation Information

This part describes the EMS server components, requirements and deliverables.



2 Managed VoIP Equipment

The following products (and product versions) can be managed by this EMS / SEM release:

- Mediant 9000 SBC versions 7.2 (including support for MTC), 7.0, 6.8
- Mediant 4000 SBC versions 7.2, 7.0, 6.8
- Mediant 4000B SBC version 7.2, 7.0
- Mediant 2600 E-SBC versions 7.2, 7.0, 6.8
- Mediant 2600B E-SBC version 7.2, 7.0
- Mediant Server Addition (SE) SBC versions 7.2, 7.0 and 6.8
- Mediant Virtual Addition (VE) SBC versions 7.2 (including support for MTC), 7.0 and 6.8
- Mediant 3000 Media Gateways (TP-8410 and TP-6310) versions 7.0 (SIP), 6.8 (SIP) and 6.6 (SIP and MEGACO)
- Mediant 2000 Media Gateways version 6.6
- *Mediant 1000 Gateway version 6.6 (SIP and MEGACO)
- Mediant 1000B Gateway and E-SBC versions **7.2** 7.0, 6.8 and 6.6
- Mediant 1000B MSBR versions 6.8
- Mediant 800B Gateway and E-SBC versions 7.2 7.0, 6.8 and 6.6
- Mediant 500 E-SBC version 7.2
- Mediant 500L E-SBC version 7.2
- Mediant 800B MSBR versions 7.2, 6.8 and 6.6
- Mediant 600 version 6.6
- Mediant 500L MSBR, Mediant 500 MSBR version 7.2, 6.8
- MediaPack MP-11x series version 6.6 (SIP and MEGACO)
- MediaPack MP-124 Rev. **D** and **E** version 6.6 (SIP and MEGACO
- MP-20x B version 7.2
- MP-1288 version 7.2
- *Mediant 800B SBA, *Mediant 1000B SBA, and *Mediant 2600B SBA devices with SBA version 1.1.12.x and above and gateway versions 7.2, 7.0, 6.8
- 400 HD Series: from version 2.0.13 (Skype for Business) and from version 2.2.2 (Non-Lync): 420HD, 430HD and 440HD and 405.
- 400 HD Series from version 3.0 (Skype for Business): 450 HD
- CloudBond 365 Series: version 7.4: Standard Edition (Mediant 800B platform); Standard Plus Edition (Mediant 800B platform); Pro Edition (Mediant Server platform); Enterprise Edition (Mediant Server platform); Virtualized Edition (Mediant Server platform).
- Mediant 800 CCE Appliance and Mediant Server CCE Appliance¹

¹ Contact AudioCodes regarding version support for these products.





Note:

- * Refers to products that are not supported by the SEM.
- All versions VoIP equipment work with the SIP control protocol.
- Bold refers to new product support and version support.

3 Hardware and Software Specifications

This section describes the hardware and software specifications of the EMS server.

3.1 EMS Server and Client Requirements

This section lists the platform and software required to run the EMS dedicated hardware version and the VMware and Hyper-V version.

	EMS/SEM Server			EMS Client
Resource	Dedicated EMS Server - Linux OS	Virtual EMS - Low Profile	Virtual EMS - High Profile	
Hardware	HP ProLiant DL360p Gen8	-	-	Monitor resolution: 1152*864 or higher
Operating System	Linux CentOS 64-bit, kernel version 5.9, Rev7	Linux CentOS 64-bit, kernel version 5.9 Rev7 (Rev8 for EMS-	Linux CentOS 64-bit, kernel version 5.9	Windows™ 10/Windows 8/Windows 8.1/ Windows 7/ Windows 7 Enterprise/ Windows Server 2012 R2 Standard
	For EMS HA: Linux CentOS 64-bit, kernel version 5.9, Rev8	[–] HA).	Rev7	
Virtualization	_	VMware: ESXi 6.0, 5.0 and 4.1		_
plation		VMware HA cluster: VMware ESXi 5.5		
		Microsoft Hyper-V Windows server 2012R2		
Memory	32 GB RAM	8 GB RAM	32 GB RAM	512 MB RAM
Disk space	Disk: 2 X 1.2 TB SAS 10K RPM in RAID 0	500 GB	1.2 TB	300 MB
Processor	CPU: Intel Xeon E5- 2690 (8 cores 2.9 GHz each)	1 core not less than 2.5 GHz	6 cores not less than 2 GHz	
DVD-ROM	Local	_	_	_

Table 3-1	: EMS-	Minimal	Platform	Rea	uirements

AudioCodes

- The working space requirements on the EMS server are as follows:
 - Linux: Executable bash
- The EMS server works with the Java Development Kit (JDK) version 1.8 (JDK 1.8 for Linux[™]). The EMS client works with the JRE version 1.8 for Windows[™].
- The Oracle database used is version 11g.
- Supported browsers for client applications are as follows:
 - Internet Explorer version 11 and higher
 - Mozilla Firefox version 38 and higher
 - Google Chrome version 43 and higher

Note:

- The JDK and Oracle database component versions mentioned above are provided as part of the EMS server and EMS client installation images.
- The Java Runtime Environment (JRE) is not provided in the client installation and therefore you must install it on your PC in order to launch the EMS client using the Java Webstart.
- The above browsers are supported to run the following client applications: EMS/devices Single Sign-on, JAWS, NBIF, SEM and IP Phone Manager.

3.2 Bandwidth Requirements

This section lists the EMS and SEM bandwidth requirements.

3.2.1 EMS Bandwidth Requirements

The bandwidth requirement is for EMS/SEM Server <-> Device communication. The network bandwidth requirements per device is 500 Kb/sec for faults, performance monitoring and maintenance actions.

3.2.2 SEM Bandwidth Requirements

The following table describes the upload bandwidth speed requirements for monitoring the different CPE devices using the SEM. The bandwidth requirement is for EMS/SEM Server <-> Device communication.

Device	SBC Sessions (each session has two legs)	Required Kbits/sec or Mbit/sec
SBC		
MP-118	-	_
MP-124	_	_

Table 3-2: SEM Bandwidth Requirements

Device	SBC Sessions (each session has two legs)	Required Kbits/sec or Mbit/sec	
Mediant 800 Mediant 850	60	135 Kbits/sec	
Mediant 1000	150	330 Kbits / sec	
Mediant 2000	-	_	
Mediant 2600	600	1.3 Mbit/sec	
Mediant 3000	1024	2.2 Mbit/sec	
Mediant 4000	4,000	8.6 Mbit/sec	
Gateway			
MP-118	8	15 Kbits/sec	
MP-124	24	45 Kbits/sec	
Mediant 800 Mediant 850	60	110 Kbits/sec	
Mediant 1000	120	220 Kbits/sec	
Mediant 2000	480	880 Kbits/sec	
Mediant 2600	-	_	
Mediant 3000	2048	3.6 Mbit/sec	
Mediant 4000	-	_	
Endpoints	-	56 Kbits/sec	

3.3 Performance and Data Storage

The following table shows the performance and data storage capabilities for the EMS managed devices, EMS for IP Phones managed devices and for the SEM.

Machine Specifications	HP DL360p G8	VMware/Microsoft Hyper-V - Low Profile	VMware/Microsoft Hyper-V – High Profile
EMS Managed Devices	5000	100	5000
Maximum number of managed endpoints in EMS (IP Phone Manager only).	-	-	30,000 (VMware only).
SEM			
Maximum Number of CAPS (calls attempts per second) per device.	160	30	120
Maximum number of CAPS per server (SBC and Skype for Business).	300	30	120
Maximum concurrent sessions	30,000	3,000	12,000
Maximum number of devices per region	500	100	300
Maximum number of managed devices.	3,000	100	1,200
Maximum number of links between devices.	6,000	200	2,400
Call Details Storage - Detailed information per Call	Up to two months or 80 million calls.	Up to two months or 6 million rows.	Up to two months or 80 million rows.
Calls Statistics Storage - Statistic information storage.	Up to six months or 150 million intervals.	Up to six months or 12 million rows.	Up to six months or 150 million rows.
Maximum number of managed endpoints in both EMS (IP Phone Manager) and SEM.	10000	1000	5000
Maximum number of CAPS per endpoint.	10	1	5

Table 3-3: Performance and Data Storage

3.4 Microsoft Lync Monitoring SQL Server Prerequisites

Following are the Microsoft Lync Monitoring SQL Server prerequisites:

- The server must be defined to accept login in 'Mix Authentication' mode.
- The server must be configured to collect calls before the SEM can connect to it and extract Lync calls.
- Call Detail Records (CDRs) and Quality of Experience (QoE) Data policies must be configured to capture data.
- Network administrators must be granted the correct database permissions (refer to the SEM User's Manual).
- Excel macros must be enabled so that the SQL queries and reports can be run. This was tested with Excel 2010.
- Detailed minimum requirements for Microsoft Lync SQL Server can be found in the following link:

http://technet.microsoft.com/en-us/library/gg412952.aspx



This page is intentionally left blank.

4 **EMS Software Deliverables**

This section describes the EMS software deliverables.

4.1 Dedicated Hardware Installation – DVDs 1-4

This section describes the DVDs supplied in the EMS software delivery.

- **DVD1:** Operating System DVD for Linux:
 - Linux (CentOS) 5.9 Installation for EMS server, REV7

The following machine is currently supported:

- HP DL360p G8 Linux (CentOS) 64-bit kernel version 5.9 Installation for EMS server, Linux CentOS 5.9 REV7 or REV8 for EMS-HA.
- **DVD2:** Oracle Installation: Oracle installation version *11g* DVD for the Linux platform.
- **DVD3:** Software Installation and Documentation DVD for Linux:

The DVD 'SW Installation and Documentation' DVD comprises the following folders:

- Documentation All documentation related to the present EMS version. The documentation folder includes the following documents and sub-folders:
 - EMS Release Notes Document includes the list of the new features introduced in the current software version as well as version restrictions and limitations.
 - EMS Server IOM Manual Installation, Operation and Maintenance Guide.
 - EMS Product Description Document
 - EMS User's Manual Document
 - OAMP Integration Guide Document
 - 'GWs_OAM_Guides' folder document set describing Provisioning parameters and Alarm/Performance measurements parameters supported for each one of the products or product families.
 - 'Private_Labeling' folder includes all the information required for the OEM to create a new private labeling DVD. EmsClientInstall – EMS client software to be installed on the operator's Windows[™] based workstation.
- 'EmsClientInstall'-EMS client software to install on the designated client workstation PC.
- 'EmsServerInstall' EMS server software, to install on the dedicated Linux based EMS server machine.
- DVD4: (relevant for future releases) EMS Server Patches: Upgrade patches DVD containing OS (Linux) patches, Oracle patches, java patches or any other EMS required patches. This DVD enables the upgrading of the required EMS patches without the EMS application upgrade.

4.2 VMware – DVD 5

The EMS software delivery for the VMware DVD includes the following folders:

- VMware for clean install
- EMS client Install
- Documentation

4.3 Hyper-V – DVD 5

The EMS software delivery for the Hyper-V DVD includes the following folders:

- Hyper-V for clean install
- EMS client Install
- Documentation

Part II

EMS Server Installation

This part describes the testing of the installation requirements and the installation of the EMS server.



5 Testing Installation Requirements -Dedicated Hardware

Before commencing the EMS server installation procedure, verify that your system meets the hardware, disk space, operating system and other requirements that are necessary for a successful installation.

5.1 Hardware Requirements

Operating System – the Linux Operating Systems are supported.

To determine the system OS, enter the following command:

uname

This command returns Linux. Proceed to the following section :

• Testing Hardware Requirements on Linux OS (see Section 5.1.1 on page 33).

5.1.1 Testing Hardware Requirements on the Linux Platform

To ensure that your machine meets the minimal hardware requirements for running the EMS application on both dedicated and virtual hardware, run the commands described below in **tcsh**.

RAM - A minimum of <machine type_RAM> GB is required (see Chapter 3 on page 23). To determine the amount of random access memory installed on your system, enter the following command:

more /proc/meminfo | grep MemTotal

Swap Space - Swap space is twice the system's physical memory, or 4 GB, whichever is greater.

To determine the amount of swap space currently configured in your system, enter the following command:

more /proc/meminfo | grep SwapTotal

Disk Space – A minimum of <machine type_disk space> GB is required (see Chapter 3 on page 23). To determine the amount of disk space on your system, enter the following command:

fdisk -l | grep Disk

During the application installation, you are required to reserve up to 2 GB of Temporary disk space in the **/tmp**. If you do not have enough space in the **/tmp** directory, set the **TMPDIR** and **TMP** environment variables to specify a directory with sufficient space.



DVD-ROM device - A DVD-ROM drive capable of reading ISO 9660 format.

Figure 5-1: Linux Testing Requirements

```
[root@EMS-Server-Linux113 ~]# tcsh
[root@EMS-Server-Linux113 ~]# uname
Linux
[root@EMS-Server-Linux113 ~]# more /proc/meminfo | grep MemTotal
MemTotal: 2017056 kB
[root@EMS-Server-Linux113 ~]# more /proc/meminfo | grep SwapTotal
SwapTotal: 3020180 kB
[root@EMS-Server-Linux113 ~]# fdisk -1 | grep Disk
Disk /dev/sda: 250.0 GB, 250059350016 bytes
[root@EMS-Server-Linux113 ~]#
```



Note: Use the AudioCodes' DVD1 to install the Linux Operating System.

6

Installing the EMS Server on Dedicated Hardware

The EMS server installation process supports the Linux platform. The installation includes four separate components, where each component is supplied on a separate DVD:

- **DVD1:** OS installation: OS installation DVD.
- **DVD2:** Oracle Installation: Oracle installation DVD platform.
- **DVD3:** EMS application: EMS server application installation DVD.
- DVD4: (relevant for future releases) EMS Server Patches: Upgrade patches DVD containing OS (Linux) patches, Oracle patches, java patches or any other EMS required patches. This DVD enables the upgrading of the EMS required patches without the EMS application upgrade.

While a clean installation requires the first three DVDs (DVD1, DVD2 and DVD3), an EMS application upgrade requires only the 'EMS server application (DVD3)'. The 'Patches upgrade' requires only the 'EMS server Patches (DVD4)'.

6.1 **ISO Files Verification**

If you have received an ISO file from AudioCodes instead of a burned DVD, its contents must be verified using an MD5 checksum. As an Internet standard (RFC 1321), MD5 has been used in a wide variety of security applications, and is also is commonly used to check the integrity of file, and verify download. Perform the following verifications on the relevant platform:

- Windows (see below)
- Linux (see Section 6.1.2).

6.1.1 Windows

Use the WinMD5 tool to calculate md5 hash or checksum for the ISO file:

Verify the checksum with WinMD5 (see www.WinMD5.com)

6.1.2 Linux

Copy the checksum and the ISO files to a Linux machine, and then run the following command:

md5sum -c filename.md5

The "OK" result should be displayed on the screen (see figure below).

Figure 6-1: ISO File Integrity Verification


6.2 Installing the EMS Server on the Linux Platform

This section describes how to install the EMS server on the Linux platform.

6.2.1 DVD1: Linux CentOS 5.9

The procedure below describes how to install Linux CentOS 5.9. This procedure takes approximately 20 minutes.



Note: If you are installing the EMS server on an HP ProLiant DL360p Gen8 server, before commencing this procedure, you must configure RAID-0 (see Appendix C on page 227).

To perform DVD1 installation:

- 1. Insert the DVD1-Linux for EMS Rev7 (CentOS 5.9) into the DVD ROM.
- 2. Connect the EMS server through the serial port with a terminal application and login with 'root' user. Default password is *root*.
- 3. Perform EMS server machine reboot by specifying the following command: reboot
- 4. Press Enter; you are prompted whether you which to start the installation through the RS-232 console or through the regular display.
- 5. Press Enter to start the installation from the RS-232 serial console or type **vga**, and then press Enter to start the installation from a regular display.

Figure 6-2: Linux CentOS Installation





Figure 6-3: CentOS 5



- 6. Wait for the installation to complete.
- 7. Reboot your machine by pressing Enter.



Note: Do not forget to remove the Linux installation DVD from the DVD-ROM before rebooting your machine.

Figure 6-4: Linux CentOS Installation Complete



8. Login as 'root' user with password *root*.

9. Type **network-config**, and then press Enter; the current configuration is displayed:

I 0.7.19.100:22 - Tera Term VT

I e Edt Setup Control Window Help

ems-server login: root
Password:
[root@ems-server ~]# network-config
Current network configuration:
Hostname : ems-server
IP Address : 169.254.101.1
Subnet Mask : 255.255.0.0
Default Gateway : 169.254.0.1
Do you wish to change it? (y/[n]) : y
Hostname : EMS-Linux145
IP Address : 10.7.14.145
Subnet Mask : 255.255.0.0
Default Gateway : 10.7.0.1
Apply new configuration? ([y]/n) : y
-----EMS server must be rebooted to activate the new
network configuration.
Reboot now? ([y]/n) : y

Figure 6-5: Linux CentOS Network Configuration

- **10.** You are prompted to change the configuration; enter **y**.
- **11.** Enter your Hostname, IP Address, Subnet Mask and Default Gateway.
- **12.** Confirm the changes; enter **y**.
- **13.** You are prompted to reboot; enter **y**.

6.2.2 DVD2: Oracle DB Installation

The procedure below describes how to install the Oracle database. This procedure takes approximately 30 minutes.



Note: Before starting the installation, it is highly recommended to configure the SSH client (e.g. Putty application) to save the session output into a log file.

> To perform DVD2 installation:

- 1. Insert DVD2-Oracle DB installation into the DVD ROM.
- 2. Login into the EMS server by SSH, as 'acems' user, and enter password acems.
- 3. Switch to 'root' user and provide root password (default password is root):

su - root

- On some machines, you need to mount the CDROM in order to make it available: mount /misc/cd
- 5. Run the installation script from its location:

cd /misc/cd ./install

Figure 6-6: Oracle DB Installation (Linux)



6. Enter y, and then press Enter to accept the License agreement.

Figure 6-7: Oracle DB Installation - License Agreement (Linux)



7. Type the 'SYS' user password, type **sys** and then press Enter.

Figure 6-8: Oracle DB Installation (Linux) (cont)



8. Wait for the installation to complete; reboot is not required at this stage.

Figure 6-9: Oracle DB Installation (Linux) (cont)



6.2.3 DVD3: EMS Server Application Installation

The procedure below describes how to install the EMS server application. This procedure takes approximately 20 minutes.

To perform DVD3 installation:

- 1. Insert DVD3-EMS Server Application Installation into the DVD ROM.
- 2. Login into the EMS server by SSH, as 'acems' user, and enter the password *acems*.
- 3. Switch to 'root' user and provide *root* password (default password is *root*):

```
su - root
```

4. Run the installation script from its location:

```
cd /misc/cd/EmsServerInstall/
./install
```

Figure 6-10: EMS Server Application Installation (Linux)



5. Enter **y**, and then press Enter to accept the License agreement.

Figure 6-11: EMS Server Application Installation (Linux) – License Agreement

based upon the net income of bicensol.
11.4. Severability If any provision herein is ruled too broad in any respe
on shall be limited only so far as it is necessary to allow conformance to
shall be deleted from the Agreement, but the remaining provisions shall r
11.5. Assignment Neither this Agreement or any of Licensee's rights or obl
tten permission of Licensor and any attempt to do so shall be without effe
sferred to any person; (ii) the Licensee being merged or consolidated with
11.6. Export Licensee understands that the Licensed Software may be a regu
, and may require a license to export such. Licensee is solely responsible
11.7. Relationship of Parties Nothing herein shall be deemed to create an
the parties. Neither party shall have the right to bind the other to any o
11.8. Integration This Agreement is the complete and exclusive agreement b
ated hereto. Any Licensee purchase order issue for the software, documenta
erms hereof.
11.9. Counterparts This Agreement may be executed in multiple original cou
ing an authorized signature of Licensor and Licensee.
Do you accept this agreement? (y/n)y

6. When you are prompted to change the *acems* and *root* passwords, enter new passwords or enter existing passwords. You are then prompted to reboot the EMS server machine; press Enter.

udev.x86_64	095-14.20.e15_3	ems-local			
wget.x86_64	1.11.4-2.e15_4.1	ems-local			
wireshark.x86_64	1.0.11-1.el5_5.5	ems-local			
Hardening Linux OS for DoD STIG compliancy					
>>> Enter new password for user 'acems'					
Changing password for user acems.					
New UNIX password:					
BAD PASSWORD: it is too short					
Retype new UNIX password:					
passwd: all authentication tokens updated	successfully.				
>>> Enter new password for user 'root'					
Changing password for user root.					
New UNIX password:					
BAD PASSWORD: it is too short					
Retype new UNIX password:					
passwd: all authentication tokens updated	successfully.				
*****	+++++++++++++++++++++++++++++++++++++++				
EMS Server must be rebooted to proceed wit	h the installation.				
After the reboot completes, re-login to th	e EMS Server and				
re-run the installation script to complete the installation.					
***************************************	+++++++++++++++++++++++++++++++++++++++				
Press Enter to reboot					

Figure 6-12: EMS Server Application Installation (Linux) (cont)

- 7. After the EMS server has successfully rebooted, repeat steps 2 4.
- 8. At the end of Java installation, press Enter to continue.



Figure 6-13: EMS Server Application Installation (Linux) - Java Installation



9. Wait for the installation to complete and reboot the EMS server by typing reboot.

Done >>>	
>>> Remove /tmp/EmsServerInstall [root@EMS-Linux145 EmsServerInstall]#	

- **10.** When the EMS server has successfully restarted login into the EMS server by SSH, as 'acems' user and enter password *acems*.
- 11. Switch to 'root' user and provide root password (default password is root):

su - root

- **12.** Verify that the Date and Time are set correctly (see Section 18.3 to set the date and time).
- **13.** Verify that the EMS server is up and running (see Chapter 13 and login by client to verify a successful installation).

6.3 EMS Server Users

EMS server OS user permissions are differentiated according to the specific application task. This feature is designed to prevent security breaches and to ensure that a specific OS user is authorized to perform a subset of tasks on a subset of machine directories. The EMS server includes the following OS user permissions:

- 'root' user: User permissions for installation, upgrade, maintenance using EMS server manager and EMS application execution.
- acems user: The **only available user** for Login through SSH/SFTP tasks.
- *emsadmin* user: User with permissions for mainly the EMS server manager and EMS application for data manipulation and database access.
- oracle user: User permissions for the Oracle database access for maintenance such as installation, patches upgrade, backups and other Oracle database tasks.
- oralsnr user: User in charge of oracle listener startup.



This page is intentionally left blank.

7

Installing the EMS on Virtual Server Platform

This chapter describes how to install the EMS on a Virtual Server platform. The following procedures are described:

- Installing the EMS server on the VMware platform (see Section 7.1 on page 47).
- Installing the EMS server on Microsoft Hyper-V platform (see Section 7.2 on page 60).



Note: The AudioCodes EMS supports the VMware vSphere High Availability (HA) feature.

7.1 Installing the EMS Server on the VMware Platform

The installation of the EMS server on VMware vSphere platform includes the following procedures:

- Installing the Virtual Machine (VM) (see Section 7.1.1).
- Configuring the Virtual machine hardware settings (see Section 7.2.2).
- Connecting EMS server to network (see Section 7.1.3).
- Configuring EMS Virtual Machines (VMs) in a VMware Cluster (see Section 7.1.4)

7.1.1 Installing the VMware Virtual Machine

This section describes how to install the EMS server on the VMware vSphere platform. This procedure takes approximately 30 minutes. This time is estimated on the HP DL 360 G8 platform (with CPU, disk and memory as specified in Section 5.1.1). The upgrade time depends on the hardware machine where the VMware vSphere platform is installed.

To install the EMS Server on VMware vSphere:

- 1. Insert the vEMS installation DVD (**DVD5**) into the disk drive on the PC where the vSphere client is installed.
- 2. Login to the VMware vSphere Web client.



vmware [,] vSphere Web	Clie	ent ≜ ≣			Ūlvmr	ware@QASVVV		l Help 👻 l 🔍	Bearch	
Navigator	Ŧ	岱 Home							🔯 Alarms	ĮΧ
Hosts and Clusters 🕨 🔊		Home							All (1) Ne A	:k
ሰ Home		Inventories						*	🔶 qaswvcenter01.	:or
by vCenter Inventory Lists	>		_				_	-	Performance Ch	art
Hosts and Clusters	>			5		<u> </u>				
VMs and Templates	>	vCenter	Hosts and	VMs and	Storage	Networking	Con	ent		
Storage	> ::	Inventory Lists	Clusters	Templates			Libra	ries		_
Q Networking	>					-			Work in Progre.	- +
Policies and Profiles	>		O			G			UP VEMS-Edi	
🚳 Hybrid Cloud Manager	>	~	\mathbf{V}							
O vRealize Orchestrator	>_	Hybrid Cloud Manager (vRealize Orchestrator							
🚳 Update Manager	>									
🍇 Administration	>	Monitoring								
🕄 Tasks	-	Watch How-to V	ideos	5				L .		
		191		1474/201	100	1-34				_
Recent Tasks										¥×
Task Name		Target	Status		Initiator	Queu	ed For	Start Time	Completion Time	Se
Reconfigure virtual machine	1	VEMS-7.2.3030-10	 Complete 	d	vmware		19 ms	11/17/2016 10:03:1	11/17/2016 10:03:2	q
4				::						Þ

Figure 7-1: Vmware vSphere Web Client

3. In the vCenter Navigator, select **Hosts and Clusters**. A list of Hosts and Clusters is displayed.

vmware [®] vSphere Web Cli	ent n≜≣		ଧ । vmware@	QASWVCENTER01 -	 I Help 	o ▼ 1 🔍 S	earch
Navigator I	🗊 qaswCluster01	Actions 👻			=*	🔯 Alarms	Ϊ×
Home Generation of the second	Summary Monitor Summary Monitor Cluster Resourc Cluster Consum Related Objects Datacenter	Manage Related Objects gaswCluster01 Total Processors: 20 Total Vhotion Migrations: 48 Sign Related Object es es ASWDatacenter More Related Object	CPU USED: 14.07 GHz MEMORY USED: 128.08 GB STORAGE USED: 2.47 TB	Category I Category I This list is empty	• GHz • GHz • GHz • GB • GB • GB • GB • GB • GHz • GHZ	All (1)	New (1) Acknowl ter01.corp.audiocodes nce Charts Service He Progress I dit Resource S
😨 Recent Tasks							ŤΧ
Task Name	Target	Status	Initiator	Queued For	Start Tim	e	Completion Time

Figure 7-2: Hosts and Clusters



Figure 7-3: Deploy OVF Template Option

4. In the Navigator, select the cluster and from the right-click menu, choose **Deploy OVF Template**.

The following screen may be displayed if the Client Integration Plug-in is not installed on your PC. Click the **Download the Client Integration Plug-in** link to download this application to your PC and then install it.

Deploy OVF Template		?
1 Source	Select source Select the source location	
1b Review details 2 Destination	The Client Integration Plug-in must be installed to enable OVF functionality. Click the link below to download the installer. If installed, refresh the browser and allow access.	
2a Select name and folder 2b Select storage 3 Ready to complete	Download the Client Integration Plug-in	
	Back Next Finish Cam	cel

Figure 7-4: Client Integration Plug-in



D

eploy OVF Template		? H
1 Source	Select source Select the source location	
1b Review details 2 Destination 2a Select name and folder	Enter a URL to download and install the OVF package from the Internet, or browse to a location accessible from your computer, such as a local hard drive, a network share, or a CD/DVD drive.	;
2b Select a resource 2c Select storage 3 Ready to complete	• Local file Browse	•
	Back Next Finish Can	ncel
🔳 Open		
Look in:	🐌 OVF 🗨 🖻 📸 📰 🕶	
Recent Places	Name VEMS_7.2.271.ova	
Desktop Libraries		
Computer		
Network	۲	
	File name: Open	
	Files of type: OVF Packages (*.ovf, *.ova) Cancel	

Figure 7-5: Browse to OVF Package

5. Browse to the OVF file with extension OVA from the DVD disk, and click **Next**.

Deploy OVF Template			(? H
1 Source	Review details Verify the OVF tem	plate details	
1b Review details	Product	7.2	
2 Destination	Version		
2a Select name and folder	Vendor		
2b Selectistorage	Publisher	No certificate present	
2c Setup networks	Download size	7.6 GB	
3 Ready to complete	Size on disk	26.7 GB (thin provisioned) 60.0 GB (thick provisioned)	
	Description		
		Back Next Finish Ca	ncei

Figure 7-6: OVF Template Details Screen

6. In the OVF Template Details screen, click **Next**.

Figure 7-7: Virtual Machine Name and Location Screen

Deploy OVF Template		? **
1 Source ✓ 1a Select source ✓ 1b Review details	Select name and folder Specify a name and location for the deployed template Name: AudioCodes_EMS	
 2 Destination 2a Select name and folder 2b Select storage 2c Setup networks 3 Ready to complete 	Select a folder or datacenter Q Search Image: Comparison of the search of th	The folder you select is where the entity will be located, and will be used to apply permissions to it. The name of the entity must be unique within each vCenter Server VM folder.
		Back Next Finish Cancel

7. In the Name and Location screen, enter the desired virtual machine name and choose the inventory location (the Data Center to locate the machine), and then click **Next**.



Figure 7-8: Destination Storage Screen

Deploy OVF Template						(?)
 1 Source 1a Select source 1b Review details 2 Destination 2a Select name and folder 2b Select storage 	Select storage Select location to store the fil Select virtual disk format: VM Storage Policy: The following datastores are virtual machine configuration	es for the deployed templ hin Provision latastore Default accessible from the desti files and all of the virtual	ate ate) pu selected. Select th	ne destination data	astore for the
2c Setup networks	Name	Capacity	Provisioned	Free	Туре	Storage DRS
3 Ready to complete	Netapp04.lun2	3.00 TB	3.58 TB	1.55 TB	VMFS	
	Netapp04.lun1	1.50 TB	1.70 TB	840.06 GB	VMFS	
	datastore211	1.08 TB	310.22 GB	808.19 GB	VMFS	
	4					Þ
				Back	lext Finis	h Cancel

- 8. In the Storage screen, do the following:
 - Select Virtual Disk Format- choose the desired provisioning option ("Thin Provisioning' is recommended),
 - Select the data store where wish to locate your machine, and click Next.

Figure 7-9:: Setup Networking Screen

Deploy OVF Template				(?)
1 Source	Setup networks Configure the networks the deployed templ	ate should use		
 1a Select source 				
 1b Review details 	Source	Des	tination	Configuration
2 Destination	VM Network 4	VM Network		▼ ⊘
 2a Select name and folder 				
✓ 2b Select storage				
✓ 2c Setup networks				
 3 Ready to complete 	IP protocol: IPv4	IP allocatio	on: Static - Manual 🚯	
	Source: VM Network 4 - Description The VM Network 4 network			
	Destination: VM Network - Protocol settings			
	No configuration needed for this network			
			Back Next F	Finish Cancel

9. In the Network setup screen, select the network where the deployed template should apply, and click **Next**.

Figure 7-10: Ready to Complete Screen

Deploy OVF Template		
1 Source	Ready to complete Review your settings selection	ns before finishing the wizard.
 tb Review details 2 Destination 2a Select name and folder 2b Select storage 2c Setup networks 3 Ready to complete 	OVF file Download size Size on disk Name Datastore Target Folder Disk storage Network mapping IP allocation	110.3.180.180/isharelovfivEMS_7.2.229.ova 7.6.GB 60.0.GB AudioCodes_EMS Netapp04.lun2 10.3.180.211 QASWDatacenter Thick Provision Lazy Zeroed VM Network 4 to VM Network Static - Manual, IPV4
	Power on after deploymen	t Back Next Frish Canc

10. In the Ready to Complete screen, ensure the the option 'Power on after deployment' is not selected, and click **Finish**.

	Deploying Audiocodes Element Management System Deploying disk 1 of 1 18 minutes remaining Close this dialog when completed	Cancel	
Recent Tasks Name Deploy OVF template	Target	Status 14%	Requested Start Time 21/05/2012 09:32:26
	Deployment Completed Successfully Deploying Audiocodes Element Management Syste Completed Successfully	em Close	

Figure 7-11: Deployment Progress Screen

11. Wait until deployment process has completed. This process may take approximately half an hour.

Status

Completed

7.1.2 Configuring the Virtual Machine Hardware Settings

ent Management Sys

This section shows how to configure the Virtual Machine's hardware settings.

Before starting this procedure, select the required values for your type of installation (high or low profile) and note them in the following table for reference. For the required VMware Disk Space allocation, CPU, and memory, see Chapter 3 on page 23.

Requested Start Time

21/05/2012 09:32:26

Start Time

21/05/2012 09:32:26

Completed Time

21/05/2012 10:06:12

Table 7-1: Virtual Machine Configuration

Required Parameter	Value
Disk size	Fill-in here
Memory size	Fill-in here
CPU cores	Fill-in here

Recent Tas

Deploy OVF template

Target

🔂 Audiocodes Ele

To configure the virtual machine hardware settings:

- 1. Before powering up the machine, go to the virtual machine Edit Settings option.
 - SSBC_01 SSBC_02 SSBC Actions - vEMS VCent Power vEMS Guest OS VEMS Snapshots 💕 Open Console 📥 Migrate.. Clone Template Fault Tolerance VM Policies Compatibility Export System Logs. 🛃 Edit Resource Settings 🦆 Edit Settings...

Figure 7-12: Edit Settings option

 In the CPU, Memory and Hardware tabs set the required values accordingly to the desired EMS server VMware Disk Space allocation. (See Chapter 3 on page 23), and then click OK.

1			
	• 0		
	- 18	•	
	: 08		
ogic Paratel			
letwork.			Connect
store ISO File		- 8	Connect.
t Device			Connect.
ity custom setting		-	
hedule VM Comp	atbility Upgra	de	
	ogic Paraller retwork store ISO File it Device ify custom setting hedule VM Comp	opic Parater Retwork store ISO File It Device Ify custom settings hedule VM Compatibility Upgra	Image: Comparison of the second se

Figure 7-13: CPU, Memory and Hard Disk Settings

Note:

•

- Once the hard disk space allocation is increased, it cannot be reduced to a lower amount.
- If you wish to create EMS VMs in a cluster environment supporting High Availability and you are using shared network storage, then ensure you provision a VM hard drive on the shared network storage on the cluster (see Section 7.1.4).

3. Wait until the machine reconfiguration process has completed.

Figure 7-14: Recent Tasks

Recent Tasks					
Name	Target	Status	Requested Start Time 🛛 🗢	Start Time	Completed Time
Reconfigure virtual machine	Audiocodes Element Management System	Completed	21/05/2012 11:03:39	21/05/2012 11:03:39	21/05/2012 11:03:41

7.1.3 Connecting EMS Server to Network

After installation, the EMS server is assigned a default IP address that will most likely be inaccessible from the customer's network. This address is assigned to the first virtual network interface card connected to the 'trusted' virtual network switch during the EMS server installation. You need to change this IP address to suit your IP addressing scheme

To assign EMS Server IP address to network:

 Power on the machine; in the vCenter tree, right-click the AudioCodes Element Management System and in the drop-down menu, choose Power > Power On. Upon the initial boot up after reconfiguring the disk space, the internal mechanism configures the server installation accordingly to version specifications (see Chapter 3 on page 23).



Low-7.2.2055	Actions - vEMS	Floppy drive 1 Power o
SSBC_02	Power Guest OS Snapshots Popen Console	Power On Power Off Suspend Reset
vEMS 7.2.1000	Migrate Clone Template	Shut Down Guest OS

- 2. Wait until the boot process has completed, and then connect the running server through the vSphere client console.
- 3. Login into the EMS server by SSH, as 'acems' user and enter *acems* password.
- 4. Switch to 'root' user and provide *root* password (default password is *root*):

su - root

- 5. Proceed to the network configuration using the Ems Server Manager. To run the manager type 'EmsServerManager', and then press Enter.
- 6. Set the EMS server network IP address as described in Section 17.1.
- Perform configuration actions as required using the EMS Server Manager (see Chapter 12 on page 103).

7.1.4 Configuring EMS Virtual Machines (VMs) in a VMware Cluster

This section describes how to configure EMS VMs in a VMware cluster.

7.1.4.1 Site Requirements

Ensure that your VM cluster site meets the following requirements:

- The configuration process assumes that you have a VMware cluster which contains at least two ESXi servers controlled by vCenter server.
- The clustered VM servers should be connected to a shared network storage of type iSCSI or any other types supported by VMware ESXi.

For example, a datastore "QASWDatacenter" which contains a cluster named "qaswCluster01" and is combined of two ESXi servers (see figure below).

Verify that Shared Storage is defined and mounted for all cluster members:

Navigator ¥	10.3.180.211 Actions -								
Hosts and Clusters F	Summary Monitor Manage	Related Objects							
asswycenter01.corp.audioc	Settings Networking Storage	Aarm Definitions Tags	Permissions Schedule	ed Taska Up	date Manager				
GASWDatacenter GastaCluster01	41	Storage Adapters							
10.3.180.211	Storage Adapters	- 0 - 0	Da-						Q Filler
10.3.160.212	Storage Devices	Adapter	7/04	Statute	Identifier	Tarpeta	Devices.	Paths	
0 7.2.2123	Host Cache Configuration	Patsburg 4 port SATA	IDE Controller						
A EMS 203-7 2 2123	Protocol Endpoints	💽 vmhba32	Block SCSI	Unknown		0	0	0	
High-7 2 2055		vmhba1	Block SCSI	Unknown		1	1	1	
Aligh217-LyDS-7.2.2110		Smart Array P420i							
Low-7.2.2055		I vmhba0	Block SCSI	Unknown		1	1	1	
SSBC_01		ISC SI Software Adapt	0f					1000 C	
10 3380_02 10 3380_02		Vmhba33	ISCSI	Online	ign 1998-01.com vmware:10.3.180.211	1	2	2	
A vCenter		Adapter Details							
WEMS 7 2 1000		Properties Devices Paths Targets Network Port Binding Advanced Options							
		Adopter Status							Disable
	Status Enabled								
		General							Edt.
		Name	vmhba33						
		Model	ISCSI Software Adapter						

Figure 7-16: Storage Adapters

Ensure that the 'Turn On vSphere HA' check box is selected:

Figure 7-17:Turn On vSphere HA





Ensure that HA is activated on each cluster node:

Figure 7-18: Activate HA on each Cluster Node

Navigator 📕	🗐 10.3.180.211 Actions 🔻	10.3.180.211 Actions -						
✓ Hosts and Clusters →	Summary Monitor Manag	e Related Objects						
Construction Constend Constend Construction Construction Constru	10.3.180. Type: Model Processo Logical P NCa: Virtual Me State: Uptime:	211 ESXI HP ProLlant DL360p Gen8 r Type: Intel(R) Xeon(R) CPU E5-26 coressors: 20 4 chines: 6 Connected 29 days	80 v2 @ 2.80GHz					
SSBC_01	▼ Hardware		 Configuration 					
SSBC_03	Manufacturer	HP	ESX/ESXi Version	VMware ESXi, 6.0.0, 3620759				
VCenter	Model	ProLiant DL360p Gen8	Image Profile HPE-ESXi-6.0.0-Update2-iso-600.9.5.0.48					
WEMS 7.2.1000	CPU	10 CPUs x 2.79 GHz	▶ vSphere HA State	 Running (Master) 				
	Memory	70,639 MB / 98,269 MB	 Fault Tolerance (Legacy) 	Unsupported				
	🕨 🔚 Virtual Flash Resource	0.00 B / 0.00 B	 Fault Tolerance 	Unsupported				
	Networking	localhost.corp.audiocodes.com	▶ EVC Mode	Intel® "Sandy Bridge" Generation				
	Storage	3 Datastore(s)	 Related Objects 					
	▶ Tags		Cluster 🗊 qaswCluste	r01				
	Update Manager Compl	iance 🗆		More Related Objects ش				

Ensure that the networking configuration is identical on each cluster node:

Figure 7-19: Networking

Navigator #	10.3.180.211 Actions *		#**
(Mosts and Clusters)	Summary Monitor Manage Relate	d Objects	
	Imposition Storage Aam * What What * Whene Adapters * Whene Adapters * Physical adapters TC-PID configuration Advanced Stand	Definitions Tags Permissions Scheduled Tasks Update Manager if switches	°

Ensure that the vMotion is enabled on each cluster node. The recommended method is to use a separate virtual switch for vMotion network (this should be defined in all cluster nodes and interconnected):

Navigator I	10.3.180.211 Actions -	10.3.180.211 Actions -										
• Hosts and Clusters / 🕤	Summary Monitor Manage	Related Objects										
Nevrgatur I •••••••••••••••••••••••••	10.3.180.211 Actions - Summary Monitor Manage Settings Networking Storage Vincai switches Vincernet adapters TCH/IP configuration Advanced	Related Objects Alarm Definitions, Tags, Permissions Weiternel adapters	Scheduled Tasis	P Address TI 10 3 100 211 C	er Offint Stack Default	Woolon Tryffic Enabled	Previoung Trafe Disabled	FT Looping Disatcled	Management Treffic Enabled	võphen Reskonen Taife Disabied	Q. Filer. Vorhen Restation MC Tarle Disabled	
		VLAN D Services 1 Enabled services 1 NC settings MAC address 3 MTU	vone (0) Motion traffic Vanagement traffic (8:80:23 afte0:88 (500									La La

Figure 7-20: Switch Properties

A VM will be movable and HA protected only when its hard disk is located on shared network storage on a cluster. You should choose an appropriate location for the VM hard disk when you deploy the EMS VM. If your configuration is performed correctly, a VM should be marked as "protected" as is shown in the figure below:

Navigator 4	CD Low-7.2.2055 Actions *						
Hosts and Clusters	Getting Started Summary Monito	or Manage Related Objects					
Image: Constraint of the second se	Powered On Launch Remote Console Download Remote Console	Low-7.2.2055 Guest OS: CentOS 4/5/6/7 (64-bit) Compatibility: ESXI 5.0 and later (VM vv VMw are Tools: Running, version:10246; DNS Name: VMw are-low PAddresses Host: 10.3.180.201 View all 3 Paddresses Host: 10.3.180.211	srsion 8) Current)				
High-7.2.2055							
High217-LyDS-7.2.2110	VM Hardware		VM Storage Policies				
Low-7.2.2055	 Advanced Configuration 		VM Storage Policies				
SSBC_02	 Notes 		VM Storage Policy Compliance				
SSBC_03	VM Failure Response		Check Com	anlianco			
VCenter	Failure	Failure response					
WEMS 7.2.1000	Host failure	Restart	▶ Tags				
	Host network isolation	Leave powered on	Related Objects				
	Datastore under PDL	Disabled	- vAnn Detaile				
	Datastore under APD	Disabled	Product				
	Guest not heartbeating	Ignore heartbeats	Version				
	vSphere HA Protect	tion: Protected 🚯	Vendor				
	Update Manager Compliance	Protect	he				
	Status 😵 Non-Compliant	vSphere w	Il attempt to restart the VM after supported failure.				
		Scan Detailed Status)				

Figure 7-19: Protected VM



Note: If you wish to manually migrate the EMS VMs to another cluster node, see Appendix D.

7.1.4.2 Cluster Host Node Failure

In case a host node where the VM is running fails, then the VM is restarted on the redundant cluster node automatically.



Note: When one of the cluster nodes fail, the EMS VM is automatically migrated to the redundant host node. During this process, the EMS VM is restarted and consequently any running EMS or SEM processes are dropped. The migration process may take several minutes.

7.2 Installing the EMS Server on Microsoft Hyper-V Platform

This section describes how to install the EMS server on the Microsoft Hyper-V Server 2012 R2 platform. This procedure takes approximately 30 minutes and predominantly depends on the hardware machine where the Microsoft Hyper-V platform is installed.



Note: The AudioCodes EMS supports the Failover Clustering feature in Windows Server 2012 R2 (see Chapter 3 on page 17).

The installation of the EMS server on Microsoft Hyper-V includes the following procedures:

- Install the Virtual Machine (VM) (see Section 7.2.1).
- Configure the Virtual machine hardware settings (see Section 7.2.2).
- Change MAC Addresses from 'Dynamic' to 'Static' (see Section 7.2.3).
- Connect EMS server to network (see Section 7.2.4).
- Configure VMs in a Microsoft Hyper-V cluster (see Section 7.2.5)

7.2.1 Installing the Microsoft Hyper-V Virtual Machine

The EMS server is distributed as a VM image (see Section 4.2 on page 30).

To install the EMS server on Microsoft Hyper-V:

- Extract the zip file containing the EMS server installation received from AudioCodes to a local directory on the Hyper-V server (see Appendix F on page 259 for instructions on how to transfer files).
- Open Hyper-V Manager by clicking Start > Administrative Tools > Hyper-V Manager; the following screen opens:

			Нур	ber-V Manager				
<u>F</u> ile <u>A</u> ction <u>¥</u> iew <u>H</u> elp								
🗢 🔿 🙍 🖬								
📑 Hyper-V Manager								Actions
WIN-VO01RE7B70M	Virtual Machines							WIN-VO01RE7B70M -
	Name	State	CPU Usage	Assigned Memory	Uptime	Status		New 🕨
	SSBC_AlexR3_HA1	Running	7%	4128 MB	20:17:00			💫 Import Virtual Machine
								🖆 Hyper-V Settings
								👯 Virtual Switch Manager
								🔬 Virtual SAN Manager
								💋 Edit Disk
							/	🖳 Inspect Disk
	<u>C</u> heckpoints							Stop Service
			No vitu	al machine selected				🗙 Remove Server
			110 1110	a maonine selected.				🔉 Refresh
								View 🕨
								? Help
	Details							
			N	o item selected.				
							Activ	ate Windows
							Go to	System in Control Panel to activa

Figure 7-210: Installing the EMS server on Hyper-V – Hyper-V Manager



 Start the Import Virtual Machine wizard: click the Action tab, and then select Import Virtual Machine from the menu; the Import Virtual Machine screen shown below opens:

Figure 7-221: Installing EMS server on Hyper-V – Import Virtual Machine Wizard

	Import Virtual Machine						
Before You B	Begin						
Before You Begin Locate Folder Select Virtual Machine Choose Import Type Summary	This wizard helps you import a virtual machine from a set of configuration files. It guides you through resolving configuration problems to prepare the virtual machine for use on this computer.						
	< Previous Next > Einish Cancel						

4. Click **Next**; the Locate Folder screen opens:

	Import Virtual Machine	x
Locate Folder	r	
Before You Begin	Specify the folder containing the virtual machine to import.	
Locate Folder Select Virtual Machine Choose Import Type Summary	Folder: C:\Export_115\EMS_test\ Brows	e
	< Previous Next > Finish Can	cel

Figure 7-232: Installing EMS server on Hyper-V – Locate Folder

- 5. Enter the location of the VM installation folder, which was previously extracted, from the zip file as shown in the figure above, and then click **Next**; the Select Virtual Machine screen opens.
- 6. Select the virtual machine to import, and then click **Next**; the Choose Import Type screen opens:

Figure 7-243: Installing EMS server on Hyper-V – Choose Import Type

2	Import Virtual Machine	
Choose Import Type		
Before You Begin Locate Folder Select Virtual Machine Choose Import Type Choose Destination Choose Storage Folders Summary	Choose the type of import to perform: Register the virtual machine in-place (use the existing unique ID) Restore the virtual machine (use the existing unique ID) Octopy the virtual machine (create a new unique ID)	
	< Previous Next > Einish Cancel	

7. Select the option "Copy the virtual machine (create a new unique ID)", and then click **Next**; the Choose Folders for Virtual Machine Files screen opens:

Figure 7-254: Installing EMS server on Hyper-V – Choose Destination

2	Import Virtual Machine	x
Choose Folders for Virtual Machine Files		
Before You Begin Locate Folder Select Virtual Machine Choose Import Type Choose Destination	You can specify new or existing folders to store the virtual machine files. Otherwise, the wizard imports the files to default Hyper-V folders on this computer, or to folders specified in the virtual machine configuration.	
Choose Storage Folders Summary	C:\ProgramData\Microsoft\Windows\Hyper-V\ Browse. Checkpoint store: C:\ProgramData\Microsoft\Windows\Hyper-V\ Browse. Smart Paging folder: C:\ProgramData\Microsoft\Windows\Hyper-V\ Browse. C:\ProgramData\Microsoft\Windows\Hyper-V\ Browse. Browse.	
< Previous Next > Einish Cancel		



8. Select the location of the virtual hard disk, and then click **Next**; the Choose Storage Folders screen opens:

Figure 7-265: Installing EMS server on Hyper-V – Choose Storage Folders

2	Import Virtual Machine
Choose Folde	ers to Store Virtual Hard Disks
Before You Begin Locate Folder Select Virtual Machine Choose Import Type Choose Destination Choose Storage Folders Summary	Where do you want to store the imported virtual hard disks for this virtual machine? Location: C:\Users\Public\Documents\Hyper-V\Virtual Hard Disks\ Browse
	<pre></pre>

- **9.** Select the Storage Folder for the Virtual Hard Disk, and then click **Next**; the Summary screen opens.
- **10.** Click **Finish** to start the creation of the VM; a similar installation progress indicator is shown:

Figure 7-276: File Copy Progress Bar



This step may take approximately 30 minutes to complete.

11. Proceed to Section 7.2.2 on page 65.

7.2.2 Configuring the Virtual Machine Hardware Settings

This section shows how to configure the Virtual Machine's hardware settings.

Before starting this procedure, select the required values for your type of installation (high or low profile) and note them in the following table for reference. For the required VMware Disk Space allocation, CPU, and memory, see Chapter 3 on page 23.

Required Parameter	Value
Disk size	Fillhere
Memory size	Fill-in here
CPU cores	Fill-in here

Table 7-2: Virtual Machine Configuration

> To configure the VM for EMS server:

1. Locate the new EMS server VM in the tree in the Hyper-V Manager, right-click it, and then select **Settings**; the Virtual Machine Settings screen opens:

Figure 7-287: Adjusting VM for EMS server – Settings - Memory

MS-QA	✓ 4 ▶ Q.
Hardware M Add Hardware BloS Boot from CD Memory 4096 MB Processor 6 Virtual processors IDE Controller 0 IDE Controller 1 DD Drive None Virtual Switch 1 COM 1 None COM 2 None Diskette Drive None Management Name EMS-QA EMstep Drive None Smart Paging File Location C:\ClusterStorage\Volume 1[EM] None None Naturatic Start Action None	Image: Memory You can configure options for assigning and managing memory for this virtual machine. Specify the amount of memory that this virtual machine will be started with. Startup RAM: 4096 MB Dynamic Memory You can manage the amount of memory assigned to this virtual machine dynamically within the specified range. Enable Dynamic Memory Minimum RAM: 512 Maximum RAM: 1048576 Maximum RAM: 1048576 Specify the percentage of memory that Hyper-V should try to reserve as a buffer. Hyper-V uses the percentage and the current demand for memory to determine an amount of memory for the buffer. Memory weight Specify how to prioritize the availability of memory for this virtual machine compared to other virtual machines on this computer. Low High Image: Specifying a lower setting for this virtual machine might prevent it from starting when other virtual machines are running and available memory is low.

AudioCodes

- 2. In the Hardware pane, select **Memory**, as shown above, enter the 'Startup RAM' parameter as required, and then click **Apply**.
- 3. In the Hardware pane, select **Processor**; the Processor screen shown in the figure below opens.

-	
EMS_QA_High	
 ★ Hardware ▲ Add Hardware ▲ BIOS Boot from CD ■ Memory 20000 MB ● Processor 6 Virtual processors ■ IDE Controller 0 ● Africk Add Trive EMS_QA_High.vhdx ■ IDE Controller 1 ● DVD Drive None SCSI Controller ■ Network Adapter Virtual Switch 1 ♥ COM 1 None 	▶ Processor You can modify the number of virtual processors based on the number of processors on the physical computer. You can also modify other resource control settings. Number of virtual processors: 6 → Resource control 6 → You can use resource controls to balance resources among virtual machines. Virtual machine geserve (percentage): 100 Percent of total system resources: 37 Virtual machine limit (percentage): 100 Percent of total system resources: 37 Relative weight: 100 Image: Some settings cannot be modified because the virtual machine was running when the view of the
 COM 2 None Diskette Drive None Management Name EMS_QA_High Integration Services Some services offered Checkpoint File Location C:\ClusterStorage\volume1\EM Smart Paging File Location C:\ClusterStorage\volume1\EM Automatic Start Action None 	this window was opened. To modify a setting that is unavailable, shut down the virtual machine and then reopen this window.
	<u>Q</u> K <u>Cancel</u> Apply

Figure 7-298: Adjusting VM for EMS server - Settings - Processor

- 4. Set the 'Number of virtual processors' parameters as required.
- 5. Set the 'Virtual machine reserve (percentage)' parameter to **100%**, and then click **Apply**.



Note:

- Once the hard disk space allocation is increased, it cannot be reduced.
- If you wish to create EMS VMs in a Cluster environment that supports High Availability and you are using shared network storage, then ensure you provision a VM hard drive on the shared network storage on the cluster (see Section 7.2.5).

7.2.2.1 Expanding Disk Capacity

The EMS server virtual disk is provisioned by default with a minimum volume. In case a higher capacity is required for the target EMS server then the disk can be expanded.

> To expand the disk size:

- 1. Make sure that the target EMS server VM is not running Off state.
- 2. Select the Hard Drive, and then click Edit.

1	Settings for EMS_test-new on QAHYPERV1	x
EMS_test-new	✓ ◀ ▶ Q	
★ Hardware M Add Hardware Image: Soot from CD Image: Some services Image: Some services offered Image: Some services o	 Hard Drive You can change how this virtual hard disk is attached to the virtual machine. If an operating system is installed on this disk, changing the attachment might prevent the virtual machine from starting. Controller: Location: IDE Controller 0 (0 (in use)) Media You can compact, convert, expand, merge, reconnect or shrink a virtual hard disk by editing the associated file. Specify the full path to the file. Virtual hard disk: C: Users \Public\Documents \Hyper-V\Virtual Hard Disks \EMS_test.vhdx New Edit Inspect Browse Physical hard disk: Disk 1 1.00 GB Bus 0 Lun 0 Target 0 If the physical hard disk you want to use is not listed, make sure that the disk is offline. Use Disk Management on the physical computer to manage physical hard disks. To remove the virtual hard disk, dick Remove. This disconnects the disk but does not delete the associated file.	
	OK Cancel Apply	

Figure 7-300: Expanding Disk Capacity

The Edit Virtual Disk Wizard is displayed as shown below.

Figure 7-311: Edit Virtual Hard Disk Wizard

ø	Edit Virtual Hard Disk Wizard
Locate Virt	ual Hard Disk
Before You Begin Locate Disk Choose Action Summary	 Where is the virtual hard disk file located? Location: C:\Users\Public\Documents\Hyper-V\Virtual Hard Disks\EMS_test.vhdx Browse A Editing the following types of virtual hard disks might result in data loss: Virtual hard disks in a differencing disk chain that have child virtual hard disks associated with them. Virtual hard disks (.avhd/.avhdx) associated with virtual machine checkpoints. Virtual hard disks associated with a virtual machine that has replication enabled and is currently involved in initial replication, resynchronization, test failover, or failover.
	< Previous Next > Finish Cancel

3. Click **Next**; the Choose Action screen is displayed:

Figure 7-32: Edit Virtual Hard Disk Wizard-Choose Action

💋 Edit Virtual Hard Disk Wizard 🗙	
Choose Action	on
Before You Begin Locate Disk Choose Action Configure Disk Summary	 What do you want to do to the virtual hard disk? Compact This option compacts the file size of a virtual hard disk. The storage capacity of the virtual hard disk remains the same. Convert This option converts a virtual hard disk by copying the contents to a new virtual hard disk. The new virtual hard disk can use a different type and format than the original virtual hard disk. Expand This option expands the capacity of the virtual hard disk.
	< Previous Next > Finish Cancel

4. Select the **Expand** option, and then click **Next**; the Expand Virtual Hard Disk screen opens.

Figure 7-33: Edit Virtual Hard Disk Wizard-Expand Virtual Hard Disk

ø	Edit Virtual Hard Disk Wizard	x
Expand Virtu	al Hard Disk	
Before You Begin Locate Disk Choose Action Configure Disk Summary	What size do you want to make the virtual hard disk? Current size is 170 GB. New size: 300 GB (Maximum: 64 TB)	
	< Previous Next > Finish Cancel	

5. Enter the required size for the disk, and then click **Next**; the Summary screen is displayed.

Figure 7-34: Edit Virtual Hard Disk Wizard-Completion

ø	Edit Virtual Hard Disk Wizard
Completing t	he Edit Virtual Hard Disk Wizard
Before You Begin Locate Disk Choose Action	You have successfully completed the Edit Virtual Hard Disk Wizard. You are about to make the following changes. Description:
Configure Disk Summary	Virtual Hard Disk: EMS_test.vhdx (VHDX, dynamically expanding) Action: Expand Configuration: New virtual disk size: 300 GB To complete the action and close the wizard, click Finish.
	< Previous Next > Finish Cancel

- 6. Verify that all of the parameters have been configured, and then click **Finish**. The settings window will be displayed.
- 7. Click OK to close.

7.2.3 Changing MAC Addresses from 'Dynamic' to 'Static'

By default, the MAC addresses of the EMS server Virtual Machine are set dynamically by the hypervisor. Consequently, they might be changed under certain circumstances, for example, after moving the VM between Hyper-V hosts. Changing the MAC address may lead to an invalid license for features such as the SEM.

To prevent this from occurring, MAC Addresses should be changed from 'Dynamic' to 'Static'.

To change the MAC address to 'Static' in Microsoft Hyper-V:

- 1. Shutdown the EMS server (see Section 16.6 on page 121).
- 2. In the Hardware pane, select Network Adapter and then Advanced Features.
- 3. Select the MAC address 'Static' option.
- 4. Repeat steps 2 and 3 for each network adapter.

Figure 7-35: Advanced Features - Network Adapter – Static MAC Address

Settings for EMS-QA on QAHYPERV1							
EMS-QA	~	▲ ▶ Q.					
★ Hardware ▲ ▲ ▲ ■	^	Advanced Features MAC address Dynamic Static 00 - 15 - 5D - 5E - 73 - 1B MAC address spoofing allows virtual machines to change the source MAC address in outgoing packets to one that is not assigned to them. Enable MAC address spoofing DHCP guard DHCP guard drops DHCP server messages from unauthorized virtual machines pretending to be DHCP servers.	< _				
Virtual Switch 1 Hardware Acceleration Advanced Features COM 1 None COM 2 None Diskette Drive None	≡	≡	Enable DHCP guard Router guard Router guard Router guard drops router advertisement and redirection messages from unauthorized virtual machines pretending to be routers. Enable router advertisement guard Destended extende				
Management IName EMS-QA Integration Services Some services offered Checkpoint File Location C:\ClusterStorage\Volume1\EM Smart Paging File Location C:\ClusterStorage\Volume1\EM Automatic Start Action	~	Protected network Move this virtual machine to another duster node if a network disconnection is detected. Protected network Port mirroring Port mirroring allows the network traffic of a virtual machine to be monitored by copying incoming and outgoing packets and forwarding the copies to another virtual machine configured for monitoring. Mirroring mode: None	~				
		<u>OK</u> <u>Cancel</u> <u>Apply</u>					

7.2.4 Connecting EMS Server to Network

After installation, the EMS server is assigned a default IP address that will most likely be inaccessible from the customer's network. This address is assigned to the first virtual network interface card connected to the 'trusted' virtual network switch during the EMS server installation. You need to change this IP address to suit your IP addressing scheme.

To reconfigure the EMS server IP address:

1. Start the EMS server virtual machine, on the Hyper-V tree, right-click the EMS server, and then in the drop-down menu, choose **Start**.

V <u>i</u> rtual Machines										
Name	State	CPU Usage	e Assigned Memory	l						
Stress_tool SSBC_AlexR3_HA1 SSBC_AlexR2_HA2 SSBC_AlexR2_HA1	Running Off Off Off	0 %	2048 MB	1						
ESBC_alexr1	Running	0 %	2048 MB	1						
EMS-QA	Off	Conr	Connect							
EMS_QA_High	Running	Setti	ngs	1						
		Start	Start							
		Chec	:kpoint							
<		Mov	e							

Figure 7-36: Power On Virtual Machine

2. Connect to the console of the running server by right-clicking the EMS server virtual machine, and then in the drop-down menu, choose **Connect**.

Figure 7-37: Connect to EMS Server Console

V <u>i</u> rtual Machines										
Name	State	CPU Usage	Assigned Memory	Uptime						
Stress_tool	Running	0 %	2048 MB	1.04:34:22						
SSBC_AlexR3_HA1	Off									
SSBC_AlexR2_HA2	Off	Off								
SSBC_AlexR2_HA1	Off									
ESBC_alexr1	Running	0 %	2048 MB	1.04:10:46						
EMS-QA	Off									
EMS_QA_High	Running	0.9/	20000 MD	1.02:37:53						
		Connect								
		Settings								
		Turn Off								
<		Shut Down								

3. Login into the EMS server by SSH, as 'acems' user and enter password acems.
4. Switch to 'root' user and provide *root* password (default password is *root*):

su - root

- Start the EMS Server Manager utility by specifying the following command:
 # EmsServerManager
- 6. Set the EMS server network IP address to suit your IP addressing scheme (see Section 17.1).
- Perform other configuration actions as required using the EMS Server Manager (see Chapter 12 on page 103).

7.2.5 Configuring EMS Virtual Machines in a Microsoft Hyper-V Cluster

This section describes how to configure EMS VMs in a Microsoft Hyper-V cluster for HA.

7.2.5.1 Site Requirements

Ensure that your Hyper-V cluster site meets the following requirements:

- The configuration process assumes that your Hyper-V failover cluster contains at least two Windows nodes with installed Hyper-V service.
- The cluster should be connected to a shared network storage of iSCSI type or any other supported type. For example, "QAHyperv" contains two nodes.

Figure 7-38: Hyper-V-Failover Cluster Manager Nodes

		Fail	over Cluster Manag	ger						
<u>File Action View H</u> el	р									
🗢 🌩 🖄 🖬 🖬 🖬										
📲 Failover Cluster Manage	Nodes (2)									
⊿ 10 QAHyperv-Cl.corp.a ■ Roles	Search	Search P Queries 🔻 🖬 👻 📀								
Nodes	Name	Status	Assigned Vote	Current Vote	Information					
🗚 📇 Storage	R QAHyperV1	🛞 Up	1	1	10000000					
Disks Pools Networks	a QAHyperv2	🛞 Up	1	1						
	<				>					
	*									
< III >										

AudioCodes

The EMS VM should be created with a hard drive which is situated on a shared cluster storage.

7.2.5.2 Add the EMS VM in Failover Cluster Manager

After you create the new EMS VM, you should add the VM to a cluster role in the Failover Cluster Manager.

To add the EMS VM in Failover Cluster Manager:

1. Right-click "Roles" and in the pop up menu, choose **Configure Role**:

Figure 7-39: Configure Role

Poler (2)					
Polor (2)					
Polos (2)					
Search					P Queries ▼ ▼
Failover Cluster Manager W Help Image Cluster Manager Image Cluster Cluster Manager Image Cluster Cluster Manager Image Cluster Manager	Information				
Machines +	Running	Virtual Machine	QAHyperV1	Priority Information Medium Medium	
Create Empty Role	Failover Cluster Manager P Queries Queries				
•					
h					
	Searco ure Role) Empty Role ;	Search ure Role Status Machines Empty Role Machines	Search ure Role Status Type Machines Empty Role	Search Status Type Owner Node Machines	Search Status Type Owner Node Priority Machines + Running Virtual Machine QAHyperV1 Medium Running Virtual Machine QAHyperV2 Medium Medium Image: Search Medium Image: Search Medium Medium

2. In the Select Role window, select the Virtual Machine option and then click Next.

Figure 7-40: Choose Virtual Machine

灎			Failover Clu	ister Manager			
Eile Action View Hel	lp						
🗢 🔿 🖄 🖬 📓 🖬							
Failover Cluster Manage A OAHyperv-Cl.corp.	Roles (2)						
Roles	Search	0.1	T	Querry March	Die 1	P Quenes ▼	<u> • •</u>
Jan Prodes ⊿ 🛃 Storage	Name	Status	lich Aveila	Owner Node	Priority	Information	
E Pools	- 2 0/		High Availa	onity wizard			
Networks	Select Ro	le					
Cluster Events							
	Before You Begin Select Role Confirmation Configure High Availability Summary	Select the role th Generic Serv GSCSI Target SISS Server Message Qu Contex Server	at you want to configur ice ica Broker Server suing ne	e for high availability:	Description: A virtual machin computer syste computer. Multi run on one com	re is a vitualized m running on a physical ple vitual machines can puter.	y node
< III >				< Pr	evious Nei	kt> Cancel	~

A list of available VMs are displayed; you should find the your new created EMS VM:

			Failover Clu	uster Manag	ger	
Eile Action View Help	p					
 Failover Cluster Manage QAHyperv-Cl.corp.a Roles Nodes Storage Disks Pools Networks Cluster Events 	Roles (2) Search Name	Status Tual Machine	Type High Availat	Owner No Dility Wizard	ode Priority	P Queries V
	Failover Cluster M ew Help rt Manage Cluster M rt M Rt Manage Cluster M rt M R	t to configure fo	r high availability, Host Server QAHyperV1.corp.au	diocodes.com Refresh		
< <u> </u>	_			[< <u>P</u> revious <u>N</u> ex	t> Cancel

Figure 7-41: Confirm Virtual Machine

Select the check box, and then click Next.
 At the end of configuration process you should see the following:

Figure 7-42: Virtual Machine Successfully Added

\$ 7	High A	Availability Wizard		x
🌄 Summary				
Before You Begin Select Role	High availability was succ	cessfully configured for the role.		
Select Virtual Machine Confirmation Configure High Availability	Virtual Mac	chine	ccentully made highly	^
Summary	available.	oningurations chosen were sur	ccessiony made nighty	
	Name EMS	Result	Success	
	To view the report created by the	wizard click View Report.	View Beng	₩ vt
	To close this wizard, click Finish.		Tiew Hebe	
			<u> </u>	sh

 Click Finish to confirm your choice. Now your EMS VM is protected by the Windows High Availability Cluster mechanism.



Note: If you wish to manually move the EMS VMs to another cluster node, see Appendix D on page 235.

7.2.5.3 Cluster Host Node Failure

In case a host node where the VM is running fails, then the VM is restarted on the redundant cluster host node automatically.



Note: When one of the cluster hosts fails, the EMS VM is automatically moved to the redundant server host node. During this process, the EMS VM is restarted and consequently any running EMS or SEM processes are dropped. The move process may take several minutes.



This page is intentionally left blank.

Part III

EMS Server Upgrade

This part describes the upgrade of the EMS server on dedicated hardware and on the VMware platform.

8

Upgrading the EMS Server on Dedicated Hardware

This section describes the upgrade of the EMS server on dedicated hardware.



Important: Prior to performing the upgrade, it is highly recommended to perform a complete backup of the EMS server. For more information, see Appendix A on page 215.

You can perform the EMS version upgrade using AudioCodes supplied DVD3.

For EMS versions 2.2 until version 6.6:

A major version upgrade of the EMS from the above versions is not supported. Instead, users must perform a full installation of version 7.0 as described in Section 6 on page 35.

8.1 Upgrading the EMS Server-DVD

This section describes how to upgrade the EMS server from the AudioCodes supplied installation DVD on the Linux platform.

To upgrade the EMS server on the Linux platform to version 7.2, only DVD3 is required. Verify in the EMS Manager 'General Info' screen that you have installed the latest Linux revision (see Chapter 3 on page 23). If you have an older OS revision, a clean installation must be performed using all three DVDs (see Section 6.2 on page 37).



Note: Before starting the installation, it is highly recommended to configure the SSH client (e.g. Putty application) to save the session output into a log file.

> To upgrade the EMS server on the Linux platform:

- 1. Insert DVD3-EMS Server Application Installation into the DVD ROM.
- 2. Login into the EMS server by SSH, as 'acems' user and enter password *acems* (or customer defined password).
- 3. Switch to 'root' user and provide root password (default password is root):

su - root

 On some machines you need to mount the CDROM in order to make it available: mount /misc/cd



5. Run the installation script from its location:

cd /misc/cd/EmsServerInstall/ ./install

Figure 8-1: EMS Server Upgrade (Linux)



6. Enter **y**, and then press Enter to accept the License agreement.

Figure 8-2: EMS Server Upgrade (Linux) – License Agreement



7. OS patches are installed.

After the OS patches installation, you are prompted to press Enter to reboot.



Note: This step is optional and depends upon which version you are upgrading. After the EMS server has rebooted, repeat steps 2 to 6.

8. If the EMS version you are upgrading to is packaged with a later version of Java than the one that is currently installed, type **yes**, and then press Enter to upgrade the Java version, otherwise, skip this step:

```
Java DB version 10.4.2.1.1 is currently installed.
Upgrade to version 10.6.2.1.1 ? [yes,no]yes
```

9. At the end of Java installation, press Enter to continue.

Figure 8-3: EMS Server Application Upgrade (Linux) - Java Installation



10. Wait for the installation to complete and reboot the EMS server.



Figure 8-4: EMS Server Upgrade (Linux) Complete



Note: For SEM Users: each time the EMS server version is upgraded, the operator should perform CTRL – F5 (refresh) action on the SEM Page, and then relogin to the application.

AudioCodes

8.2 Upgrading the EMS Server-ISO File

This section describes how to upgrade the EMS server using an ISO file. Before performing this procedure, you need to verify the ISO file contents (see Section 6.1.2).

To upgrade using an ISO file:

- 1. Use SFTP or SCP to copy the iso file to /home/acems in the server
- 2. Replace "7.2.xxx" in the filename with the relevant version in two of the following commands.

```
mkdir /ins
cp ~acems/DVD3_EMS_7.2.xxx.iso /ins
mkdir /tmp/cd
umount -1 /tmp/cd
mount -t iso9660 -o loop,ro /ins/DVD3_EMS_7.2.xxx.iso
/tmp/cd
cd /tmp/cd/EmsServerInstall
```

3. Run the installation script from its location:

./install





4. Proceed to step 6 in Section 8.1.

9

Upgrading the EMS Server on the VMware Platform

This section describes how to upgrade the EMS server on the VMware platform. This can be performed by running the Upgrade media (CD/DVD or ISO file) using either the VMware Remote Console Application (VMRC) or the VMware Server Host.

The following steps must be performed:

- Step 1: Setting up the VMware vSphere Web Client (see Section 9.1)
- Step 2: Running the VMware upgrade script (see Section 9.2)

Note:



- A Remote connection to the VMware host is established using the VMware Remote Console application (VMRC). You must download this application or use a pre-installed remote connection client to connect to the remote host.
- The procedures below show screen examples of the vSphere Web Client. However, you should refer to the VMware documentation for more information.

9.1 Step 1: Setting Up the VMware vSphere Web Client

This section describes how to setup the VMware vSphere Web Client.

> To upgrade the EMS server on the VMware platform:

- 1. Place the media **DVD3-EMS Server Application Installation** into the DVD/CD disk drive or if you are using an ISO file to the desired directory on either the machine where the VMware vSphere Web client is installed or on the VMware server host.
- 2. Login to the VMware vSphere Web client.



vmware [®] vSphere Web Cli	ent n≜≘			ပီ I vmwa	re@QASWVCENT		I Help 👻 I 🔍 🤤	Search	-
Navigator I	付 Home							🔯 Alarms	Ψ×
Hosts and Clusters	Home							All (1) Ne	Ack
🕺 Home	Inventories						•	🔶 qaswvcenter0	1.cor
🕞 vCenter Inventory Lists 🔰		_				_	-	Performance C	hart
Hosts and Clusters			(G)		<u> </u>				
🕑 VMs and Templates 🔰 📏	vCenter H	losts and	VMs and	Storage	Networking	Cont	ent		
Storage > ::	Inventory Lists	Clusters	Templates			Libra	ries	Mark In Dear	
Q Networking					~				e 🗕
Policies and Profiles		O			G			UP VEMIS-Edi	
🍘 Hybrid Cloud Manager 💦 📏		$\mathbf{\vee}$							
💙 vRealize Orchestrator 🔰 🔉	Hybrid Cloud Manager Oi	vRealize rchestrator							
🚳 Update Manager 🔰 🔪									
🖏 Administration 🛛 📏	Monitoring								
🗊 Tasks 🖉	Watch How-to Vid	leos	6				<u>.</u> -		
🛱 Recent Taska									
								1	+^
Task Name	Target	Status		Initiator	Queued For		Start Time	Completion Time	Se
Reconfigure virtual machine	M VEMS-7.2.3030-10	 Completed 		vmware		19 ms	11/17/2016 10:03:1	11/17/2016 10:03:	2 qa
4									Þ

Figure 9-1: VMware vSphere Web Client

3. In the vCenter Navigator, select **Hosts and Clusters**. A list of Hosts and Clusters is displayed.

vmware [®] vSphere Web Cli	ent n≜≣		ບ I vmware@0	QASVVVCENTER01 -	l Help	🗸 I 🔍 Search
Navigator I	🗊 qaswCluster01	Actions 👻			=*	🔯 Alarms 🛛 🖡 🗙
Home Image: Construction of the second	Summary Monitor	Manage Related Objects asswChuster01 Total Processors: 20 Total Vibition Migrations: 48 Total Vibition Migration: 48 Total Vibition Migrations: 48 Total Vib	CPU USED: 14.07 GHz MEMORY USED: 128.05 GB STORAGE USED: 2.47 TB	CAPACITY: 55.86 FREE: 63.83 CAPACITY: 191.93 FREE: 4.20 CAPACITY: 0.67 Category P This list is empty.	GH2 GB GB 17B 77B 31 escoli	All (1) New (1) Acknowl qaswcenter01.corp.audiocodes Performance Charts Service He work In Progress vEMS - Edit Resource S
😨 Recent Tasks						Į×
Task Name	Target	Status	Initiator	Queued For	Start Tim	e Completion Time

Figure 9-2: Hosts and Clusters

4. Right-click the AudioCodes EMS node that you wish to upgrade and choose the **Edit Settings** option.



Figure 9-3: Edit Settings Option

The vCenter Edit Settings screen is displayed.

Figure 9-4: Connection Options

VCenter - Edit Settings Virtual Hardware VM Options SDRS Rules VApp Options VIrtual Hardware VM Options SDRS Rules VApp Options Virtual Hardware VM Options SDRS Rules Vapp Options Vapp Options Virtual Hardware VM Options SDRS Rules Vapp Options Vapp Options	(?) ₩						
Virtual Hardware VM (Options S	DRS Rules	vApp Opti	ons			
F 🔲 CPU	4		• 0				
▶ 🏧 Memory	10240		▼ MB	-			
▶ 🚐 Hard disk 1	150		▲ ▼ GB	-)		
▶ 🛃 SCSI controller 0	LSI Logic	SAS					
🕨 🎫 Network adapter 1	VM Netw	ork		•	🗹 Conne	ected	
▶ i CD/DVD drive 1	Client De	vice		-	🔲 🗌 Conne	ected	\otimes
🕨 📩 Floppy drive 1	Client De	vice			Conne	ected	
🕨 🛄 Video card	Host Dev	ice IRO Eilo					
🕨 🌼 VMCI device	Datastore	; 180 File					
 Other Devices 							
▶ Upgrade	Sched	ule VM Comp	atibility Up <u>c</u>	rade			
New device:		Select		-	Add		
Compatibility: ESXi 5.0 ai	nd later (VM	version 8)				ОК	Cancel

AudioCodes

- 5. In the Virtual Hardware tab, select the CD/DVD drive item, and from the drop-down list, select the relevant option according to where you placed the Upgrade Media (CD/DVD or ISO image file):
 - **Client Device:** This option enables you to run the upgrade from the PC running the remote console (see Section 9.1.1).
 - **Host Device:** This option enables you to run the upgrade from the CD/DVD drive of the VMware server host (see Section 9.1.2).
 - **Datastore ISO file:** This option enables you to run the upgrade from the image file on the storage device of the VMware server host. When you choose this option, browse to the location of the ISO file on the VMware storage device (see Section 9.1.2).

9.1.1 Setting up Upgrade Using VMware Remote Console Application (VMRC)

This section describes how to run the upgrade from the VMware host. This procedure requires connecting to the VMware host using the VMware Remote Console application (VMRC).

To run the upgrade using VMRC:

 In the Manage tab under Settings> VM Hardware, select the Help icon adjacent to the CD/DVD drive item and then from the pop-up, click the Launch Remote Console to launch the VMware Remote Console application (VMRC). If necessary, click the Download Remote Console link to download this application.



Note: If you already have a remote console application installed on your machine, you can use your pre-installed application.

🕞 vEMS-7.2.3030-10.3.180.218 Actions 👻 =-👩 Alarms Getting Started Summary Monitor Manage Related Objects All (1) Ne 🔶 qaswvce Settings Alarm Definitions Tags Permissions Policies Scheduled Tasks Update Manager Perform: VM Hardware Edit... VM Hardware ► CPU 1 CPU(s), 195 MHz used VM Options 8192 MB, 1146 MB memory active ▶ Memory Hard disk 1 📝 Work In VM SDRS Rules 500.00 GB 🖆 vems-e vApp Options Network adapter 1 VM Network (connected) Guest User Mappings OD/DVD drive 1 Disconnected 6 - الاو 📑 Floppy drive 1 🔹 Disconnected \mathbf{x} Help Video card 8.00 MB Use the external VMware Remote Console Additional Hardware ▶ Other application (VMRC) to connect client devices to this virtual machine Launch Remote Console Download Remote Console Target Status Initiator Queued For

Figure 9-5: Help Link to Launch Remote Console

The remote console application is displayed.

Figure 9-6: Remote Console Application

*	VEMS-7.2.3030-10.3.180.	218 - VMware Remot	e Cons	ole		
<u>v</u>	MRC 🕶 📔 👻 🖶	凵				*
C) Power	•				
0	Removable Devices	•				
d	Send Ctrl+Alt+Del					
٩	Manage	•	•	Reinstall VMware Tools		
Ţ	🖞 Full Screen	Ctrl+Alt+Enter	P	Message Log		
	Preferences		Ş	Virtual Machine Settings	Ctrl +D	
	Help	•				
	Exit					

 In the toolbar, from the VMRC drop-down list, choose Manage > Virtual Machine Settings. The Virtual Machine Settings screen is displayed:

Figure 9-7: Virtual Machine Settings

Virtual Machine Setting	js		×
Hardware Options			
Device Memory Processors Hard disk 1 CD/DVD drive 1 Floppy drive 1 Network adapt Video card	Summary 10 GB 4 150 GB (Preallocated) Using local drive VM Network 1 monitor	Device status	
	Add <u>R</u> emove]	
		OK Cancel Help	

3. From the Location drop-down list, select Local Client.

AudioCodes

- 4. Select the CD/DVD drive item and then choose one of the following:
 - Use physical drive: from the drop-down list, select the CD/DVD drive where you placed the Upgrade media.
 - Use ISO image file: browse to the location of the ISO image file.
- 5. Click OK.
- 6. Proceed to Section 9.2.

9.1.2 Setting up Upgrade Using VMware Server Host

This section describes how to run the upgrade using the VMware server host.

> To run the upgrade using the VMware Server host:

- 1. Select the **Manage** tab, right-click the Connect icon and select one of the following options:
 - Connect to host CD device
 - Connect to CD/DVD image on a datastore

Figure 9-8: Connect to Host CD Device/ Datastore ISO file

CD/DVD drive 1	Disconnected	—	• [¢] ∗ 0	
📊 Floppy drive 1	Disconnected	1	Connect to he	ost CD device
 Video card 	4.00 MB		Connect to C	D/DVD image on a datastore
▶ Other	Additional Hardwa	re	-	
Compatibility	ESXi 5.0 and later	(VM version 8)		

2. Wait until the machine reconfiguration has completed, and then verify that the 'Connected' status is displayed:

Figure 9-9: CD/DVD Drive - Connected Status

0	Alexen Defeitions	T	Demainaiana	Delision	Ocheckular	Teele	Lin data 1		
Settings	Alarm Definitions	Tags	Permissions	Policies	Scheduled	Tasks	Update	Manag	jer
44			VM Hardware					E	dit
VM Har	dware		▶ CPU	1	CPU(s), 19	5 MHz us	sed		
VM Opt	tions		Memory		8192 MB,	12 MB, 1064 MB memory active			
VM SDI	RS Rules		Hard disk 1	5	00.00 GB				
Guest I	Iser Mannings		 Network ada 	pter 1 V	M Network	(connec	ted)		
Guest	soor mappingo		▶ 🎯 CD/DVD	drive 1 C	onnected			ø -	0
			금 Floppy dr	ive 1 D	isconnecteo	ł		ø-	0
			 Video card 	8	.00 MB				
			▶ Other	A	dditional Ha	rdware			

3. Proceed to Section 9.2.

9.2 Step 2: Running the VMware Upgrade

This section describes how to run the VMware upgrade script.

To run the VMware upgrade:

- 1. Open an SSH connection or the VM console.
- 2. Login into the EMS server as 'acems' user with password *acems* (or customer defined password).
- 3. Switch to 'root' user and provide root password (default password is root):

```
su - root
```



Note: Before starting the installation, it is highly recommended to configure the SSH client (e.g. Putty application) to save the session output into a log file.

4. Change directory to '/misc/cd/EmsServerInstall' and run the install script.

```
cd /misc/cd/EmsServerInstall
./install
```

Figure 9-10: EMS Server Installation Script



5. Enter **y**, and then press Enter to accept the License agreement.

Figure 9-11: EMS Server Upgrade (Linux) – License Agreement

based upon one neo income of processi.
11.4. Severability If any provision herein is ruled too broad in any respe
on shall be limited only so far as it is necessary to allow conformance to
shall be deleted from the Agreement, but the remaining provisions shall r
11.5. Assignment Neither this Agreement or any of Licensee's rights or obl
tten permission of Licensor and any attempt to do so shall be without effe
sferred to any person; (ii) the Licensee being merged or consolidated with
11.6. Export Licensee understands that the Licensed Software may be a regu
, and may require a license to export such. Licensee is solely responsible
11.7. Relationship of Parties Nothing herein shall be deemed to create an
the parties. Neither party shall have the right to bind the other to any o
11.8. Integration This Agreement is the complete and exclusive agreement b
ated hereto. Any Licensee purchase order issue for the software, documenta
erms hereof.
11.9. Counterparts This Agreement may be executed in multiple original cou
ing an authorized signature of Licensor and Licensee.
Do you accept this agreement? (y/n)y

6. OS patches are installed.

After the OS patches installation, you are prompted to press Enter to reboot.



Note: This step is optional and depends upon which version you are upgrading. After the EMS server has rebooted, repeat steps 2 to 5.

7. If the EMS version you are upgrading to is packaged with a later version of Java than the one that is currently installed, type yes, and then press Enter to upgrade the Java version, otherwise, skip this step:

```
Java DB version 10.4.2.1.1 is currently installed.
Upgrade to version 10.6.2.1.1 ? [yes,no]yes
```

8. At the end of Java installation, press Enter to continue.

Figure 9-12: EMS Server Application Upgrade (Linux) - Java Installation



9. Wait for the installation to complete and reboot the EMS server.

Figure 9-13: EMS Server Upgrade (Linux) Complete





Note: For SEM Users: each time the EMS server version is upgraded, the operator should perform CTRL – F5 (refresh) action on the SEM Page, and then re-login to the application.



This page is intentionally left blank.



EMS Server Machine Backup and Restore

This part describes how to restore the EMS server machine from a backup.

10 EMS Server Backup

There are two main backup processes that run on the EMS server:

Weekly backup: runs once a week at a pre-configured date & time (default is Saturday 02:00). In this process, the whole database is backed up into several "RMAN" files that are located in /data/NBIF/emsBackup/RmanBackup directory. In addition, many other configuration and software files are backed up to a TAR file in the /data/NBIF/emsBackup directory. In general, this TAR file contains the entire /data/NBIF directory's content (except 'emsBackup' directory), EMS Software Manager content and server_xxx directory's content.

To change the weekly backup's time and date, see Section 16.3.

Daily backup: runs daily except on the scheduled week day (see above). The daily backup process backs up the last 24 hours. There are no changes in the TAR file in this process.



Warning: The Backup process does not backup configurations performed using EMS Server Manager, such as networking and security.

It is highly recommended to maintain all backup files on an external machine.

These files can be transferred outside the server directly from their default location by SCP or SFTP client using 'acems' user. These backup files are as follows:

- /data/NBIF/emsBackup/emsServerBackup_<time&date>.tar file.
- All files in /data/NBIF/emsBackup/RmanBackup directory (including control.ctl and init.ora files)



This page is intentionally left blank.

11 EMS Server Restore

This section describes how to restore the EMS server. This can be done on the original machine that the backup files were created from or on any other machine.

Note:



- If you're running the restore process on a different machine, its disk size should be the same as the original machine from which the backup files were taken.
- Restore actions can be performed only with backup files which were previously created in the same EMS version.
- If you are restoring to a new machine, make sure that you have purchased a new license file machine ID. AudioCodes customer support will assist you to obtain a new license prior to the restore process.

To restore the EMS server:

1. Install (or upgrade) EMS to the same version from which the backup files were created. The Linux version must also be identical between the source and target machines.

For more details, see Chapter 8 on page 81.

2. Use the EMS Server Management utility to perform all the required configurations, such as Networking and Security, as was previously configured on the source machine.

For more details, see Chapter 12 on page 103.

- **3.** Make sure all server processes are up in EMS Server Manager / Status menu and the server functions properly.
- 4. Copy all backup files to /data/NBIF directory by SCP or SFTP client using the 'acems' user.
- 5. In EMS Server Manager, go to the Application Maintenance menu and select the **Restore** option.
- 6. Follow the instructions during the process. For more details, see Section 16.4 on page 118.
- 7. After the restore process has completed, you will be asked to reboot the machine.
- If you installed custom certificates prior to the restore, you must reinstall these certificates (see Appendix E).



This page is intentionally left blank.

Part V

EMS Server Manager

This part describes the EMS server machine maintenance using the EMS Server Management utility.

The EMS Server Management utility is a CLI interface that is used to configure networking parameters and security settings and to perform various maintenance actions on the EMS server.



Warning: Do not perform EMS Server Manager actions directly through the Linux OS shell. If you perform such actions, EMS application functionality may be harmed.



Note: To exit the EMS Server Manager to Linux OS shell level, press q.

12 Getting Started

This section describes how to get started using the EMS Server Manager.

12.1.1 Connecting to the EMS Server Manager

You can either run the EMS Server Manager utility locally or remotely:

- If you wish to run it remotely, then connect to the EMS server using Secure Shell (SSH).
- If you wish to run it locally, then connect using the management serial port or keyboard and monitor.

Do the following:

- 1. Login into the EMS server by SSH, as 'acems' user and enter password *acems*.
- 2. Switch to 'root' user and provide root password (default password is root):

su - root

3. Type the following command:

EmsServerManager



The EMS Server Manager menu is displayed:

Figure 12-1: EMS Server Manager Menu

Important:

- Whenever prompted to enter Host Name, provide letters or numbers.
- Ensure IP addresses contain all correct digits.



For menu options where reboot is required, the EMS server automatically reboots after changes confirmation.

For some of the configuration options, you are prompted to authorize the changes. There are three options: Yes, No, Quit (y,n,q). Yes implements the changes, **No** cancels the changes and returns you to the initial prompt for the selected menu option and **Quit** returns you to the previous menu.

The following describes the full menu options for the EMS Management utility:

- **Status** Shows the status of current EMS processes (see Chapter 13)
- General Information Provides the general EMS server current information from the Linux operating system, including EMS Version, EMS Server Process Status, Oracle Server Status, Apache Server Status, Java Version, Memory size and Time Zone. See Chapter 14.
- Collect Logs Collates all important logs into a single compressed file (see Chapter 15):
 - General Info
 - Collect Logs

- Application Maintenance Manages system maintenance actions (see Chapter 16):
 - Start / Stop the Application
 - Web Servers
 - Change Schedule Backup Time
 - Restore
 - High Availability
 - License
 - Shutdown the machine
 - Reboot the machine
- Network Configuration Provides all basic, advanced network management and interface updates (see Chapter 17):
 - Server IP Address (The server will be rebooted)
 - Ethernet Interfaces (The server will be rebooted)
 - Ethernet Redundancy (The server will be rebooted)
 - DNS Client
 - NAT
 - Static Routes
 - SNMP Agent
 - SNMPv3 Engine ID
- **Date & Time** Configures time and date settings (see Chapter 18):
 - NTP
 - Timezone Settings
 - Date and Time Settings
- **Security** Manages all the relevant security configurations (see Chapter 19):
 - EMS user
 - SSH
 - DB Password (EMS and SEM applications will be stopped)
 - OS Users Passwords
 - File Integrity Checker
 - Software Integrity Checker (AIDE) and Prelinking
 - USB Storage
 - Network Options
 - Audit Agent Options (the server will be rebooted)
 - HTTPS Authentication
 - Enable SEM client secured connection (EMS application will be restarted).

AudioCodes

- Enable IP Phone Manager client and JAWS secured communication (Apache will be restarted).
- Server Certificates Update
- SEM-AudioCodes device communication
- Diagnostics Manages system debugging and troubleshooting (see Chapter 20):
 - Server Syslog
 - Devices Syslog
 - Devices Debug

12.1.2 Using the EMS Server Manager

The following describes basic user hints for using the EMS Server Manager:

- The screens displaying the Main menu options in the procedures described in this section are based on a Linux installation with 'root' user permissions.
- The current navigation command path is displayed at the top of the screen to indicate your current submenu location in the CLI menu. For example, Main Menu > Network Configuration > Ethernet Redundancy.
- You can easily navigate between menu options using the keyboard arrow keys or by typing the menu option number.
- Each of the menu options includes an option to return to the main Menu "Back to Main Menu" and in some cases there is an option to go back to the previous menu level by specifying either "Back" or "Quit".



This page is intentionally left blank.
13 Viewing Process Statuses

You can view the statuses of the currently running EMS applications.

To view the statuses of the current EMS applications:

1. From the EMS Server Management root menu, choose **Status**, and then press Enter; the following is displayed:

Figure 13-1: Application Status

A 34 / 4		• • •					
Hpplication		Stat	tus				
i Watchdog		UP					
EMS Server		UP					
I SEM CPEs Server		UP					
I SEM MS Lync Server	H	UP					
: SEM Endpoints Server	8 H	UP					
l Tomcat Server	H	UP					
Apache Server		UP					
Oracle DB		UP					
Oracle Listener		UP .					
SNMP Agent		DOWN					
NTP Daemon		ПР					
Риссо	2 E	ntow	key to	back to	n main	menu	
11655	L.	ncer	VEA CO	DACK C	о нати	menu	

The following table describes the application statuses.

Application	Status
Watchdog	Indicates the status of the EMS Watchdog process.
EMS Server	Indicates the status of the EMS Server process.
SEM CPEs Server	Indicates the status of the XML based SEM communication between the devices and the SEM CPEs Server.
SEM MS Lync Server	Indicates the status of the SEM MS Lync Server, which manages the HTTP/S connection with the MS-SQL Server.
SEM Endpoints Server	Indicates the status of the Endpoint Server, which manages the UDP connection with the Endpoints (IP Phones) for SIP Publish RFC 6035 messages.
Tomcat Server	Indicates the status of the Tomcat server, which manages the connection with the SEM Web client.
Apache Server	Indicates the status of the Apache server, which manages the following connections: HTTP/S connection with the AudioCodes device,

Table 13-1: Application Statuses



Application	Status
	The EMS Server-Client connection.
	The HTTP connection that is used by Endpoints for downloading firmware and configuration files from the EMS server.
Oracle DB	Indicates the status of the Oracle Database process.
Oracle Listener	Indicates the status of the Oracle Listener process.
SNMP Agent	Indicates the status of the Linux SNMP Agent process. This agent is not responsible for the SNMPv2/SNMPv3 connection with the AudioCodes devices.
NTP Daemon	Indicates the status of the NTP Daemon process.

14 Viewing General Information

This section describes the General Information and Logs collection options. The General Information option provides detailed information about the EMS server configuration and current status variables. The following information is provided:

- Components versions: EMS, Linux, Java, Apache
- Components Statuses: EMS server process and security, Watchdog, Apache, Oracle, SNMP agent, Tomcat and SEM.
- Memory size and disk usage
- Network configuration
- Time Zone and NTP configuration
- User logged in and session type

> To view General Information:

1. From the EMS Server Management root menu, choose **General Information**, and then press Enter; the following is displayed:

Figure 14-1: General Information

```
chine information
Invironment: Virtual(Manufacturer: VMware,
IPU: Intel(R) Xeon(R) CPU X5650
Iemory: 2059588 kB
                                                                     Inc.)
@ 2.67GHz
            sage: 629M
    k /dev/sda: 64.4 GB, 64424509440 bytes
Data usage:
'dev/mapper/vg-data
                                    40G 6.1G
                                                       31G 17% /data
 ersions
EMS Version
OS Version
OS Revision
                          Linux
                                               -19
                                                          1.el5
                                                                          64
                                      5.3 for
                          CentOS
                                                                              lized (Rev. 4)
Java Version : java full ve
Apache version: Apache/2.2.3
                                                 ion
                                                    ver built:
                                                                                 9
                                                                                   2013 08:22:33
                                              Ser
more>
```



2. Press **<more>** to view more information; the following is displayed:

Figure 14-2: General Information

Hachine information Environment: Uirtual(Manufacturer: UMware, Inc.) ICPU: Intel(R) Xeon(R) CPU X5650 @ 2.67GHz IMemory: 2059588 kB IACEMS Usage: 629M IDisk: Disk /dev/sda: 64.4 GB, 64424509440 bytes IData usage: /dev/mapper/vg-data 40G 6.1G 31G 17% /data
Versions IEMS Version : 6.8.49 IOS Version : Linux 2.6.18-194.32.1.e15 x86_64 IOS Revision : CentOS 5.3 for EMS Server Virtualized (Rev. 4) IJava Version : java full version "1.6.0_43-b01" IApache version: Apache/2.2.3 Server built: Jan 9 2013 08:22:33 (more)
Network Configuration Server's Network: Interface : eth0 Host Name : global-logic-2 IP Address : 10.4.100.17 Subnet Mask : 255.255.0.0 Network Address : 10.4.0.0
Hetwork Time Protocol Server #1 Peer: : *LOCAL(0) Sync source : LOCL. Stratum: : 13 Type : Local Last response : 47 seconds ago Polling interval: : 64 seconds Reach : 377 (all attempts successful) Delay : 0.000 ms. Offset : 0.001 ms. Jitter : 0.001 ms. Press 'Enter' key to back to main menu

15 Collecting Logs

This option enables you to collect important log files. All log files are collected in a single file log.tar that is created under the user home directory. The log file size is approximately 5MB. The following log files are collected:

- EMS Server Application logs
- Server's Syslog Messages
- Oracle Database logs
- Tomcat logs
- Hardware information (including disk)
- Relevant network configuration files (including static routes)

> To collect logs:

From the EMS Server Management root menu, choose Collect Logs, and then press Enter; the EMS server commences the log collection process:

Figure 15-1: EMS Server Manager – Collect Logs

```
Collecting logs

Collecting EMS Server logs...

Collecting OS logs...

Collecting Oracle DB logs...

Collecting hardware configuration...

Collecting OS configuration...

Collecting OS configuration...

Collecting Rman Log Files

Collecting Tomcat Log Files

Collecting Insallation Log Files

Collecting Yafic Scan Files

Collecting GeneralInfo

Collecting Topology File

Packing TAR file...

adding: logs.tar (deflated 83%)
```



This process can take a few minutes. Once the file generation has completed, a message is displayed on the screen informing you that a Diagnostic tar file has been created and the location of the tar file:

Figure 15-2: TAR File Location

Collecting	logs
Collecting	EMS Server logs
Collecting	OS logs
Collecting	Oracle DB logs
Collecting	hardware configuration
Collecting	OS configuration
Collecting	Rman Log Files
Collecting	Tomcat Log Files
Collecting	Insallations_Log Files
Collecting	Yaffic_Scan Files
Collecting	GeneralInfo
sh: HA: com	mand not found
Packing TAR	file
updating: h	ome/acems/logs.tar (deflated 95%)
The diagnos	tics IAR file can be found in /home/acems/logs.tar
Press Enter	to continue

16 Application Maintenance

This section describes the application maintenance.

To configure application maintenance:

From the EMS Server Manager root menu, choose Application Maintenance; the following is displayed:

Figure 16-1: Application Maintenance

EMS Server 7.2.126 Management
Main Menu> Application Maintenance
>1.Start/Stop Application 2.Web Servers 3.Change Schedule Backup Time 4.Restore 5.High Availability 6.License 7.Shutdown the Machine 8.Reboot the Machine q.Quit to main Menu

This menu includes the following options:

- Start/Stop Application (see Section 16.1 on page 115).
- Web Servers (see Section 16.2on page 116).
- Change Schedule Backup Time (see Section 16.316.3 on page 118).
- Restore (see Section 16.4 on page 118)
- High Availability (see Chapter 21 on page 175).
- License (see Section 16.5 on page 119).
- Shutdown the Machine (see Section 16.6 on page 121).
- Reboot the Machine (see Section 16.7 on page 121).

16.1 Start /Stop the Application

This section describes how to start or stop the application.

To start/stop the application:

1. From the Application Maintenance menu, choose **Start / Stop the Application**, and then press Enter; the following is displayed:

	EMS Server 7.2.126 Management									
Ma	Main Menu> Application Maintenance									
EMS	Server >1 2	is started. . Hes .No	Stop EMS	Server?						

Figure 16-2: Start or Stop the EMS Server

2. Select Yes to start the EMS server or No to stop it.

16.2 Web Servers

From the Application maintenance menu, choose **Web Servers**, and then press Enter; the following is displayed:

Figure 16-3: – Web Servers

EMS Server 7.2.3037 Management
Main Menu> Application Maintenance> Web Servers
The Web Server's Processes are: UP The Tomcat Server's Processes are: DOWN Port 80 (HTTP): OPEN Port 8080 (IPPs FILES): OPEN Port 8081 (IPPs HTTP): OPEN Port 8082 (IPPs HTTPS): OPEN JAWS Service: ENABLED
1.Stop the Apache Server 2.Start the Tomcat Server 3.Close HTTP Service (Port 80) 4.Close IPPs FILES (Port 8080) >5.Elose IPPs HTTP (Port 8081) 6.Close IPPs HTTPS (Port 8082) 7.Disable JAWS 8.JAWS IP Configuration b.Back q.Quit to main Menu

16.2.1 Apache and Tomcat Server Processes

This section describes how to open and close the Apache and Tomcat Web server connections.

To stop the Apache server:

In the Web Servers menu, choose option Stop/Start Apache Server, and then press Enter.

To stop the Tomcat server:

In the Web Servers menu, choose option Stop/Start Tomcat Server, and then press Enter.

16.2.2 HTTP/HTTPS Services

This section describes how to open and close the different HTTP/HTTPS services.

To open/close HTTP Service (Port 80):

In the Web Servers menu, choose option Open/Close HTTP Service (Port 80), and then press Enter.

This HTTP port is used for the connection between the EMS server and all AudioCodes devices, with the JAWS client and with the IP Phone Management Server Web browser.

To open/close IPPs FILES (Port 8080):

In the Web Servers menu, choose option Open/Close IPPs FILES (Port 8080), and then press Enter.

This HTTP port is used for downloading firmware and configuration files from the EMS server to the endpoints.

To open/close IPPs HTTP (Port 8081):

In the Web Servers menu, choose option Open/Close IPPs HTTP (Port 8081), and then press Enter.

This HTTP port is used for sending REST updates from the endpoints to the EMS server, such as alarms and statuses.

To open/close IPPs HTTPS (Port 8082):

In the Web Servers menu, choose option Open/Close IPPs HTTPS (Port 8082), and then press Enter.

This HTTPS port is used for sending secure REST updates from the endpoints to the EMS server, such as alarms and statuses (HTTPS without certificate authentication).

To enable/disable JAWS client:

In the Web Servers menu, choose option Enable/Disable JAWS, and then press Enter.

16.2.2.1 JAWS IP Configuration

By default, logging into the EMS server using JAWS can only be performed through the EMS server's first interface only. This option allows you to configure an alternative interface for the JAWS login.

> To change the JAWS login interface:

- 1. From the Web Server configuration menu, choose option **6**, and then press Enter.
- 2. Type the desired interface IP address, press Enter, and then confirm by typing y.

Figure 16-4: JAWS IP Configuration

		JAVS	IP C	onfig	gurat:	ion		
		IP Ad	ldress	s [10,	.4.10	9.17	7]:	
Are	you	sure	that	you	want	to	continue?	(y/n/q)

16.3 Change Schedule Backup Time

This step describes how to reschedule the backup time.

To schedule backup time:

- 1. From the Application Maintenance menu, choose Change Schedule Backup Time.
- 2. Choose the day of the week that you wish to perform the backup.
- 3. Copy all files in /data/NBIF/emsBackup/RmanBackup/ directory to an external machine.
- 4. Copy /data/NBIF/emsBackup/emsServerBackup_<time&date>.tar file to an external machine.

Where <time&date> is only an example; replace this path with your filename.

16.4 Restore

This step describes how to restore the EMS server.

To restore the EMS server:

- 1. Copy all files that you backed up in Section 16.3 to the /data/NBIF directory on the Restore server. Overwrite existing files if required.
- 2. From the EMS server Application Maintenance menu, choose **Restore**; a script is started.
- 3. Follow the instructions; you might need to press Enter a few times.
- 4. After the restore operation has completed and then reboot the EMS server (see Section 17.1).

16.5 License

The License menu enables you to view the details of the existing license or upload a new license.

The OVOC Server License (SBC License pool, EMS for IP Phones, SEM device and IP Phone Management) should have a valid license loaded to the server in order for it to be fully operational.

In order to obtain a valid license for your OVOC Server License you should activate your product through AudioCodes License Activation tool at <u>http://www.audiocodes.com/swactivation</u>. You will need your Product Key (see below) and the Server Machine ID (see below) for this activation process:**Product Key:** the

- Product Key string represents the customer order for the EMS and SEM products. For more information, contact your AudioCodes partner.
- Machine ID: the Machine ID should be taken from the server as shown in the screen below (enter this ID in the Fingerprint field in the Activation form).
- License Status: indicates whether the EMS license is enabled (see Section 16.5.1 below).
- **Expiration Date:** indicates the expiration date of the EMS time license. If an expiration date is not configured, this field displays 'Unlimited' (see below).

The timezone is determined by the configured date and time in the Date & Time menu (see Section 18.2).

You will receive an e-mail with your product license file.

Note:

- When you order AudioCodes devices (Mediant SBC and Mediant Gateway AudioCodes products), ensure that for those devices that you wish to manage using EMS, a valid feature key is enabled with the "EMS" parameter. Note that this feature key is a separate license to the OVOC Server license that is described in this Section.
- The Mediant 3000 and MP-1288 products are not supported by the SBC License Pool Manager.

16.5.1 EMS Time License

The EMS and SEM license may control the time period for which the product can be used. When the time license is enabled and the configured license time expires, the connection to the EMS/SEM server is denied. The time based license affects all the features in the EMS and SEM including the SBC License Pool, EMS for IP Phones and SEM. When the EMS server time license approaches or reaches its expiration date, the 'EMS License' alarm is raised (Refer to the relevant product *Performance Monitors and Alarms Guide*).

To view the license details or upload a new license:

1. Copy the license file that you have obtained from AudioCodes to the following path on the EMS server machine:



/home/acems/<License_File>

2. From the Application Maintenance menu, choose **License** option, and then press Enter; the current SEM License Manager details are displayed:

Figure 16-5: License Manager

EMS Server 7.2.3035 Management					
Main Menu> Application Maintenance> License					
License Configuration Manager: Server Machine ID: B9C8B237B0DF Product Key: License Status: ENABLED Expiration Date: Unlimited					
EMS License Pool Managed Devices: 32,000 SBC Sessions: 32,000 SBC Registrations: 32,000 SBC Transcoding: 32,000 SBC Signaling: 32,000 CB Users: CB PBX Users: CB PBX Users: CB Analog Devices: CB Voicemail Accounts:					
EMS for IP Phones IP Phones Number: 15,000					
SEM Devices Number: 5,000 IP Phones Number: 15,000 SEM Sessions: 64,000 SEM Users: 140,000					
>1.Load License b.Back q.Quit to main Menu					

EMS (SBC) License Pool:

- The supported number of managed devices
- The supported number of managed SBC sessions
- The supported number of managed SBC user registrations
- The supported number of managed SBC transcoding sessions
- The supported number of managed SBC signaling sessions
- The supported number of managed CB users
- The supported number of managed CloudBond (CB) PBX users

- The supported number of managed CB Analog devices (for future support)
- The supported number of managed CB Voicemail accounts
- EMS for IP Phones:
 - The supported number of IP Phones that can be managed by the EMS.
- SEM:
 - The supported number of devices that can be managed by the SEM.
 - The supported number of IP Phones that can be managed by the SEM.
 - The supported number of simultaneous call sessions that can be managed by the SEM.
 - The supported number of users that can be managed by the SEM.
- 3. To load a new license, choose option 1.
- 4. Enter the license file path and name.
- 5. Restart the EMS server.

16.6 Shutdown the EMS Server Machine

This section describes how to shut down the EMS Server machine.

To shut down the EMS server machine:

- 1. From the Application Maintenance menu, choose **Shutdown the Machine**, and then press Enter.
- 2. Type **y** to confirm the shutdown; the EMS server machine is shutdown.

16.7 Reboot the EMS Server Machine

This section describes how to reboot the EMS server machine.

To reboot the EMS server machine:

- 1. From the Application Maintenance menu, choose **Reboot the Machine**, and then press Enter.
- 2. Type **y** to confirm the reboot; the EMS server machine is rebooted.



This page is intentionally left blank.

17 Network Configuration

This section describes the networking options in the EMS Server Manager.

To run the network configuration:

From the EMS Server Manager root menu, choose Network Configuration; the following is displayed:

Figure 17-1: Network Configuration

EMS Server 7.2.126 Management								
Main Menu> Network Configuration								
<pre>>1.Server IP Address 2.Ethernet Interfaces 3.Ethernet Redundancy 4.DNS Client 5.NAT 6.Static Routes 7.SNMP Agent 8.SNMPv3 Engine ID q.Quit to main Menu</pre>	(The server will be rebooted) (The server will be rebooted) (The server will be rebooted)							

This menu includes the following options:

- Server IP Address (the server will be rebooted) (see Section 17.1 on page 123).
- Ethernet Interfaces (the server will be rebooted) (see Section 17.2 on page 125).
- Ethernet Redundancy (the server will be rebooted) (see Section 17.3 on page 128).
- DNS Client (see Section 17.4 on page 133).
- NAT (see Section 17.5 on page 134).
- Static Routes (see Section 17.6 on page 134).
- SNMP Agent (see Section 17.7 on page 135).
- SNMPv3 Engine ID (see Section 17.8 on page 135).

17.1 Server IP Address

This option enables you to update the EMS server's IP address. This option also enables you to modify the EMS server host name.



Note: When this operation has completed, the EMS automatically reboots for the changes to take effect.

> To change Server's IP address:

1. From the Network Configuration menu, choose **Server IP Address**, and then press Enter; the following is displayed:

Figure 17-2: EMS Server Manager – Change Server's IP Address



2. Configure IP configuration parameters as desired.

Each time you press Enter, the different IP configuration parameters of the EMS server are displayed. These parameters include the Server Host Name, IP address, Subnet Mask, Network Address and Default Gateway.

3. Type **y** to confirm the changes, and then press Enter.

Figure 17-3: IP Configuration Complete



Upon confirmation, the EMS automatically reboots for the changes to take effect.

17.2 Ethernet Interfaces

This section describes how to configure Ethernet interfaces.

17.2.1 EMS Client Login on all EMS Server Network Interfaces

The EMS server can be configured with up to four network interfaces (connected to different subnets) as described above. You can connect to any one of the above interfaces directly from the EMS client login dialog.

The "Server IP" field in EMS client login dialog is set to the desired EMS server network interface IP address.



Figure 17-4: EMS Server: Triple Ethernet Interfaces

In case gateways are located in different subnets, static routes should be provisioned to allow the connection from 'Southbound Network' to each one of the subnets. For Static Routes configuration, see Section 17.6 on page 134.

To ensure that the network configuration is performed successfully, test that the EMS is successfully connected to each one of the gateways by running the following basic tests:

- Adding the gateway to the EMS application
- Reviewing its status screen
- Performing basic configuration action (set of 'MG Location' in Media Gateways Provisioning Frame / General Setting tab)
- Ensuring that the EMS receives traps from the gateway by adding TP boards in one of the empty slots and ensuring that the 'Operational Info' Event is received.

To configure Ethernet Interfaces:

1. From the Network Configuration menu, choose **Ethernet Interfaces**, and then press Enter; the following is displayed:

Figure 17-5: EMS Server Manager – Configure Ethernet Interfaces

	EMS	Server 7.2.126	Managemei	nt
Main	Menu> Network	Configuration>	Ethernet	Interfaces
	>1. Add Inte 2. Remove I 3. Modify I b. Back q. Quit to	nterface nterface main Menu		

- **2.** Choose from one of the following options:
 - Add Interface Adds a new interface to the EMS server (see Section 17.2.2 on page 126).
 - **Remove Interface** Removes an existing interface from the EMS server (see Section 17.2.3 on page 127).
 - **Modify Interface** Modifies an existing interface from the EMS server (see Chapter 3 on page 127).

17.2.2 Add Interface

This section describes how to add a new interface.

To add a New Interface:

- **1.** From the Ethernet Interfaces menu, choose option **1**; a list of currently available interfaces (not yet configured) is displayed.
- 2. Choose an interface (on HP machines the interfaces are called 'eth0', 'eth1', etc).
- **3.** Choose the Network Type.
- 4. Enter values for the following interface parameters and confirm:
 - IP Address
 - Hostname
 - Subnet Mask

The new interface parameters are displayed.

5. Type **y** to confirm the changes; the EMS server automatically reboots for the changes to take effect.

Figure 17-9: Add Interface Parameters

Add Ir	iterface:
Choose 1) eth 2) eth 3) eth q) Qui : 1	e Interface: 11 12 13 14 15
Choose	<pre>Network Type: 1) Network 1 (MG's Network) 2) Network 2 3) Network 3 4) Quit : 1</pre>
New Ir	aterface Parameters:
IP Add Hostna Subnet	iress : 10.4.100.55 ame : GWs : Mask : 255.255.0.0
Note: Reboot	will be performed immediately at the end of configuration process.
Are you sure t	that you want to continue? (y/n/q)

17.2.3 Remove Interface

This section describes how to remove an interface.

To remove an existing interface:

- 1. From the Ethernet Interfaces menu, choose option **2**; the following is displayed:
- 2. Choose the interface to remove.
- **3.** Type **y** to confirm the changes; the EMS server automatically reboots for the changes to take effect.

17.2.4 Modify Interface

This section describes how to modify an existing interface.

To modify an existing interface:

- 1. From the Ethernet Interfaces menu, choose option **3**.
- 2. Choose the interface to modify; the following is displayed:
- 3. Change the interface parameters.
- **4.** Type **y** to confirm the changes; the EMS server automatically reboots for the changes to take effect.

17.3 Ethernet Redundancy

This section describes how to configure Ethernet Redundancy.

Physical Ethernet Interfaces Redundancy provides failover when you have multiple network interface cards that are connected to the same IP link.

The EMS server supports up to four Ethernet interfaces. For enhanced network security, it is recommended to use two interfaces and to define Ethernet ports redundancy on both of them. For example, EMS Clients [Northbound] and Gateways [Southbound]).

This option enables you to configure Ethernet ports redundancy.



Note: When the operation is finished, the EMS server automatically reboots for the changes to take effect.





To configure Ethernet Redundancy:

1. From the Network Configuration menu, choose **Ethernet Redundancy** option, and then press Enter; the following is displayed:

```
Figure 17-7: Ethernet Redundancy Configuration
```



- 2. This menu includes the following options:
 - Add Redundant Interface (see Section 17.3.1 on page 129.
 - Remove Redundant Interface (see Section 17.3.2 on page 131).
 - Modify Redundant Interface (see Section 17.3.3 on page 132).

17.3.1 Add Redundant Interface

Remove a redundant interface under the following circumstances:

- You have configured an Ethernet interface (see Section 17.3.1).
- Your default router can respond to a 'ping' command, due to a heartbeat procedure between interfaces and the default router (to verify activity).

To add a redundant interface:

- **1.** From the Ethernet Redundancy menu, choose option **1**.
- 2. Choose the network type for which to create a new redundant interface (for example, 'EMS Client-Server Network').
- 3. Choose the interface in the selected network that you wish to make redundant (for example, 'bge1', 'bge2', 'bge3').
- 4. Choose the redundancy mode (for example, 'balance-rr', 'active-backup').

AudioCodes

5. Type **y** to confirm the changes; the EMS server automatically reboots for changes to take effect.

Figure 17-8: Add Redundant Interface (Linux)

Interface: eth0 Network: Server's Network IP Address: 10.7.14.141 Interface: eth1 Not configured 1) Add Redundant Interface 2) Remove Redundant Interface 3) Modify Redundant Interface 4) Back to Main Menu 1 1 Add Redundant Interface: Choose Network Type: 1) Server Network 2) Quit 1 1 Choose Redundant Interface: 1) eth1 q) Quit 1 Ethernet Redundancy Settings: Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup - recommended 2) balance-xor (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregation 5) balance-alb (adaptive load balancing) 6) balancenalb (adaptive load balancing)		Ethernet Redundancy Configuration
Network: Server's Network IP Address: 10.7.14.141 Interface: eth1 Not configured 1) Add Redundant Interface 2) Remove Redundant Interface 3) Modify Redundant Interface 4) Back to Main Menu 1 Add Redundant Interface: Choose Network Type: 1) Server Network 2) Quit 1 Choose Redundant Interface: 1) eth1 q) Quit 1 Choose Redundant Interface: 1) eth1 q) Quit 1 Ethernet Redundancy Settings: Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup - recommended 2) balance-xor (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregation 5) balance-alb (adaptive load balancing) 4) alancenalb (adaptive load balancing) 5) balance-alb (adaptive load balancing)		Interface: eth0
IP Address: 10.7.14.141 Interface: eth1 Not configured 1) Add Redundant Interface 2) Remove Redundant Interface 3) Modify Redundant Interface 4) Back to Main Menu 5 1 Add Redundant Interface: Choose Network Type: 1) Server Network 2) Quit 5 1 Choose Redundant Interface: 1) eth1 4) Quit 5 1 Ethernet Redundancy Settings: Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup - recommended 2) balance-xor (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregation 5) balance-alb (adaptive load balancing) 6) balance-alb (adaptive load balancing) 7		Network: Server's Network
Interface: eth1 Not configured 1) Add Redundant Interface 2) Remove Redundant Interface 3) Modify Redundant Interface 4) Back to Main Menu 5 1 Add Redundant Interface: Choose Network Type: 1) Server Network 2) Quit 5 1 Choose Redundant Interface: 1) eth1 4) Quit 5 1 Ethernet Redundancy Settings: Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup - recommended 2) balance-ru (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregation 5) balance-alb (adaptive load balancing) 6) balance-alb (adaptive load balancing) 7		IP Address: 10.7.14.141
Not configured 1) Add Redundant Interface 2) Remove Redundant Interface 3) Modify Redundant Interface 4) Back to Main Menu 1 1 Add Redundant Interface: Choose Network Type: 1) Server Network 2) Quit 1 1 Choose Redundant Interface: 1) eth1 4) Quit 1 1 Ethernet Redundancy Settings: Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup - recommended 2) balance-xor (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregation 5) balance-alb (adaptive load balancing) 6) balance-alb (adaptive load balancing)		Interface: eth1
<pre>1) Add Redundant Interface 2) Remove Redundant Interface 3) Modify Redundant Interface 4) Back to Main Menu 5 1 Add Redundant Interface: Choose Network Type: 1) Server Network 2) Quit 5 1 Choose Redundant Interface: 1) eth1 9) Quit 5 1 Ethernet Redundancy Settings: Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup - recommended 2) balance-xor (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregation 5) balance-alb (adaptive load balancing) 6) balance-alb (adaptive load balancing) 7</pre>		Not configured
<pre>2) Remove Redundant Interface 3) Modify Redundant Interface 4) Back to Main Menu ; 1 Add Redundant Interface: Choose Network Type: 1) Server Network 2) Quit ; 1 Choose Redundant Interface: 1) eth1 q) Quit ; 1 Ethernet Redundancy Settings: Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup = recommended 2) balance-xor (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregation 5) balance-alb (adaptive load balancing) 6) balance-alb (adaptive load balancing) 71</pre>		1) Add Redundant Interface
<pre>3) Modify Redundant Interface 4) Back to Main Menu ; 1 Add Redundant Interface: Choose Network Type: 1) Server Network 2) Quit ; 1 Choose Redundant Interface: 1) eth1 q) Quit ; 1 Ethernet Redundancy Settings: Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup - recommended 2) balance-xor (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregatio 5) balance-alb (adaptive load balancing) 6) balancing)</pre>		Remove Redundant Interface
<pre>4) Back to Main Menu : 1 Add Redundant Interface: Choose Network Type: 1) Server Network 2) Quit : 1 Choose Redundant Interface: 1) eth1 q) Quit : 1 Ethernet Redundancy Settings: Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup - recommended 2) balance-ror (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregation 5) balance-alb (adaptive load balancing) 6) balance-alb (adaptive load balancing) 71</pre>		 Modify Redundant Interface
<pre>: 1 Add Redundant Interface: Choose Network Type: 1) Server Network 2) Quit : 1 Choose Redundant Interface: 1) eth1 q) Quit : 1 Ethernet Redundancy Settings: Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup - recommended 2) balance-ror (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregation 5) balance-alb (adaptive load balancing) 6) balance-alb (adaptive load balancing) 71</pre>		4) Back to Main Menu
Add Redundant Interface: Choose Network Type: 1) Server Network 2) Quit 1 1 Choose Redundant Interface: 1) eth1 q) Quit 1 1 Ethernet Redundancy Settings: Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup = pecommended 2) balance-xor (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregation 5) balance-tlb (transmit load balancing) 6) balance-alb (adaptive load balancing)		
<pre>Add Redundant Interface: Choose Network Type: 1) Server Network 2) Quit 2 Quit 2 1 Choose Redundant Interface: 1) eth1 q) Quit 2 1 Ethernet Redundancy Settings: Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup - recommended 2) balance-rr (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregation 5) balance-tlb (transmit load balancing) 6) balance-alb (adaptive load balancing) 1) - 1</pre>		add and a second second
Choose Network Type: 1) Server Network 2) Quit : 1 Choose Redundant Interface: 1) eth1 q) Quit : 1 Ethernet Redundancy Settings: Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup - recommended 2) balance-ror (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregation 5) balance-tlb (transmit load balancing) 6) balance-alb (adaptive load balancing)		Add Redundant Interface:
<pre>1) Server Network 2) Quit 2) Quit 2) Quit 2) 1 2) Choose Redundant Interface: 1) eth1 q) Quit 2) 1 2) Ethernet Redundancy Settings: Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup - recommended 2) balance-rr (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregation 5) balance-tlb (transmit load balancing) 6) balance-alb (adaptive load balancing) 7.1</pre>		Choose Network Type:
<pre>2) Quit : 1 Choose Redundant Interface: 1) eth1 q) Quit : 1 Ethernet Redundancy Settings: Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup - recommended 2) balance-ror (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregation 5) balance-tlb (transmit load balancing) 6) balance-alb (adaptive load balancing) 7 1</pre>		1) Server Network
<pre>: 1 Choose Redundant Interface: 1) eth1 q) Quit : 1 Ethernet Redundancy Settings: Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup - recommended 2) balance-xor (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregation 5) balance-tlb (transmit load balancing) 6) balance-alb (adaptive load balancing) 1.1</pre>		2) Quit
Choose Redundant Interface: 1) eth1 q) Quit : 1 Ethernet Redundancy Settings: Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup - recommended 2) balance-ror (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregation 5) balance-tlb (transmit load balancing) 6) balance-alb (adaptive load balancing) 1)		
<pre>1) eth1 q) Quit ; 1 Ethernet Redundancy Settings: Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup - recommended 2) balance-xor (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregation 5) balance-tlb (transmit load balancing) 6) balance-alb (adaptive load balancing) 1.1 </pre>		Choose Redundant Interface:
<pre>q) Quit : 1 Ethernet Redundancy Settings: Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup - recommended 2) balance-xor (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregation 5) balance-tlb (transmit load balancing) 6) balance-alb (adaptive load balancing) 1 1</pre>		1) ethl
<pre>: 1 Ethernet Redundancy Settings: Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup - recommended 2) balance-xor (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregation 5) balance-tlb (transmit load balancing) 6) balance-alb (adaptive load balancing) </pre>		q) Quit
Ethernet Redundancy Settings: Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup - recommended 2) balance-xor (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregation 5) balance-tlb (transmit load balancing) 6) balance-alb (adaptive load balancing) 1. 1		
Ethernet Redundancy Mode: 0) balance-rr (round-robin load balancing) 1) active-backup - recommended 2) balance-xor (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregatic 5) balance-tlb (transmit load balancing) 6) balance-alb (adaptive load balancing) 1		Ethernet Redundancy Settings:
0) balance-rr (round-robin load balancing) 1) active-backup - recommended 2) balance-xor (XOR-policy load balancing) 3) broadcast 4) 802.3ad (IEEE 802.3ad dynamic link aggregatic 5) balance-tlb (transmit load balancing) 6) balance-alb (adaptive load balancing)		Pthernet Redundancy Mode:
 active-backup - recommended balance-xor (XOR-policy load balancing) broadcast 802.3ad (IEEE 802.3ad dynamic link aggregatic balance-tlb (transmit load balancing) balance-alb (adaptive load balancing) 		() balance-rr (round-robin load balancing)
 balance-xor (XOR-policy load balancing) broadcast 802.3ad (IEEE 802.3ad dynamic link aggregatic balance-tlb (transmit load balancing) balance-alb (adaptive load balancing) 		1) active-backup - recommended
 Broadcast 802.3ad (IEEE 802.3ad dynamic link aggregatic balance-tlb (transmit load balancing) balance-alb (adaptive load balancing) 		2) balance-xor (XOR-policy load balancing)
 4) 802.3ad (IEEE 802.3ad dynamic link aggregation 5) balance-tlb (transmit load balancing) 6) balance-alb (adaptive load balancing) 1 		3) broadcast
 balance-tlb (transmit load balancing) balance-alb (adaptive load balancing) 		4) 802.3ad (IEEE 802.3ad dynamic link aggregation)
balance-alb (adaptive load balancing)		5) balance-tlb (transmit load balancing)
		6) balance-alb (adaptive load balancing)
e you sure that you want to continue? (y/n/g)	e vou	sure that you want to continue? (y/n/g)

17.3.2 Remove Ethernet Redundancy

This section describes how to remove an Ethernet redundancy interface.

To remove the Ethernet Redundancy interface:

- 1. From the Ethernet Redundancy menu, choose option 2.
- 2. Choose the network redundancy to remove.

The current Ethernet redundancy configuration is displayed.

3. Type **y** to confirm the changes; the EMS server automatically reboots for the changes to take effect.

Figure 17-9: Ethernet Redundancy Interface to Disable

```
Ethernet Redundancy Configuration

Interface: eth0

Network: Server's Network

IP Address: 10.7.14.141

Interface: eth1

Network: Server's Network (redundant interface)

1) Add Redundant Interface

2) Remove Redundant Interface

3) Modify Redundant Interface

4) Back to Main Menu

: 2

Remove Redundant Interface:

Choose Redundant Interface:

Choose Redundant Network

1) Server's Network (eth0, eth1)

q) Quit

: 1

Are you sure that you want to continue? (y/n/q) y
```

AudioCodes

17.3.3 Modify Redundant Interface

This section describes how to modify a redundant interface.

To modify redundant interface and change redundancy settings:

- 1. From the Ethernet Redundancy, choose option **3**.
- 2. Choose the Ethernet redundancy interface to modify.
- 3. Change the redundancy settings.
- **4.** Type **y** to confirm the changes; the EMS server automatically reboots for the changes to take effect.

Figure 17-10: Modify Redundant Interface (Linux)

```
Ethernet Redundancy Configuration
       Interface: eth0
               Network: Server's Network
               IP Address: 10.7.14.141
       Interface: eth1
               Network: Server's Network (redundant interface)
       1) Add Redundant Interface
       2) Remove Redundant Interface
       3) Modify Redundant Interface
       4) Back to Main Menu
       Modify Redundant Interface:
       Choose Redundant Network
       1) Server's Network (eth0, eth1)
       q) Quit
       Ethernet Redundancy Settings:
       Ethernet Redundancy Mode:
       0) balance-rr (round-robin load balancing)
       1) active-backup - recommended
       2) balance-xor (XOR-policy load balancing)
       broadcast
       4) 802.3ad (IEEE 802.3ad dynamic link aggregation)
       5) balance-tlb (transmit load balancing)
       6) balance-alb (adaptive load balancing)
        [1]: 0
Are you sure that you want to continue? (y/n/q) y
```

17.4 DNS Client

Domain Name System (DNS) is a database system that translates a computer's fully qualified domain name into an IP address. If a DNS server cannot fulfill your request, it refers the request to another DNS server - and the request is passed along until the domain-name-to-IP-address match is made.

This option enables you to configure the client side (Resolver). If there is no existing DNS configuration, the option **Configure DNS** is displayed. If already configured, the option **Modify DNS** is displayed.

To Configure the DNS Client:

1. From the Network Configuration menu, choose **DNS Client**, press Enter, and then in the sub-menu, choose **Configure DNS**; the following is displayed:

```
Do you want to specify the local domain name ? (y/n)y
Local Domain Name: Brad
Do you want to specify a search list ? (y/n)y
Search List (use "," between domains names): Brad
DNS IP Address 1: 10.1.1.10
DNS IP Address 2: 10.1.1.11
DNS IP Address 3: 10.1.1.12
New DNS Configuration:
Domain Name: Brad
Search List: Brad
Search List: Brad
DNS IP 1: 10.1.1.10
DNS IP 2: 10.1.1.11
DNS IP 3: 10.1.1.12
Are you sure that you want to continue? (y/n/q)
```

Figure 17-11: DNS Setup

- 2. Specify the location domain. Type **y** to specify the local domain name or type **n**, and then press Enter.
- **3.** Specify a search list; type **y** to specify a list of domains (use a comma delimiter to separate search entries in the list) or type **n**, and then press Enter.
- 4. Specify DNS IP addresses 1, 2 and 3.
- 5. Type **y** to confirm your configuration; the new configuration is displayed.

17.5 NAT

NAT is the process of modifying network address information in datagram packet headers traversing a traffic routing device for the purpose of remapping a given address space to another.

To configure NAT:

- 1. From the Network Configuration menu, choose **NAT**, and then press Enter.
- 2. Enable a NAT address; type y.
- 3. Enter the NAT address, and then press Enter.
- 4. Type **y** to confirm the changes.
- 5. Stop and start the EMS server for the changes to take effect.

To remove NAT configuration:

- 1. Enter the value -1.
- 2. Type y to confirm the changes.
- 3. Stop and start the EMS server for the changes to take effect.

17.6 Static Routes

This option enables you to add or remove static route rules. Static routes are usually only used in conjunction with /etc/defaultrouter. Static routes may be required for network topology, where you don't want to traverse your default Gateway/Router. In this case, you will probably wish to make the routes permanent by adding the static routing rules.

To configure static routes:

1. From the Network Configuration menu, choose **Static Routes**, and then press Enter; the Static Routes Configuration is displayed:

Figure 17-12	: Routing	Table and	Menu
--------------	-----------	-----------	------



- 2. From the Static Routes configuration screen, choose one of the following options:
 - Add a Static Route
 - Remove a Static Route

To add a static route:

- **1.** From the Static Routes menu, choose option **1**.
- 2. Enter the Destination Network Address.
- 3. Enter the router's IP address.
- 4. Type **y** to confirm the changes.

To remove a static route:

- 1. From the Static Routes menu, choose option 2.
- 2. Enter the Destination Network Address for the static route you wish to remove.
- **3.** Enter the router's IP address.
- **4.** Type **y** to confirm the changes.

17.7 SNMP Agent

The SNMP Management agent enables access to system inventory and monitoring and provides support for alarms using the industry standard management protocol: Simple Network Management Protocol (SNMP).

This option enables you to configure the SNMP agent on the EMS server and determines whether or not to forward system alarms from the EMS server to the NMS.

To configure SNMP Agent:

- 1. From the Network Configuration menu, choose **SNMP Agent**, and then press Enter.
- 2. Enter the NMS IP.
- Enter the Community string.
 The new configuration is applied.

17.8 Server SNMPv3 Engine ID

The EMS server Engine ID is used by the SNMPv3 protocol when alarms are forwarded from the EMS to an NMS. By default, the EMS server SNMPv3 Engine ID is automatically created from the EMS server IP address. This option enables the user to customize the EMS server Engine ID according to their NMS configuration.

To configure the SNMPv3 Engine ID:

1. From the Network Configuration menu, choose **SNMPv3 Engine ID**, and then press Enter; the following is displayed:



Figure 17-13: EMS Server Manager – Configure SNMPv3 Engine ID



- 2. Enter '12' separate bytes ranges of the Engine ID (each valid range from between -128 to 127). In each case, press Enter to confirm the current value insertion and then proceed to the next one.
- 3. When all Engine ID bytes are provided, type **y** to confirm the configuration. To return to the root menu of the EMS Server Manager, press **q**.

Figure 17-14: SNMPv3 Engine ID Configuration – Complete Configuration

Server's SNMPv3 Engine ID (0 in all values return to default config	
	uration)
Byte[0] (valid range -128 127):21	
Byte[1] (valid range -128 127):23	
Byte[2] (valid range -128 127):2	
Byte[3] (valid range -128 127):5	
Byte[4] (valid range -128 127):3	
Byte[5] (valid range -128 127):78	
Byte[6] (valid range -128 127):-17	
Byte[7] (valid range -128 127):-56	
Byte[8] (valid range -128 127):121	
Byte[9] (valid range -128 127):117	
Byte[10] (valid range -128 127):-111	
Byte[11] (valid range -128 127):127	
Engine ID: 21.23.2.5.3.781756.121.117111.127 Are you sure that you want to continue? (y/n/q)	

18 Date and Time Settings

This option enables you to change the system time and date.

To change system time and date:

From the EMS Server Management root menu, choose Date & Time, and then press Enter; the following is displayed:

Figure 18-1: EMS Server Manager - Change System Time & Date



This menu includes the following options:

- NTP
- Timezone Settings
- Date & Time Settings

See Chapter 18 on page 140.

18.1 NTP

Network Time Protocol (NTP) is used to synchronize the time and date of the EMS server (and all its components) with other devices in the IP network.

This option enables you to configure the EMS server to obtain its clock from an external NTP clock source and other devices that are connected to the EMS server in the IP network can synchronize with this clock source. These devices can be any device containing an NTP server or client.

Alternatively, you can configure the NTP server to allow other devices in the IP network to synchronize their clocks according to the EMS server clock.

Note:

• It is recommended to configure the EMS server to synchronize with an external clock source because the EMS server clock is less precise than other NTP devices.



- When working with the Session Experience Manager (SEM), you should configure the same NTP server on both the EMS server and the AudioCodes device.
- When connecting the Lync Front-End server to the SEM, ensure that the same NTP server clock is used on both the EMS server and Microsoft Lync server.
- If you configure NTP server on the device, it is recommended to configure the same NTP server settings on the device and the EMS server.

To configure NTP:

1. From the Date & Time menu, choose **NTP**, and then press Enter; the following is displayed:

Figure 18-2: EMS Server Manager - Configure NTP

El	MS Server 7.2.1	196 Mana	igeme:	nt				
Main Menu> Date a	Main Menu> Date & Time> NTP							
Current N Allow/Rest	TP status: ON trict access to	NTP cli	ients	: Al	low			
remote	refid	st t wi	nen p	011	reach	delay	offset	jitter
60-56-214-78f2. 106.247.248.106 >1.Config 2.Stop N 3.Restric 4.Activa	.INIT. .INIT. ure NTP IP ct access to N1 te DDoS protect	16 u 16 u IP client	- - :s	64 64	0 0	0.000 0.000	0.000 0.000	0.000 0.000
6.Remove b.Back q.Quit to	authorized sublet	onet from	n NTP	rul	es			

- 2. From the NTP menu, choose option **1** to configure NTP.
- **3.** At the prompt, do one of the following:
 - Type **y** for the EMS server to act as both the NTP server and NTP client. Enter the IP addresses of the NTP servers to serve as the clock reference source for the NTP client (Up to four NTP servers can be configured).
 - Type n for the EMS server to act as the NTP server only. The EMS server is configured as a Stand-alone NTP server. The NTP process daemon starts and the NTP status information is displayed on the screen.

18.1.1 Stopping and Starting the NTP Server

This section describes how to stop and start the NTP server.

To start NTP services:

- From the NTP menu, choose option 2, and then choose one of the following options:
 - If NTP Service is on: Stop NTP
 - If NTP Service is off: Start NTP

The NTP daemon process starts; when the process completes, you return to the NTP menu.

18.1.2 Restrict Access to NTP Clients

This section describes how to restrict access to NTP clients.

To allow access to NTP clients:

From the NTP menu, choose option 3 to allow or restrict access to NTP clients; the screen is updated accordingly.

18.2 Timezone Settings

This option enables you to change the timezone of the EMS server.



Note: The Apache server is automatically restarted after the timezone changes are confirmed.

To change the system timezone:

- 1. From the Date & Time menu, choose **Time Zone Settings**, and then press Enter.
- 2. Enter the required time zone.
- **3.** Type **y** to confirm the changes; the EMS server restarts the Apache server for the changes to take effect.

18.3 Date and Time

This option enables you to set the date and time.

To set the date and time:

1. From the Date & Time menu, choose **Date & Time Settings**, and then press Enter; the current server time is displayed:

Figure 18-3: Change System Time and Date Prompt

```
Server's Time Is: [23/10/2013 09:56:38]
New Time (mmddHHMMyyyy.SS) []:
```

2. Enter the new time as shown in the following example:

```
mmddHHMMyyyy.SS :
month(08),day(16),Hour(16),Minute(08),year(2007),"."
Second.
```

19 Security

The EMS Management security options enable you to perform security actions, such as configuring the SSH Server Configuration Manager, and user's administration.

To configure security settings:

From the EMS Server Manager root menu, choose Security, and then press Enter, the following is displayed:

Figure 19-1: Security Settings

EMS Server 7.2.3035 Management
Main Menu> Security
>1.Add EMS User
2.SH
3.DB Password (EMS & SEM applications will be stopped)
4.0S Users Passwords
5.File Integrity Checker
6.Software Integrity Checker (AIDE) and Prelinking
7.USB Storage
8.Network options
9.Audit Agent Options (The server will be rebooted)
10.HTTPS Authentication
11.Disable SEM client secured communication (EMS application will be restarted)
12.Enable IP Phone Manager client and JAWS secured communication (Apache will be restarted)
13.Server Certificates Update
14.SEM - AudioCodes devices communication
q.Quit to main Menu

This menu includes the following options:

- Add EMS User (see Section 19.1 on page 142).
- SSH (see Section 19.2 on page 135).
- DB Password (EMS and SEM applications will be stopped) (see Section 19.3 on page 151).
- OS Users Password (see Section 19.4on page 152).
- File Integrity Checker (see Section 19.5 on page 155).
- Software Integrity Checker (AIDE) and Pre-linking (see Section 19.6 on page 155).
- USB Storage (see Section 19.7 on page 156).
- Network options (see Section 19.8 on page 156).
- Audit Agent Options (see Section 19.9 on page 157).
- HTTPS Authentication (see Section 19.10.1 on page 159).
- Enable SEM client secured connection (EMS application will be restarted) (see Section 19.10.2 on page 160).

- Enable IP Phone Manager client and JAWS secured communication (Apache will be restarted) (see Section 19.10.3 on page 160).
- Server Certificates Update (see Section 19.10.4 on page 160)
- SEM-AudioCodes devices communication (see Section 19.10.2 on page 168).

19.1 EMS User

This option enables you to add a new administrator user to the EMS server database. This user can then log into the EMS client. This option is advised to use for the operator's definition only in cases where all the EMS application users are blocked and there is no way to perform an application login.

To add an EMS user:

- 1. From the Security menu, choose Add EMS User, and then press Enter.
- 2. Enter the name of the user you wish to add.
- 3. Enter a password for the user.
- 4. Type y to confirm your changes.



Note: Note and retain these passwords for future access.

19.2 SSH

This section describes how to configure the EMS server SSH connection properties using the SSH Server Configuration Manager.

To configure SSH:

1. From the Security menu, choose **SSH**; the following is displayed:

Figure 19-2: SSH Configuration



This menu includes the following options:

- Configure SSH Log Level (see Section 19.2.1 on page 143).
- Configure SSH Banner (see Section 19.2.2 on page 144).
- Configure SSH on Ethernet Interfaces (see Section 19.2.3 on page 145).
- Disable SSH Password Authentication (see Section 19.2.4 on page 147).
- Enable SSH Ignore User Known Hosts Parameter (see Section 19.2.5 on page 147).
- Configure SSH Allowed Hosts (see Section 19.2.6 on page 148).

19.2.1 SSH Log Level

You can configure the log level of the SSH daemon server. The log files are found at the location '/var/log/secure' (older records are stored in secure.1, secure.2 etc.).

To configure the SSH Log Level:

1. From the SSH menu, choose option **1**, and then press Enter; the following is displayed:



EMS Server 7.2.126 Management
Main Menu> Security> SSH> Configure SSH Log Level
<pre>#LogLevel INFO Note: Changing LogLevel will restart SSH >1</pre>

Figure 19-3: SSH Log Level Manager

2. To configure the desired log level, choose the number corresponding to the desired level from the list, and then press Enter.

The SSH daemon restarts automatically.

The Log Level status is updated on the screen to the configured value.

19.2.2 SSH Banner

The SSH Banner displays a pre-defined text message each time the user connects to the EMS server using an SSH connection. You can customize this message. By default this option is disabled.

> To configure the SSH banner:

1. From the SSH menu, choose option **2**, and then press Enter; the following is displayed:

EMS Server 7.2.126 Management
Main Menu> Security> SSH> Configure SSH Banner
Current Banner State: <mark>DISABLED</mark> To change SSH Banner, please, change ⁄etc⁄issue file. Note: Changing Banner state will restart SSH
>1. <mark>Enable SSH Banner</mark> b.Back g.Quit to main Menu

Figure 19-4: SSH Banner Manager
- 2. Edit a '/etc/issue' file with the desired text.
- Choose option 1 to enable or disable the SSH banner.
 Whenever you change the banner state, SSH is restarted.
 The 'Current Banner State' is displayed in the screen.

19.2.3 SSH on Ethernet Interfaces

You can allow or deny SSH access separately for each network interface enabled on the EMS server.

To configure SSH on Ethernet interfaces:

From the SSH menu, choose option 3, and then press Enter; the following is displayed:

Figure 19-5:	Configure SSH	on Ethernet	Interfaces
--------------	---------------	-------------	------------



This menu includes the following options:

- Add SSH to All Ethernet Interfaces (see Section 19.2.3.1 on page 145.
- Add SSH to Ethernet Interface (see Section 19.2.3.2 on page 146).
- Remove SSH from Ethernet Interface (see Section 19.2.3.3 on page 146).

19.2.3.1 Add SSH to All Ethernet Interfaces

This option enables SSH access for all network interfaces currently enabled on the EMS server.

To add SSH to All Ethernet Interfaces:

From the Configure SSH on Ethernet Interfaces menu, choose option 1, and then press Enter.

The SSH daemon restarts automatically to update this configuration action. The column 'SSH Listener Status' displays ALL for all interfaces.

19.2.3.2 Add SSH to Ethernet Interface

This option enables you to allow SSH access separately for each network interface.

To add SSH to Ethernet Interfaces:

1. From the Configure SSH on Ethernet Interfaces menu, choose option **2**, and then press Enter.

After entering the appropriate sub-menu, all the interfaces upon which SSH access is currently disabled are displayed.

Enter the appropriate interface number, and then press Enter.
 The SSH daemon restarts automatically to update this configuration action.
 The column 'SSH Listener Status' displays 'YES' for the configured interface.

19.2.3.3 Remove SSH from Ethernet Interface

This option enables you to deny SSH access separately for each network interface.

To deny SSH from a specific Ethernet Interface:

1. From the Configure SSH on Ethernet Interfaces menu, choose option **3**, and then press Enter.

All the interfaces to which SSH access is currently enabled are displayed.

Enter the desired interface number, and then press Enter.
 The SSH daemon restarts automatically to update this configuration action.
 The column 'SSH Listener Status' displays 'No' for the denied interface.



Note: If you attempt to deny SSH access for the only enabled interface, a message is displayed informing you that such an action is not allowed.

19.2.4 Enable/Disable SSH Password Authentication

This option enables you to disable the username/password authentication method for all network interfaces enabled on the EMS server.

To disable SSH Password Authentication:

1. From the SSH menu, choose option **4**, and then press Enter; the following is displayed:

Figure 19-6: Disable Password Authentication



2. Type **y** to disable SSH password authentication or **n** to enable, and then press Enter.

The SSH daemon restarts automatically to update this configuration action.



Note: Once you perform this action, you cannot reconnect to the EMS server using User/Password authentication. Therefore, before you disable this authentication method, ensure that you provision an alternative SSH connection method. For example, using an RSA keys pair. For detailed instructions on how to perform such an action, see <u>www.junauza.com</u> or search the internet for an alternative method.

19.2.5 Enable SSH IgnoreUserKnownHosts Parameter

This option enables you to disable the use of the '\$HOME/.ssh/known_host' file with stored remote servers fingerprints.

To enable SSH IgnoreUserKnowHosts parameter:

1. From the SSH menu, choose option **5**, and then press Enter; the following is displayed:

Figure 19-7: SSH IgnoreUserKnowHosts Parameter - Confirm

Enable SSH IgnoreUserKnownHosts parameter:	
Current SSH IgnoreUserKnownHosts parameter value is NO.	
Are you sure you want to Change SSH IgnoreUserKnownHosts value to YES?(y/n) y	

2. Type **y** to change this parameter value to either 'YES' or 'NO' or type **n** to leave as is, and then press Enter.

19.2.6 SSH Allowed Hosts

This option enables you to define which hosts are allowed to connect to the EMS server through SSH.

To Configure SSH Allowed Hosts:

From the SSH menu, choose option 6, and then press Enter; the following is displayed:

EMS Server 7.2.126 Management Main Menu> Security> SSH> Configure SSH Allowed Hosts SSH Allowed for ALL Hosts. >1. Leny ALL Hosts 2. Add Host/Subnet to Allowed Hosts b.Back g.Quit to main Menu

Figure 19-8: Configure SSH Allowed Hosts

This menu includes the following options:

- Allow ALL Hosts (see Section 19.2.6.1 on page 148).
- Deny ALL Hosts (see Section 19.2.6.2 on page 149).
- Add Host/Subnet to Allowed Hosts (see Section 19.2.6.3 on page 149).
- Remove Host/Subnet from Allowed Hosts (see Section 19.2.6.4 on page 150).

19.2.6.1Allow ALL Hosts

This option enables all remote hosts to access this EMS server through the SSH connection.

To allow ALL Hosts:

- 1. From the Configure SSH Allowed Hosts menu, choose option 1, and then press Enter.
- **2.** Type **y** to confirm, and then press Enter.

The appropriate status is displayed in the screen.

19.2.6.2 Deny ALL Hosts

This option enables you to deny all remote hosts access to this EMS server through the SSH connection.

To deny all remote hosts access:

- 1. From the Configure SSH Allowed Hosts menu, choose option **2**, and then press Enter.
- **2.** Type **y** to confirm, and then press Enter.

The appropriate status is displayed in the screen.



Note: When this action is performed, the EMS server is disconnected and you cannot reconnect to the EMS server through SSH. Before you disable SSH access, ensure that you have provisioned alternative connection methods, for example, serial management connection or KVM connection.

19.2.6.3 Add Hosts to Allowed Hosts

This option enables you to allow different SSH access methods to different remote hosts. You can provide the desired remote host IP, subnet or host name in order to connect to the EMS server through SSH.

To add Hosts to Allowed Hosts:

1. From the Configure SSH Allowed Hosts menu, choose option **3**, and then press Enter; the following is displayed:

Figure 19-9: Add Host/Subnet to Allowed Hosts

```
EMS Server 7.2.126 Management

Main Menu> Security> SSH> Configure SSH Allowed Hosts> Add Host/Subnet to Allow

ed Hosts

>1.Add IP Address (x.x.x.x)

2.Add Subnet (n.n.n.n/m.m.m.m - network/netmask)

3.Add Host Name (without "/" or "," characters)

b.Back

q.Quit to main Menu
```

- 2. Choose the desired option, and then press Enter.
- 3. Enter the desired IP address, subnet or host name, and then press Enter.

Note: When adding a Host Name, ensure the following:

- Verify your remote host name appears in the DNS server database and your EMS server has an access to the DNS server.
- Provide the host name of the desired network interface defined in "/etc/hosts" file.
- 4. Type **y** to confirm the entry, and then press Enter again.

If the entry is already included in the list of allowed hosts, an appropriate notification is displayed.

When the allowed hosts entry has been successfully added, it is displayed in the SSH Allow/Deny Host Manager screen as shown in the figure below:

Figure 19-10: Add Host/Subnet to Allowed Hosts-Configured Host



19.2.6.4 Remove Host/Subnet from Allowed Hosts

If you have already configured a list of allowed hosts IP addresses, you can then remove one or more of these host addresses from the list.

To remove an existing allowed host's IP address:

- 1. From the Configure SSH Allowed Hosts menu, choose option 1, and then press Enter; the following is displayed:
- 2. Choose the desired entry to remove from the Allowed Hosts list, i.e. to deny access to the EMS server through SSH connection, and then press Enter again.
- **3.** Type **y** to confirm the entry, and then press Enter again.

When the allowed hosts entry has been successfully removed, it is displayed in the SSH Allow/Deny Host Manager screen as shown in the figure below:



Note: When you remove either the only existing IP address, Subnet or Host Name in the Allowed Hosts in the Allowed Hosts list, there are no remote hosts with access (i.e. for each respective option) to connect to the EMS server using SSH. When this action is performed, you are disconnected from the EMS server and may not be able to reconnect through SSH. Therefore, prior to disabling SSH access, ensure that alternative connection methods have been provisioned, for example, serial management connection or KVM connection.

19.3 DB Password

This option enables you to change the default Oracle Database password "pass_1234". The EMS server shuts down automatically before changing the Oracle Database password.

To change the DB Password:

- 1. From the Security menu, choose **DB Password**, and then press Enter; the EMS server is rebooted.
- 2. Press Enter until the New Password prompt is displayed.

Figure 19-11: EMS Server Manager – Change DB Password



3. Enter the new password, which should contain at least one digit, one character and one punctuation.



Note:

- The EMS server is rebooted when you change the Oracle Database password.
- Note and retain these passwords for future access. It is not possible to restore these passwords or to enter the EMS Oracle Database without them.
- **4.** After validation, a message is displayed indicating that the password was changed successfully.

19.4 OS Users Passwords

This section describes how to change the OS password settings.

To change OS passwords:

- 1. From the Security menu, choose **OS Users Passwords**, and then press Enter.
- 2. Proceed to one of the following procedures:
 - General Password Settings (see Section 19.4.1 on page 152.
 - Operating System User Security Extensions (see Section 19.4.2 on page 153).

19.4.1 General Password Settings

This option enables you to change the OS general password settings, such as 'Minimum Acceptable Password Length' and 'Enable User Block on Failed Login'. This feature also enables you to modify settings for a specific user, such as 'User's Password' and 'Password Validity Max Period'.

To modify general password settings:

1. The Change General Password Settings prompt is displayed; type **y**, and then press Enter.

Do you want to change general password settings? $(y/n)\mathbf{y}$

2. The Minimum Acceptable Password Length prompt is displayed; type **10**, and then press Enter.

Minimum Acceptable Password Length [10]: 10

3. The Enable User Block on Failed Login prompt is displayed; type **y**, and then press Enter.

Enable User Block on Failed Login (y/n) [y] **y**

4. The Maximum Login Retries prompt is displayed; type 3, and then press Enter.

Maximum Login Retries [3]: 3

5. The Failed Login Locking Timeout prompt is displayed; type **900**, and then press Enter.

Failed Login Locking Timeout [900]:900

6. You are prompted if you wish to continue; type **y**, and then press Enter.

Are you sure that you want to continue? (y/n/q) ${\boldsymbol y}$

19.4.2 Operating System Users Security Extensions

This feature enables the administrator to configure the following additional user security extensions:

- Maximum allowed numbers of simultaneous open sessions.
- Inactivity time period (days) before the OS user is locked.

To configure these parameters, in the OS Passwords Settings menu, configure parameters according to the procedure below (see also green arrows indicating the relevant parameters to configure in Figure 19-12).

To configure operating system users security extensions:

1. The Change General Password Settings prompt is displayed; type **n**, and then press Enter.

Do you want to change general password settings ? (y/n) **n**

2. The Change password for a specific user prompt is displayed; type **y**, and then press Enter.

Do you want to change password for specific user ? (y/n) ${\boldsymbol{y}}$

3. Enter the Username upon which you wish to configure, and then press Enter.

Enter Username [acems]:

4. The change User Password prompt is displayed; type **n**, and then press Enter.

Do you want to change its password ? (y/n) \boldsymbol{n}

5. An additional Password prompt is displayed, type **y**, and then press Enter.

Do you want to change its login and password properties? (y/n) \boldsymbol{y}

6. The Password Validity prompt is displayed; press Enter.

Password Validity Max Period (days) [90]:

7. The Password Update prompt is displayed; press Enter.

Password Update Min Period (days) [1]:

8. The Password Warning prompt is displayed; press Enter.

Password Warning Max Period (days) [7]:

 The Maximum number of Simultaneous Open Sessions prompt is displayed; enter the number of simultaneous open SSH connections you wish to allow for this user.

Maximum allowed number of simultaneous open sessions [0]:

10. The Inactivity Days prompt is displayed; enter the number of inactivity days before the user is locked. For example, if you'd like to suspend a specific user if they have not connected to the EMS server for a week, enter 7 days.

Days of inactivity before user is locked (days) [0]:

Figure 19-12: OS Passwords Settings with Security Extensions

OS Passwords Settings	
Do you want to change general password settings? (y/n)	n
Do you want to change password for specific user? (y/n Enter Username [acems]: testuser) Y
Do you want to change its password ? (y/n) n	
Do you want to change its login and password propertie Password Validity Max Period (days) [90]: Password Update Min Period (days) [1]: Password Warning Max Period (days) [7]: Maximum allowed number of simultaneous open sessions [Days of inactivity before user is locked (days) [0]: 3 Are you sure that you want to continue? (y/n/q) y Adjusting aging data for user testuser. passwd: Success Done.	s? (y/n) y 0]: 3

If the user attempts to open more than three SSH sessions simultaneously, they are prompted and immediately disconnected from the fourth session as displayed in the figure below.







Note: By default you can connect through SSH to the EMS server with user *acems* **only**. If you configure an inactivity days limitation on this user, the situation may arise, for example, where a user is away for an extended period and has no active user to access the EMS server. Therefore, we strongly recommend to use this limitation very carefully and preferably to configure this option for each user to connect to the EMS server through SSH other than with the *acems* user.

19.5 File Integrity Checker

The File Integrity checker tool periodically verifies whether file attributes were changed (permissions/mode, inode #, number of links, user id, group id, size, access time, modification time, creation/inode modification time). File Integrity violation problems are reported through EMS Security Events. The File Integrity checker tool runs on the EMS server machine.

From the Security menu, choose File Integrity Checker, and then press Enter; the File Integrity Checker is started or stopped.

19.6 Software Integrity Checker (AIDE) and Pre-linking

AIDE (Advanced Intrusion Detection Environment) is a file and directory integrity checker. This mechanism creates a database from the regular expression rules that it finds in its configuration file. Once this database is initialized, it can be used to verify the integrity of the files.

Pre-linking is designed to decrease process startup time by loading each shared library into an address for which the linking of needed symbols has already been performed. After a binary has been pre-linked, the address where the shared libraries are loaded will no longer be random on a per-process basis. This is undesirable because it provides a stable address for an attacker to use during an exploitation attempt.

To start AIDE and disable pre-linking:

1. From the Security menu, choose **Software Integrity Checker (AIDE) and Prelinking**; the current status of these two processes is displayed:

Figure 19-14: Software Integrity Checker (AIDE) and Pre-linking



- 2. Do one of the following:
 - Type **y** to enable AIDE and disable pre-linking
 - Type **n** to disable AIDE and enable pre-linking.

19.7 USB Storage

This menu option allows enabling or disabling the EMS Server's USB storage access as required.

To enable USB storage:

1. From the Security menu, choose **USB Storage**; the following prompt is displayed:





2. Enable or disable USB storage as required.

19.8 Network Options

This menu option provides the following options to enhance network security:

Ignore Internet Control Message Protocol (ICMP) Echo requests:

This option ensures that the EMS server does not respond to ICMP broadcasts, and therefore such replies are always discarded. This prevents attempts to discover the system using ping requests.

Ignore ICMP Echo and Timestamp requests:

This option ensures that the EMS server does not respond to an ICMP timestamp request to query for the current time. This reduces exposure to spoofing of the system time.

Send ICMP Redirect Messages:

This option disables the sending of ICMP Redirect Messages, which are generally sent only by routers.

Ignore ICMP Redirect Messages:

This option ensures that the EMS server does not respond to ICMP Redirect broadcasts, and therefore such replies are always discarded.

This prevents an intruder from attempting to redirect traffic from the EMS server to a different gateway or a non-existent gateway.

To enable network options:

 From the Security menu, choose Network Options; the following screen is displayed:

```
Figure 19-16: Network Options
```

```
EMS Server 7.2.129 Management

Main Menu> Security> Network options

|Log packets with impossible addresses to kernel log: DISABLED

|Ignore all ICMP ECHO requests: DISABLED

|Ignore all ICMP ECHO and TIMESTAMP requests: DISABLED

|Send ICMP redirect messages: DISABLED

|Accept ICMP redirect messages: DISABLED

>1. Enable log packets with impossible addresses to kernel log

2.Enable ignore all ICMP ECHO requests

3.Enable Ignore all ICMP ECHO requests

4.Enable send ICMP redirect messages

5.Enable accept ICMP redirect messages

b.Back

q.Quit to main Menu
```

2. Set the required network options.

19.9 Auditd Options

Auditid is the userspace component to the Linux Auditing System that is responsible for writing audit records to the disk. Using the Auditd option, you can change the auditd tool settings to comply with the Security Technical Information Guidelines (STIG) recommendations.

To set Auditd options according to STIG:

1. From the Security menu, choose **Auditd Options**; the following screen is displayed:





 Enable or disable Auditd options as required. Audit records are saved in the following /var/log/audit/ directory.

19.10 HTTPS/SSL/TLS Security

This section describes the configuration settings for the HTTPS/SSL/TLS connections. The figure below shows the maximum security that can be implemented in the OVOC environment. For most connections, the HTTPS/SSL/TLS protocols can be implemented; those connections where these protocols are not supported are indicated in red.





Endpoints



Note: The above figure shows all the HTTPS/SSL/TLS connections in the OVOC network. Use this figure as an overview to the procedures described below. Note that not all of the connections shown in the above figure have corresponding procedures. For more information, refer to the OVOC Security Guidelines document..

19.10.1 HTTPS Authentication

This option enables you to configure whether certificates are used to authenticate the connection between the EMS server and the devices in one direction or in both directions:

- Mutual Authentication: the EMS authenticates the device connection request using certificates and the device authenticates the EMS connection request using certificates. When this option is configured:
 - The same root CA must sign the certificate that is loaded to the device and certificate that is loaded to the EMS server.
 - Mutual authentication must also be enabled on the device (see Section E.5.2.5).
- One-way Authentication option: the EMS does not authenticate the device connection request using certificates; only the device authenticates the EMS connection request.



Note: You can use the procedure described in Section 19.10.4 to load the certificate file to the EMS server.

> To enable HTTPS authentication:

1. In the Security menu, choose the HTTPS Authentication option.

19-19: HTTPS Authentication



- 2. Choose one of the following options:
 - 1-Set Mutual Authentication
 - 2. Set One-Way Authentication

19.10.2 Enable SEM Client Secured Connection

This menu option enables you to secure the connection between the SEM client browser and the Tomcat server over HTTPS. When this option is enabled, the connection is secured through HTTPS port 9400 (instead of port 8400-HTTP).

To enable a secure connection between SEM client browser and Tomcat server:

From the Security menu, choose Enable SEM client secured connection; the connection is secured.

19.10.3 Enable IP Phone Manager Client JAWS and NBIF Secured Communication

This menu option enables you to secure the connection between the IP Phone Manager client browser, JAWS/NBIF clients and the Apache server over HTTPS. When this option is enabled, the connection is secured through HTTPS port 443 (instead of port 80-HTTP).

To enable a secure connection between the IP Phone Manager client browser/JAWS and EMS server:

From the Security menu, choose IP Phone Manager client and JAWS secured communication; the connection is secured.

19.10.4 Server Certificates Update

This menu option enables you to automatically generate custom SSL server certificates.



Note: If you are using self-generated certificates and private key, you can skip to step 4.

The procedure for server certificates update consists of the following steps:

- 1. Step 1: Generate Server Private Key.
- 2. **Step 2:** Generate Server Certificate Signing Request (CSR).
- 3. Step 3: Transfer the generated CSR file to your PC and send to CA.
- 4. **Step 4:** Transfer certificates files received from CA back to EMS server.
- 5. Step 5: Import new certificates on EMS server.
- 6. **Step 6:** Verify the installed Server certificate.
- 7. **Step 7:** Verify the installed Root certificate.

- 8. **Step 8:** Update Client Java Web Start certificates (when using non-default certificates, following an upgrade of the EMS server).
- 9. Step 9: Perform Supplementary procedures to complete certificate update process (refer to Appendix E).

To generate server certificates:

1. From the Security menu, choose Server Certificates Update.

19-20:Server Certificate Updates

Main Menu> Security> Server Certificates Update
Server's Certificate: Not Default >1. Cenerate Server Private Key 2. Generate Server Certificate Signing Request (CSR) 3. Import Server Certificates from Certificate Authority (CA) 4. Update Client (JAWS) Certificates 5. Display installed Server Certificate 6. Display installed Root Certificate b. Back q.Quit to main Menu

Information on the currently installed certificate is displayed (the currently installed certificate is the installation default).

Step 1: Generate a server private key:

1. Select option **1**. The following screen is displayed:

Figure 19-21: Generate Server Private Key



- 2. Select the number of bits required for the server private key.
- Enter and reenter the server private key password and type Y to continue. The private key is generated.



Generating a Server Private Key:
This will override the existing private key and render the existing certificates invalid until new certificates are imported.
Are you sure you want to generate a new private key? (N/y)y
Select Number Of bits for Private Key:
1. 1024
2. 2048
3. 4096
g. guit and return to menu
Select number: 1
Enter private key password:
Re-enter private key password:
Ready to generate server mrivate key. Continue? (n/Y) : u
Generating RSA private key, 1024 hit long modulus
++++++

- in (EE27 (A×10001)
E 12 03221 (0710001)
Done generating private key. Press Enter to go back to the menu

Step 2: Generate a CSR for the server:

- 1. Select option 2.
- 2. Enter the private key password (the password that you entered in the procedure above).
- 3. Enter the Country Name code, state or province, locality, organization name, organization unit name, common name (server host name) and email address.
- 4. Enter a challenge password and optionally a company name.

You are notified that a server Certificate Signing Request has successfully been generated and saved to the specified location.

Figure 19-23: Generating a Server Certificate Signing Request (CSR)

```
Generating a Server Certificate Signing Request (CSR):
Enter the passphrase used in the server private key:
You are about to be asked to enter information that will be incorporated
into your certificate request.
What you are about to enter is what is called a Distinguished Name or a DN.
There are quite a few fields but you can leave some blank
For some fields there will be a default value,
If you enter '.', the field will be left blank.
Country Name (2 letter code) [GB]:GB
State or Province Name (full name) [Berkshire]:Berkshire
Locality Name (eg, city) [Newbury]:Newbury
Organization Name (eg, company) [My Company Ltd]:EA1
Organizational Unit Name (eg, section) []:Finance
Common Name (eg, your name or your server's hostname) []:EA1
Email Address []:Bradb@enterpriseA.com
Please enter the following 'extra' attributes
to be sent with your certificate request
A challenge password []:
```

Step 3: Transfer the CSR file to your PC and send to CA:

Transfer the CSR file from the /home/acems/server_cert/server.csr directory to your PC and then sent it to the Certificate Authority (CA). For instructions on transferring files, see Appendix F.

Figure 19-24: Transfer CSR File to PC

Step 4: Transfer server certificates from the CA:

Transfer the files that you received from the CA to the /home/acems/server_certs directory. The root certificate should have the name root.crt and that the server certificate should have the name server.crt. If you received intermediate certificates, then rename them to ca1.crt and ca2.crt. Make sure that all certificates are in PEM format.

For instructions on transferring files, see Appendix F.



Note: If your certificates are self-generated (you did not perform steps 1-3), the /home/acems/server_certs directory does not exist; create it using the following commands:

mkdir /home/acems/server_certs

chmod 777 /home/acems/server_certs

Step 5: Import certificates:

Select option **3** and follow the prompts.

The certificate file is installed.

AudioCodes

Note:

- If you have installed an HA system and wish to install Custom server certificates, the HA system must first be uninstalled, and then you must perform this procedure separately on both server machines (as stand-alone machines).
- The root certificate should be named root.crt and that the server certificate should be named server.crt. If you received intermediate certificates then rename them to ca1.crt and ca2.crt.
- Make sure that all certificates are in PEM format and appear as follows:



```
-----BEGIN CERTIFICATE-----
MIIBuTCCASKGAwIBAGIFAKK1MbgwDQYJKoZIhvcNAQEFBQAwFzEVMBMGAlUEAxM
M
RU1TIFJPT1QgQ0EyMB4XDTE1MDUwMzA4NTE0MFoXDTI1MDUwMzA4NTE0MFowKjE
T
...
Tl6vqn51270q/24KbY9q6EK2Yc3K2EAadL2IF1jnb+yvREuewprOz6TEEuxNJol
0
L6V81zUYOfHrEiq/6g==
-----END CERTIFICATE-----
```

Step 6: Verify the installed server certificate:

Select option 4.

The installed server certificate is displayed:



Installed Server Certificate:
Certificate: Data: Uersion: 3 (0x2) Serial Number: 2416025747 (0x9001a093) Signature Algorithm: md5WithRSAEncryption Issuer: CN=EMS ROOT CA2 Validity Not Before: Feb 20 19:15:13 2010 GMT Not After : Feb 20 19:15:13 2020 GMT Subject: 0-AudioCodes, CN=EMS Server Subject: Public Key Info: Public Key Algorithm: rsaEncryption RSA Public Key: (1024 bit) Modulus (1024 bit): 00:d2:45:b7:4e:de:ba:0a:38:d9:fb:72:2a:c3:f2: 15:4a:c9:e1:e1:e7:bf:3f:20:52:fd:3c:43:9a:43: 7a:50:ad:a1:d5:b0:41:56:6c:7d:11:b4:23:6d:c8: 9f:d1:2b:41:94:ee:e1:63:33:90:a9:73:b3:94:2a: f6:d6:27:31:27:df:64:d0:c2:8c:62:6d:35:d7:0e: 26:09:5d:c0:71:e3:94:8e:60:b2:55:02:bd:ad:75: ef:3d:b2:94:8d:46:0d:c8:d5:be:b1:2f:4d:dd:bc:
More
Step 7: Verify the installed root certificate:
Select Option 5. The installed root certificate is displayed:
Figure 19-26: Installed Root Certificate
Installed Server Root Certificate Chain:
Certificate:
Data: Version: 3 (0x2) Serial Number: 2416023367 (0x90019747) Signature Algorithm: md5WithRSAEncryption Issuer: CN=EMS ROOT CA Validity
Not Before: Feb 20 18:54:27 2010 GMT Not After : Feb 20 18:54:27 2020 GMT Subject: CN=EMS ROOT CA2 Subject Public Key Info: Public Key Algorithm: rsaEncryption RSA Public Key: (1024 bit)
Modulus (1024 bit): 00:bc:dd:d6:eb:71:c8:79:de:f4:12:31:51:21:e6: 7b:e9:3a:a3:9f:10:bc:4c:37:90:1d:da:4a:40:58: 36:bb:43:f7:bb:c5:80:02:9e:66:21:7f:20:cc:48: c4:40:4a:ad:07:3b:48:3c:31:7a:db:9c:7c:a9:3e: 76:f8:e9:d2:1a:40:c1:7d:db:16:18:67:66:34:13: 50:74:08:ec:5b:3d:75:37:8a:d7:53:b2:59:a9:ff: a2:f2:23:2b:58:2c:b8:78:99:df:ca:3e:66:60:99:

--More--



Step 8: Update Client Java Web Start certificates:



Note:

•

- This option is only relevant when using non-default certificates, following an upgrade of the EMS server.
- The other components of the certificate configuration are maintained following an upgrade of the EMS server.
- 1. Select Option 4.

Figure 19-27: Update Client Java Web Start Certificates



2. Enter the path of the Java KeyStore file (press Enter to use the default path).





3. Enter the passphrase of the client KeyStore (default "password").





The Java files are updated. The progress of the updates are displayed on the screen.

- Step 9: Perform Supplementary procedures to complete certificate update process
- Refer to the supplementary procedures in Appendix E on page 241 to complete the certificate update process.

19.10.5 SEM - AudioCodes Devices Communication

This option allows you to configure the transport type for the XML based SEM communication from the AudioCodes devices to the SEM server. You can enable the TCP port (port 5000), the TLS port (port 5001) connections or both port connections.

To configure the SEM - AudioCodes device communication:

1. From the EMS Server Manager Root menu, select **SEM – AudioCodes device** communication.

Figure 19-30: SEM - AudioCodes Device Communication



- 2. Choose one of the following transport types:
 - TCP (opens port 5000)
 - TLS (opens port 5001)
 - TLS/TCP (this setting opens both ports 5000 and 5001)

20 Diagnostics

This section describes the diagnostics procedures provided by the EMS server Manager.

To run EMS Server diagnostics:

- From the EMS Server Manager Root menu, choose Diagnostics, and then press Enter, the following is displayed:
 - Figure 20-1: Diagnostics



This menu includes the following options:

- Server Syslog Configuration (see Section 20.1 on page 171).
- Devices Syslog Configuration (see Section 20.2 on page 171).
- Devices Debug Configuration (see Section 20.3 on page 172).

20.1 Server Syslog Configuration

This section describes how to send EMS server Operating System (OS)-related syslog EMERG events to the system console and other EMS server OS related messages to a designated external server.

To send EMERG event to the syslog console and other events to an external server:

- 1. From the Diagnostics menu, choose **Server Syslog**, and then press Enter.
- 2. To send EMERG events to the system console, type **y**, press Enter, and then confirm by typing **y** again.



Figure 20-2: Syslog Configuration

```
Syslog configuration
Send EMERG events to system console: n
Forward messages to external server: n
Send EMERG events to system console ? (y/n) y
Logging of many events on console when RS-232 console is used may cause severe p
erformance degradation (due to 9600 baud rate).
Are you sure ? (y/n)
```



Forward messages to external server ? (y/n) y
Facility (choose from this list):
*
AUTH
AUTHPRIV
CRON
DAEMON
FTP
KERN
LOCALO
LOCAL1
LOCAL2
LOCAL3
LOCAL4
LOCAL5
LOCAL6
LOCAL7
LPR
MAIL
NEWS
SYSLOG
USER
UUCP
[]: AUTH
Severity (choose from this list):
EMERG
ALERT
CRIT
ERR
WARNING
NOTICE
INFO
DEBUG
[]: EMERG
Hostname[]: MY-SYSLOG-SERVER1

- **3.** You are prompted to forward messages to an external server, type **y**, and then press Enter.
- 4. Type the desired **Facility** from the list (case-sensitive), and then press Enter.
- 5. Type the desired **Severity**.
- 6. Type the external server Hostname or IP address.

20.2 Devices Syslog Configuration

The capture of the device's Syslog can be logged directly to the EMS server without the need for a third-party Syslog server in the same local network. The EMS server Manager is used to enable this feature.



Note: This feature is only relevant for CPE products. Syslog is captured according to the device's configured Syslog parameters. For more information, see the relevant device's *SIP User's manual*.

The user needs to also enable the monitored device to send syslog messages to the standard syslog port (UDP 514) on the EMS server machine.

The syslog log file 'syslog' is located in the following EMS server directory:

/data/NBIF/mgDebug/syslog

The syslog file is automatically rotated once a week or when it reaches 100 MB. Up to four syslog files are stored.

To enable device syslog logging:

- 1. From the Diagnostics menu, choose **Devices Syslog**, and then press Enter.
- You are prompted whether you wish to send EMER events to system console; type Y or N.
- You are prompted whether you wish to send events to an external server; type Y or N.

20.3 Devices Debug Configuration

Debug recordings packets from all managed machines can be logged directly to the EMS server without the need for a 3rd party network sniffer in the same local network.



Note: This feature is only relevant for CPE products. Debug recording packets are collected according to the device's configured Debug parameters. For more information, see the relevant device's *User's Manual*.

The EMS server runs the Wireshark network sniffer, which listens on a particular configured port. The sniffer records the packets to a network capture file in the Debug Recording (DR) directory. You can then access this file from your PC through FTP.

The EMS Server Manager is used to enable this feature. The user should configure the monitored device to send its debug record messages to a specific port (UDP 925) on the EMS server IP.

The DR capture file is located in the following EMS server directory:

/data/NBIF/mgDebug/DebugRecording

The file 'TPDebugRec<DATE>.cap' is saved for each session. The user is responsible for closing (stopping) each debug recording session. In any case, each session (file) is limited to 10MB or one hour of recording (the first rule which is met causes the file to close i.e. if the file reaches 10MB in less than an hour of recording, it is closed). A cleanup process is run daily, deleting capture files that are 5 days old.

The user is able to retrieve this file from the EMS server and open it locally on their own PC using Wireshark with the debug recording plug-in installed (Wireshark version 1.6.2 supports the Debug Recording plug-in).

To enable or disable devices debug:

- From the Diagnostics menu, choose **Devices Debug**, and then press Enter. A message is displayed indicating that debug recording is either enabled or disabled.
- 2. Type y, and then press Enter.

Recording files are saved in /data/NBIF/mgDebug directory on the server.



Note: It is highly recommended to disable the 'TP Debug Recording' feature when you have completed recording because this feature heavily utilizes system resources.

Part VI

HA (High Availability)

This section describes the EMS HA Configuration options.



21

Getting Started with HA (High Availability)

EMS servers High Availability is supported for EMS server applications running on the Linux platform.

Two EMS server machines are required to support High Availability: one machine serving as the Primary machine and the other serving as the Secondary machine. When the EMS application is active and running, all data stored in the EMS server machine and database is replicated from the Primary machine to the Secondary machine. Upon Primary machine failure recognition (either on the EMS application or on the Network), activity is automatically transferred from the Primary server machine to the Secondary server machine.

Two models of High Availability are supported:

- Both EMS servers are located in the same subnet. There is a single EMS server IP address - Global (Virtual) IP address defined for all the Network Components (EMS clients and Managed Gateways). Each of the EMS server machines has an internal Private IP address and the active EMS server machine performs binding to the Global (Virtual) IP address. This setup currently does not support working with gateways behind a NAT.
- Each one of the EMS servers is located in a different network subnet and has its own IP address. During the EMS client login dialog, the user should provision both IP addresses (Geo HA), and the EMS client application will constantly search for the currently active EMS server machine. All the managed gateways relevant applications (such as Trap Sending, NTP Server, and OCSP Server) should be aware of two possible EMS server machine addresses.

The HA Configuration menu option enables you to configure EMS server machines high availability, perform HA-related actions and review the HA status for both servers.

Prior to configuring HA, both machines should be installed with an identical EMS server version and an identical operating system and network configuration.

21.1 EMS HA Pre-requisites

Before implementing an EMS HA configuration, ensure that both EMS servers have an identical configuration according to the following:

- Both servers have identical hardware. See EMS Server and Client Requirements section for supported machines (see Chapter 3 on page 23).
- An identical Linux OS is installed on both servers.
- An identical EMS version is installed on both servers.
- An identical database password should be configured on both servers.
- An identical interface configuration and the same subnets are connected to each server (N/A for Geo HA).
- An identical redundancy configuration on identical interfaces.
- The EMS application is down (use the EMS Server Manager to shut down the EMS application).
- SSH communication between the Secondary and the Primary servers exists.
- Network Bandwidth requirements between two EMS servers are as follows:
 - Initial Synchronization process: at least 80 Mbps
 - During the initial sync process, the entire /data partition is synchronized between the active and redundant servers. This partition size is 1.7 TB on HP DL360p G8 servers. A network speed of at least 80 Mbps is required to complete the initial sync process in up to 5 hours on G8 servers.

Assuming a slower network, the process will take longer. For example, on G8 servers:

- 100Mbps -> up to 40 hours
- 1Gbps -> up to 4 hours
- Ongoing server Synchronization: 10 Mbps.
- Ping between two servers: the ping time between each EMS server machine should not exceed 200 msec.
- During the HA configuration process, entire /data partition is duplicated from the primary server to the secondary server. If any of the servers contain previous backup files, these files are deleted on the secondary server. These files should be backed up on an external storage machine prior to the HA configuration.
- If you are using custom certificates (see Appendix E on page 241), they must be preinstalled on both primary and secondary machines before commencing the HA process.

21.2 EMS HA Data Synchronization

The data synchronization is performed using a distributed replicated block device for the Linux operating system. This process allows a real-time mirror of the local block devices on a remote machine.

The replicated EMS data includes the following:

- EMS Database
- EMS NBIF files including the following:
 - Backup files
 - Alarms files
 - Topology files
 - Performance files
 - MG backup files
 - Debug recording
 - EMS Software files (EMS Software Manager files)
 - MG configuration files, for upgrade and management
 - MG Auxiliary files

The initial synchronization time between two EMS server machines is estimated at 1.5-4 hours, depending on network speed/quality and servers' disk size. Every change that is performed on the primary server is immediately synchronized to the secondary server.

21.2.1 Replicate EMS Server Manager Actions

Any actions performed using the EMS Server Manager prior to the HA configuration should be manually updated on both EMS server machines. EMS Server Manager actions are logged in the following file:

/var/log/ems/EmsServerManager.txt



Note: The EMS HA process does not automatically replicate actions executed using the EMS Server Manager on the primary server to the secondary server.

21.3 EMS Server Manager

This section describes specific details in reference to the maintenance procedures available in the EMS Server Manager.

The EMS Server Manager displays dynamic menus. Each menu is displayed differently according to the current server's state.

The following menu items are not displayed on the Primary server:

- Start/Stop Application
- Change Server's IP Address
- Configure Ethernet Interfaces

AudioCodes

- Configure Ethernet Redundancy
- Configure NAT
- Restore the EMS Server
- DB Password

The following menu items are not displayed on the Secondary server:

- Start/Stop Application
- Change Server's IP Address
- Configure Ethernet Interfaces
- Configure Ethernet Redundancy
- Configure NAT
- Add EMS User
- Restore the EMS Server
- DB Password

In some cases, the menu will only be updated after running EMS Server Manager again. For instance, after HA installation, the Start/Stop EMS Server option is hidden after exiting the EMS Server Manager and running it again.

21.4 EMS Client

Once the switchover has successfully completed, the EMS client logs in again to the active server and a "Server Startup" alarm is displayed.

21.5 EMS Server Upgrade

EMS server version upgrade cannot be performed while HA is configured.

To upgrade the servers, HA must be uninstalled prior to the upgrade. It is recommended to firstly uninstall the secondary server, and then the primary server.

- To uninstall HA, see Section 22.6 on page 192.
- To upgrade the EMS server, see Section 8.1 on page 81.

21.5.1 Upgrading to Version 7.2.1000

The procedure below is designated for customers with EMS HA installed on any version prior to 7.2.1000 who wish to upgrade their servers to 7.2.1000. This procedure requires the following actions:

- **1.** Upgrade of the EMS servers to Version 7.2.1000.
- 2. Backup of all data.
- 3. Upgrade of the Linux Operating System to CentOS 5.9 Rev8.
- 4. Restore of data to new installation.



Note: For maintaining EMS HA and all data, it is mandatory for you to perform the procedure below if you are upgrading to EMS Version 7.2.1000 CentOS 5.9 Rev8.

Do the following:

- 1. Uninstall HA from Secondary server and reboot it. After reboot make sure all processes are up in EMS Server Manager Status (see Chapter 13).
- 2. Uninstall HA from Primary server and reboot it. After reboot make sure all processes are up in EMS Server Manager Status (see Chapter 13).
- **3.** Upgrade each one of the servers to 7.2.1000. After reboot make sure all processes are up in Ems Server Manager Status.
- 4. On the server which was the Primary server prior to uninstalling HA, configure the server to create a new weekly backup on the next full hour e.g. 1200 (following the time of completing this procedure) using the Change Schedule Backup Time option (see Section 16.3).
- 5. Wait for backup process to complete. This can be verified by typing the following command:

ll /data/NBIF/emsBackup/RmanBackup/init.ora" command.

When the init.ora file recieves the current timestamp, this implies that the Backup process has completed. For example:

```
[root@EMS-12 ~]# ll
/data/NBIF/emsBackup/RmanBackup/init.ora
-rw-r--r-- 1 oracle dba 1449 May 2 02:08
/data/NBIF/emsBackup/RmanBackup/init.ora
```

6. Copy all backup files to an external server:

```
/data/NBIF/emsBackup/emsServerBackup_xxx.tar
/data/NBIF/emsBackup/RmanBackup/*
```

- 7. Perform clean installations:
 - a. Install the Linux CentOS 5.9 Rev8 Operating System.
 - b. Install EMS 7.2.1000 (the same version that you installed in Step 3 above) on both servers. After reboot, make sure all processes are up in EMS Server Manager / Status (see Chapter 13).
- 8. Copy all backup files to the Primary server and run the EMS Server Manager Restore procedure (see Section 16.4).
- **9.** After reboot make sure all processes are up in the EmsServerManager Status (see Chapter 13). Make sure all data from latest backup was properly restored.
- **10.** Install HA on the Primary server and wait for all processes to be up again in EMS Server Manager Status (see Chapter 13). This step can take several minutes.
- **11.** Install HA on the Secondary server and wait for sync to complete.

21.6 EMS Server Restore

EMS server restore cannot be performed while HA is configured.

To restore the EMS server, HA must be uninstalled prior to the restore.

It is recommended to firstly uninstall the secondary server, and only then the primary server. After restoring the server, HA should be reconfigured.

To uninstall HA, see Section 22.6 on page 192.
22 EMS HA Configuration

This section describes the EMS HA Installation.

To configure the primary server:

 In the EMS Server Manager root menu, choose Application Maintenance, in the sub-menu, choose High Availability, and then press Enter; the following is displayed:

Figure 22-1: EMS Server Manager - HA Configuration
--



This menu includes the following options:

- Primary Server Installation in Global IP Model (see Section 22.1 on page 182).
- Primary Server Installation in Geo HA model (see Section 22.2 on page 184).
- Secondary Server Installation (see Section 22.3 on page 184).
- HA Status (see Section 22.4 on page 188).

22.1 Primary Server HA Installation in Global IP Model

This section describes how to install the HA application on the designated Primary server in the Global IP address model.



Note: When alarms are forwarded from the EMS, you can configure the global IP address as a source address. For more information, refer to the *EMS User's Manual.*

To install the HA primary server in Global IP Model:

- **1.** In the High Availability menu, choose option **1** to run the Primary server HA installation, and then press Enter.
- 2. After the HA packages are installed, you are prompted for the HA model:

Figure 22-2: Primary HA Server Menu



For the Global IP HA model, both EMS servers are located in the same subnet.

- 3. In the High Availability sub-menu, choose option 1 (Configure Global IP HA).
- 4. You are now prompted for the following network parameters:
 - 'Global IP' for each configured interface (physical or logical IF).
 - Secondary server's Host name and IP address.
 - Ping Nodes If you have several interfaces configured, you can add another 'ping node' (for more information, see Section 22.2.1 on page 186).

Figure 22-3: Primary HA Server Sub-menu



The current configuration is displayed for confirmation:

Figure 22-4: HA Configuration Display

```
HA Configuration:

Global IP(eth0): 10.7.14.218

Primary Server IP: 10.7.14.141

Primary Server Host: EMS-Linux141

Secondary Server IP: 10.7.14.142

Secondary Server Host: EMS-Linux142

Ping IP: 10.7.0.1

Are you sure that you want to continue ? (y/n/q)
```

- Type **y** to continue the installation process
- Type **n** to reconfigure all parameters
- Type **q** to stop the installation process

The installation process starts (this process may take a few minutes). During the installation, you may encounter one or more of the following system responses:

- "/data: device is busy" When the /data partition is currently in use by another prompt or application. You must un-mount the /data partition before continuing. In the case where the /data partition isn't busy, the above message is not displayed.
- When prompted, press Enter to continue.
- When prompted "To abort waiting type 'yes' [1]:" you can wait or press 'yes' to continue.

When the installation process for the Primary server has completed, the following message is displayed:

Figure 22-5: HA Server Configured as Primary Server - Confirmation



Note: After the installation process has completed, it takes several minutes until the HA status changes to "Online" and the EMS server status changes to 'EMS server is running'.

22.2 Primary Server HA Installation in Geo HA Model

This section describes how to install the HA application on the designated Primary server in the Geo HA model.

To install the HA primary server in Geo HA model:

- **1.** In the High Availability menu, choose option **1** to run the Primary server HA installation, and then press Enter.
- 2. After the HA packages are installed, you are prompted for the HA model:

Figure 22-6: Primary HA Server Menu



For the Geo HA model, EMS servers are located in different subnets.

- In the High Availability sub-menu, choose option 2 (Configure Geo-Redundancy HA).
- 4. You are now prompted for the following network parameters:
 - Secondary server's Host name and IP address.
 - Ping Nodes If you have several interfaces configured, you can add another 'ping node' (for more information, see Section 22.2.1 on page 186).

Figure 22-7: Primary HA Server Sub-menu



The current configuration is displayed for confirmation:

Figure 22-8: HA Configuration Display

HA Configuration:
Primary Server IP: 10.3.180.2
Primary Server Host: EMS-Linux2
Secondary Server IP: 10.17.1.200
Secondary Server Host: vEMS-GeoHA-200
Ping IP: 10.3.180.80
Are you sure that you want to continue ? $(y/n/q)y$

- Type **y** to continue the installation process.
- Type **n** to reconfigure all parameters
- Type **q** to stop the installation process

The installation process starts (this process may take a few minutes). During the installation, you may encounter one or more of the following system responses:

- "/data: device is busy" When the /data partition is currently in use by another prompt or application. You must un-mount the /data partition before continuing. In the event where the /data partition isn't busy, the above message is not displayed.
- When prompted, press Enter to continue.
- When prompted "To abort waiting type 'yes' [1]:" you can wait or press 'yes' to continue.

When the installation process for the Primary server has completed, the following message is displayed:

Figure 22-9: HA Server Configured as Primary Server - Confirmation



Note: After the installation process has completed, it takes several minutes until the HA status changes to 'Online' and the EMS server status changes to 'EMS server is running'.

22.2.1 Ping Nodes

The purpose of these nodes (IP address) is to ensure network connection along all EMS server configured interfaces. When an IP address is configured as "ping node", this implies that the HA process sends ICMP packets (at a constant interval) to this address (through the appropriate Server Ethernet interface). If no response is returned from this ping node (during a constant period of time), the HA process determines that the specific network interface connection is down and acts accordingly (i.e. initiates a possible switchover). The ping node should be a reliable host in the network, such as router or any other machine which accurately reflects the network status.

It is possible to configure several "ping nodes", where each ping node is considered to be a single point of failure. Consequently if there is no connection to one of the ping nodes, a switchover is performed (unless the Secondary server cannot takeover due to the same or different network problems or during initial synchronization between the Primary and Secondary server).



Note: It's recommended to configure a separate ping node for each configured physical Ethernet interface (to the router connected to each of the subnets); however, if Ethernet Redundancy is configured between these two interfaces, then it's sufficient to configure a single ping node.

22.3 Secondary Server HA Installation

This section describes how to install the High Availability (HA) application on the designated Secondary server.

To install the secondary server:

1. In the High Availability menu, choose option **2** to run the Secondary HA Server installation, and then press Enter.



Note: The Secondary server configuration MUST be performed after the Primary server configuration has completed and its status is 'EMS Server is running'.

2. After the HA packages are installed, you are prompted for the 'Primary IP' and *acems* user password (you might also be prompted to answer **yes** before connecting).

Figure 22-10: Primary HA Server IP



The Secondary server copies the HA configuration files from the Primary server and then starts the installation process.

Figure 22-11: Secondary HA Server Configuration

Start Heartbeat Configuration	
Primary Server IP:[-1]:	10.7.14.143
acems@10.7.14.143's password:	
drbd.conf	
ha.cf	
cib.xml	
haresources	
Copy files from primary server:	[OK]
Update primary parameters in se	condary:
Global IP(ethO):	10.7.14.215
Global IP(eth1):	10.77.10.215
Primary IP:	10.7.14.143
Primary Host:	EMS-Linux143
Secondary IP:	10.7.14.144
Secondary Host:	EMS-Linux144
Ping IP:	10.7.0.1,10.77.10.1
Press any key to continue	

- 3. When prompted '[need to type **yes** to confirm]' press **yes**.
- 4. When prompted 'Press any key to continue...' press Enter.

22.4 HA Status

The 'HA status' displays both servers' High Availability parameters.

To verify the EMS HA status:

In the High Availability menu, choose option 3 (HA Status), and then press Enter; the following status view is displayed (Example only):

```
Figure 22-12: EMS HA Status - Example Display
```

- **HA Heartbeat Service Status:** Whether the heartbeat service is installed and running.
- **HA DRBD Service Status:** Whether the data replication service is installed and running.
- **<HOST_NAME > HA Status:** The following states are available:
 - ONLINE HA is enabled and heartbeat packets have been sent.
 - OFFLINE HA is disabled or does not exist (this state usually appears for several minutes after the new installation).
 - IN Progress HA has started (this state usually appears for several seconds immediately after the new installation).
- **<HOST_NAME > HA Location Status:** the following states are available:
 - Unknown Cannot resolve if the EMS server is Primary or Secondary
 - Primary The current working server
 - Secondary the redundant server
- **<HOST_NAME > HA Data Sync Status:** the following states are available:
 - DUnknown Cannot resolve whether the EMS server data is synchronized with the other server
 - UpToDate The replicated data is synchronized with the Primary server
 - Inconsistent The replicated data is in the progress of synchronizing with the Primary server
- Network Connection (<Ping Node>):- For each configured ping node, this status verifies if there is a network connection to it.
- **HA EMS Status:** The current state of the EMS server and watchdog processes:
 - The EMS server is running the EMS server process is up.
 - The EMS is not installed
 - The EMS server is not running the EMS watchdog is trying to start the EMS server.
 - The EMS watchdog is not running.
 - Unknown, Not Primary Server This state is always displayed on the Secondary server. In addition, it displays when HA is not configured.

22.4.1 Advanced Status View

This section describes the advanced status view.

To view the advanced status:

In the High Availability Status screen, press **a**; the following is displayed:

Figure 22-13: Advanced Status View

Heartbeat Advanced Status
heartbeat OK [pid 21524 et al] is running on ems-linux6 [ems-linux6]
<pre></pre>
Node: ems-linux6 (69778371-0a03-b406-faaf-657669826990): online Node: ems-linux2 (69778371-0a03-b402-faaf-657669826990): online
Resource Group: group_1 drbddisk_1 (heartbeat:drbddisk): Started ems-linux2 Filesystem_2 (ocf::heartbeat:Filesystem): Started ems-linux2 IPAdder-resource (ocf::heartbeat:IPaddr2): Started ems-linux2 resource-EMS-Server (lsb:EMSServer): Started ems-linux2
DRBD Advanced Status
<pre>drbd driver loaded OK; device status: version: 8.2.4 (api:88/proto:86-88) GIT-hash: fc00c6e00alb6039bfcebe37afa3e7e28dbd92fa build by root@EMS-Linux143, 2011-01-26 12:04:18 0: cs:SyncTarget st:Secondary/Primary ds:Inconsistent/UpToDate C r ns:0 nr:2942588 dw:2941852 dr:0 al:0 bm:179 lo:24 pe:1372 ua:23 ap:0 [>] sync'ed: 4.4% (63685/66557)M finish: 0:16:58 speed: 63,804 (56,556) K/sec resync: used:4/31 hits:185355 misses:196 starving:0 dirty:0 changed:196 act_log: used:0/257 hits:0 misses:0 starving:0 dirty:0 changed:0</pre>
Press "s" - status view, "a" - advanced status view or any other key to continue

The advanced status view provides a more detailed view of the EMS HA status. This command is particularly important during the initial synchronization between the primary and secondary EMS servers when the precise percentage of the stage of the EMS HA synchronization process is displayed (highlighted in green in the above figure).

22.5 EMS Server Manual Switchover

Manual switchover can be performed from either the Primary HA or Secondary HA server.

To manually switchover to the active EMS server:

1. In the High Availability menu, choose option 2 (HA Switchover), and then press Enter.

EMS Server 7.2.126 Management	
Main Menu> Application Maintenance> High Availability	
>1. <u>HA Status</u> 2.HA Switchover 3.Uninstall HS b.Back g.Quit to main Menu	

2. Type **y** to confirm your selection.

During the manual switchover process, the "switchover in process..." message is displayed in the EMS server machine where the command was activated. If you run the 'HA Status' command on the other server, it will display the HA status of the Primary server as STANDBY until the Secondary server becomes the Primary server.





After the Secondary server becomes the Primary server, a few minutes are required until the EMS application is up and running.

Figure 22-16: Status after Switchover

High Availability Status	3
	-
HA Heartbeat Service Status HA DRBD Service Status	[UP] [UP]
EMS-Linux6 HA Status EMS-Linux6 HA Location Status EMS-Linux6 Data Sync Status	[ONLINE] [Primary] [UpToDate]
EMS-Linux2 HA Status EMS-Linux2 HA Location Status EMS-Linux2 Data Sync Status	[ONLINE] [Secondary] [UpToDate]
Network Connection(10.3.180.80)	[OK]
HA EMS Status	[EMS Server is running!!]

22.6 EMS HA Uninstall

The user should uninstall the EMS HA application on both the Primary and Secondary servers under the following circumstances:

- EMS software version upgrade
- EMS server network configuration changes
- Custom certificate installations

To uninstall EMS HA:

In the High Availability menu, choose option 3 (Uninstall HA), and then press Enter. The uninstall process takes 1-2 minutes with the following output:

Figure 22-17	7· Uninstall FMS H	A Status Display
I Iguic ZZ-I		A Olalus Display





Note: The EMS application doesn't start automatically after this process has completed. To start the EMS, reboot the EMS server or quit the EMS Server Manager and run it again using the 'Start EMS Server' option (see 16.1 on page 115).



This page is intentionally left blank.



Configuring the Firewall and Installing the EMS Client

This part describes how to configure the EMS firewall and install the EMS client.



23 Configuring the Firewall

The EMS interoperates with firewalls, protecting against unauthorized access by crackers and hackers, thereby securing regular communications. You need to define firewall rules to secure communications for the OVOC client-server processes. Each of these processes use different communication ports. By default, all ports are open on the EMS server side. When installing the EMS server, you need to configure its network and open the ports in your Enterprise LAN according to your site requirements; based on the firewall configuration rules (representing these port connections) that are described in the table and figure below.

Connection	Port Type	Secured Connection	Port Number	Purpose	Port side / Flow Direction			
EMS Clients and EMS Server								
EMS Client PC ↔ EMS Server	TCP	✓	22	SSH communication between EMS server and client PC. Initiator: client PC	EMS server side / Bi- directional.			
	ТСР	×	80	HTTP for JAWS. Initiator: client PC				
JAWS and NBIF Clients ↔ EMS Server	ТСР	✓	443	HTTPS for PC client/ JAWS and NBIF. Initiator: Client	EMS server side / Bi- directional.			
EMS Server ↔ Single Sign-on Connection to Device Web	TCP (HTTP)	×	8090	Direct HTTP connection between the device's embedded Web interface (Management PC) and the EMS server. Initiator: EMS Server	EMS server side / Bi- directional			
	TCP (HTTPS)	✓	8091	Direct HTTPS connection between the device's embedded Web interface (Management PC) and the EMS server. Initiator: EMS Server	EMS server side / Bi- directional.			

Table 23-1: Firewall Configuration Rules



Connection	Port	Secured	Port	Purpose	Port side /		
	туре	Connection	Number		Flow Direction		
EMS Server and Devices							
Device ↔ EMS Server (SNMP)	UDP	*	1161	Keep-alive - SNMP trap listening port (used predominantly for devices located behind a NAT). Initiator: EMS server	EMS server side / Bi-directional		
	UDP	✓	162	SNMP trap listening port on the EMS. Initiator: AudioCodes device	EMS server side / Receive only.		
	UDP	✓	161	SNMP Trap Manager port on the device that is used to send traps to the EMS. Initiator: EMS server	MG side / Bi-directional		
Device↔ EMS Server (NTP Server)	UDP (NTP server)	×	123	NTP server synchronization. Initiator: MG (and EMS Server, if configured as NTP client) Initiator: Both sides	Both sides / Bi-directional		
Device ↔ EMS Server (REST communication /upload/download)	TCP (HTTP)	×	80	HTTP connection for files transfer. Initiator: EMS server	EMS server side / Bi- directional		
	TCP (HTTPS)	✓	443	HTTPS connection for files transfer (upload and download) and REST communication. Initiator: EMS server	EMS server side / Bi- directional		

Connection	Port Type	Secured Connection	Port Number	Purpose	Port side / Flow Direction			
Endpoints (IP Phones)								
EMS Server ↔ IP Phone Management Server	TCP (HTTP)	×	80	HTTP connection between the EMS server and the IP Phone Management Server Web browser. Initiator: Client browser	EMS server side / Bi- Directional.			
	TCP (HTTPS)	×		HTTPS connection between the EMS server and the IP Phone Management Server Web browser. Initiator: Client browser	EMS server			
			443	HTTPS connection used by endpoints for downloading firmware and configuration files from the EMS server. Initiator: Endpoints	side / Bi- Directional.			
EMS Server ↔ Endpoints (IP Phones)	HTTP	×	8080	HTTP connection that is used by endpoints for downloading firmware and configuration files from the EMS server. Initiator: Endpoint	EMS server side / Bi- directional			



Connection	Port Type	Secured Connection	Port Number	Purpose	Port side /	
	- 71				Flow Direction	
	TCP (HTTP)	×	8081	HTTP connection used for sending REST updates from the endpoints to the EMS server, such as alarms and statuses. Initiator: Endpoint	EMS server side / Bi- directional	
	TCP (HTTPS)		8082	HTTPS connection used for sending secure REST updates from the IP phone to the EMS server, such as alarms and statuses (SSL authentication without certificate).	EMS server side / Bi- directional	
SEM Server and Dev	vices					
Media Gateways ↔ SEM server	ТСР	×	5000	XML based SEM communication. Initiator: Media Gateway	EMS server side / Bi- directional	
	TCP (TLS)	✓	5001	XML based SEM TLS secured communication. Initiator: Media Gateway	EMS server side / Bi- directional	
SEM Client						
SEM client ↔ Tomcat server	TCP (HTTP)	×	8400	SEM HTTP connection between the user's browser and Tomcat server. Initiator: Client's PC.	EMS server side / Bi- directional	

Connection	Port Type	Secured Connection	Port Number	Purpose	Port side / Flow Direction
	TCP (HTTPS)	✓	9400	SEM HTTPS connection between the user's browser and Tomcat server. Initiator: Client.	EMS server side / Bi- directional
MS-SQL Server	·	·			
SEM server ↔ Lync MS-SQL Server	ТСР	×	1433	Connection between the EMS server and the MS-SQL Lync server. This port should be configured with SSL. Initiator: EMS server	Lync SQL server side / Bi-directional
LDAP Active Directo	ory Server	1	1		
SEM server ↔ Active Directory LDAP server (Microsoft Lync user	TCP	×	389	Connection between the SEM server and the Active Directory LDAP server. Initiator: EMS server	Active Directory server side/ Bi-directional
with SEM)	TCP (TLS)	×	636	Connection between the SEM server and the Active Directory LDAP server with SSL configured. Initiator: EMS server	Active Directory server side/ Bi-directional
EMS server ↔ Active Directory LDAP Server (EMS user authentication)TCP×389		Connection between the EMS server and the Active Directory LDAP server (EMS Users). Initiator: EMS server	Active Directory server side/ Bi-directional		



Connection	Port Type	Secured Connection	Port Number	Purpose	Port side / Flow Direction				
	TCP (TLS)	*	636	Connection between the EMS server and the Active Directory LDAP server (EMS Users) with SSL configured. Initiator: EMS server	Active Directory server side/ Bi-directional				
RADIUS Server									
EMS server ↔ RADIUS server	ТСР	×	1812	Direct connection between the EMS server and the RADIUS server (when EMS user is authenticated using RADIUS server). Initiator: EMS server	EMS server side / Bi- directional				
EMS HA		·		·					
Primary EMS Server ↔ Secondary EMS Server (HA Setup)	ТСР	×	7788	Database replication between the servers. Initiator: Both servers	Both EMS servers / Bi-directional				
Server (nA Setup)	UDP	×	694	Heartbeat packets between the servers. Initiator: Both servers					
Mail and Syslog Ser	vers	1		·					
EMS server ↔ Mail Server	ТСР	×	25	Trap Forwarding to Mail server Initiator: EMS server	Mail server side / Bi- directional				
EMS server ↔ Syslog Server	ТСР	×	514	Trap Forwarding to Syslog server. Initiator: EMS server	Syslog server side /Bi- directional				

202

Connection	Port Type	Secured Connection	Port Number	Purpose	Port side / Flow Direction
RFC 6035					
SEM Server ↔ Endpoints	UDP	×	5060	SIP Publish reports sent to the SEM server from the endpoints, including RFC 6035 SIP PUBLISH for reporting device voice quality metrics. Initiator: Endpoint	SEM server / Bi-directional



Figure 23-1: Firewall Configuration Schema



Note: The above figure displays images of devices. For the full list of supported products, see Chapter 2 on page 21.

■ NOC ↔ EMS (Server) ports

Source IP Address Range	Destination IP Address Range	Protocol	Source Port Range	Destination Port Range
NOC/OSS		SFTP	1024 - 65535	20
	Device/SBA/IP Phone/EMS	FTP	1024 - 65535	21
		SSH	1024 - 65535	22
		Telnet	1024 - 65535	23
		NTP	123	123
		-	-	-
		HTTP/HTTPS	N/A	80,443

Table 23-2: OAM Flows: NOC ↔ Device/ IP Phone/ SBA/ EMS

Table 23-3: OAM Flows: Device/ IP Phone/ SBA/ EMS↔ NOC

Source IP Address Range	Destination IP Address Range	Protocol	Source Port Range	Destination Port Range
Device/IP Phone/ SBA/EMS		NTP	123	123
	NOC/OSS	SNMP Trap	1024 – 65535	162
		-	-	-



This page is intentionally left blank.

24 Installing the EMS Client

This section describes how to install the EMS Client on a PC or Laptop.



Note: Before you run the EMS Client exe file, ensure that you extract the entire Disk1 EMS client directory to your PC/laptop in the same relative path as the Disk image, and only then, run the exe file from this location.

To install the EMS client on a PC or Laptop:

- 1. Insert AudioCodes' EMS installation disk into the CDROM.
- 2. Open the EmsClientInstall\Disk1\InstData\VM directory.
- 3. Do one of the following:
 - On Windows 7:
 - a. Right-click the EMS client Installation file ac_ems_setup_win.exe, and then choose **Run as administrator**; the EMS client installation setup is displayed.
 - **b.** Follow the prompts to install the EMS client.

Figure 24-1: EMS Client Installation-Run as Administrator

EmsClientInstall Disk1	 Ins 	tData 🕨 VM		✓ 49 Search V
Burn New folder				
Name		Date modified	Туре	Size
ems_setup_win.exe		03-Jun-13 3:08 PM Open	Application	42 589 KB
	<u></u>	Run as administrator Troubleshoot compat	ibility	
)	Edit with Notepad++ Select Left Side to Cor		
		Scan for Viruses		

Upon the completion of the installation process, the EMS client icon is added to the desktop.



• On Windows 8:

a. Right-click the installation exe file, and then choose **Properties**; the Properties window is displayed:

Figure 24-2: EMS Client Installation File-	Windows 8 Properties
--	----------------------

ame		Date modified	Туре	Size	
ac_ems_setup_win.exe	-	Open Pin to Start Run as administrator Troubleshoot compatibility Always available offline Restore previous versions Send to Cut Conv	Application	42,474 KB	
		Paste Create shortcut Delete Rename Properties			

b. Select the **Compatibility** tab, and then select the checkbox **Run this program in compatibility mode for**.

eneral Comp	atibility Security	Details Previo	us Versions
If this program i try running the	isn't working corre compatibility trout	ectly on this versio bleshooter.	n of Windows,
Run compa	tibility troubleshoo	oter	
How do I choo	se compatibility sr	ettings manually?	
Compatibility	mode		
Run this	program in compa	tibility mode for:	
Aufordown 7			
Windows 95 Windows 98 Windows XF Windows XF Windows Vi Windows Vi Windows 7	I / Windows Me ⁹ (Service Pack 2 ⁹ (Service Pack 3 sta sta (Service Pack ¹⁰ (Service Pack)	2) 8) 1] 2]	
	<u>s s. 15</u> v	S 8	
Privilege leve	4		
- Hun this p	program as an adr	ministrator	
Change	settings for all us	ers	
a and the			
Curran 30			

Figure 24-3: EMS Client Installation File-Compatibility Tab

- c. In the Windows 7 pane, select Windows 7.
- d. Click OK.
- e. Right-click the EMS client installation file ac_ems_setup_win.exe, and then choose Run as administrator; the EMS client installation setup is displayed.
- f. Follow the prompts to install the EMS client.

Upon the completion of the installation process, the EMS client icon is added to the desktop.



Note: If you have replaced the "AudioCodes-issued" certificates with external CA certificates, and wish to uninstall the previous EMS client, ensure that you backup the **clientNssDb** files: **cert8.db**, **key3.db**, and **secmod.db**.

24.1 Running the EMS Client on a PC or Laptop

This section describes how to run the EMS client on a PC or Laptop

To run the EMS on Windows XP or older:

Double-click the EMS client icon on your desktop or run Start > Programs > EMS Client.

To run the EMS on Windows 7 or later:

 Right-click the EMS client icon on your desktop, and then choose Run as Administrator.

Figure 24-4: Running EMS Client-Run as Administrator



24.2 Initial Login

This section describes how to initially login to the EMS client.

To initially login to the EMS client:

1. Log in as user 'acladmin' with password 'pass_1234' or 'pass_12345'.



Note: First-time access defaults are case sensitive. After you login to the EMS for the first-time, you are prompted to change the default password. If you incorrectly define these or the field Server IP Address, a prompt is displayed indicating that the fields should be redefined correctly.

2. In the main screen, open the 'Users List' and add new users according to your requirements.

24.3 Installing and Running the EMS Client on a PC using Java Web Start (JAWS)

Java Web Start (JAWS) enables you to install the EMS client (compatible with your EMS server version) without using any CDs.

To install the EMS client on a PC using JAWS:

1. Open a browser and type the EMS server IP in the Address field and add /jaws as suffix, for example:

http://10.7.6.18/jaws/

- **2.** Follow the online instructions.
- To run the EMS client after JAWS install through URL:
- Specify the path http://<server_ip>/jaws.
 An 'EMS Login Screen' is opened.
 For example: http://10.7.6.18/jaws/



This page is intentionally left blank.

Part VIII

Appendix

This part describes additional EMS server procedures.



A Site Preparation

This appendix describes the procedures for backing up the EMS server.



Note: It is highly recommended to perform a complete backup of the EMS server prior to performing an installation or upgrade, according to the procedures described below.

- EMS server data backup should be performed prior to machine formatting. The Backup files should be transferred to another machine prior to the EMS server installation. Note, that these backup files cannot be used for other versions. They should be kept in case the user fails to install the new version, and decides to roll back to the previous version.
- EMS Users: all the users' names and permissions should be saved. After the new EMS version is installed, these users should be defined manually with default passwords. To perform this task, in the EMS menu, choose Security > User's List menu.
- EMS Tree: the user can export the gateways tree using the File > MGs Report command (example of the file is attached). This file is a CSV file and does not preserve secured information such as passwords. Therefore, we recommend extending it manually with columns including: SNMP read and write community strings, or SNMPv3 user details. This information will be required during the device's definition in the newly installed EMS system. It's also highly recommended to perform gateway removal and adding and to ensure that the EMS <-> GW connection has been established.

	В	C	D	E	F	G	Н	1	J	K	L	
1	IP Address	Node Name	RegionName	Description	Product Type	Software	Connectio	Administra	Operative	Mismatch	Last Ch	ŀ
2	10.7.19.88	10.7.19.88	gena		MEDIANT 8000	5.8.57	Connected	Unlocked	Enabled	No Misma	2009-1	
3	10.7.5.220	10.7.5.220	Roye		UNKNOWN MP114 FXS/FXO	5.90A.006	Connected			No Misma	2009-1	
4	10.7.5.221	10.7.5.221	Roye		UNKNOWN	5.50.020	Connected			No Misma	2009-1	
5	10.7.5.217	10.7.5.217	Roye		MP112	5.80A.020	Not Connected			No Misma	2009-1	1
6	10.7.5.214	10.7.5.214	Roye		UNKNOWN	unknown_	Not Conne	cted		No Misma	2009-1	
7	10.7.5.211	10.7.5.211	Roye		UNKNOWN	unknown_	Not Conne	cted		No Misma	2009-1	
8	10.7.5.222	10.7.5.222	Roye		UNKNOWN	unknown_	Not Conne	cted		No Misma	2009-1	
9	10.7.5.215	10.7.5.215	Roye		UNKNOWN	unknown_	Not Conne	cted		No Misma	2009-1	1

Figure A-1: Save MGs Tree Command



This page is intentionally left blank.
B EMS Application Acceptance Tests

This appendix describes the EMS Application Acceptance tests.

B.1 Introduction

The following series of tests are defined as acceptance tests for the EMS application and cover all the major areas and features of the application.

The tests should run sequentially as a single test with dependencies. For example, you can't add a media gateway to the EMS before you have added a software file.

It is also recommended to integrate the below test plan in the Acceptance Test Plan (ATP) of the complete solution of which the EMS is a component. The ATP is typically developed by the solution integrator and covers all solution components (e.g. Softswitch, Media Gateway, IP routers etc). The ATP typically verifies "end to end" functionality, for example, the calls running through the solution. The below test plan should be integrated in the ATP as part of this "end to end" functionality testing (e.g. you may send and receive calls through the media gateway, perform media gateway board switchover and verify that calls are recovered on the redundant board).

Prior to running the tests described below, the tester should have a basic understanding of how to operate the product. Next to each test case there is a reference to the relevant chapter in the documentation. The tester should read these chapters to acquire the required tools to run this test. Running this test can also be considered as an excellent hand's-on initial training session.

B.2 Configuration

This section describes the EMS application configuration acceptance tests.

B.2.1 Client Installation

Table B-1:	Acceptance	Test –	Client	Installation
------------	------------	--------	--------	--------------

Step Name	Description	Expected Result
Install	Install the client software	Verify that all the instructions are clear.

B.2.2 Server Installation

Table B-2: Acceptance Test – Server Installation

Step Name	Description	Expected Result		
Server	Run the full procedure that installs the DB software, creates the DB, creates the schema and installs the EMS server.	The EMS server directory exists under /ACEMS.		
Reboot	Reboot the EMS server	The EMS server starts automatically.		
Connect	Connect to the EMS server with the EMS client	The connection should succeed.		

B.2.3 Add Auxiliary File

Table B-3: Acceptance Test – Add Auxiliary File

Step Name	Description	Expected Result			
Software Manager	Open the Software Manager Tools >> SW manager	The Software Manager window opens.			
Auxiliary Tab	Choose the auxiliary tab	A new tab is opened with all the available auxiliary files.			
Add Auxiliary File	Choose an auxiliary file that you usually work with such as: Call Progress Tone	A new file was added to the SW Manager.			
Add file browser	Click the Add file Button (Plus sign)	Software File added to the Software Manager.			

B.2.4 Add Media Gateway

Table B-4: Acceptance Test – Add MG

Step Name	Description	Expected Result		
Add MG	Add MG to the EMS	The media gateway appears in the EMS GUI.		
MG Status	Click on the Media Gateway	The Media Gateway status is available in the GUI, including all LEDS and boards.		

B.2.5 Entity Profile – Digital CPE Devices

Step Name	Description	Expected Result		
Go to trunk level	Drill down to trunk level	Trunks list appears according to board type.		
Trunk Properties	Open trunk#1 properties	The frame provisioning opens and all the parameters are available.		
Trunk Configuration	Configure the trunk	The new set of values appears on the provisioning screen.		
Apply	Apply the new configuration	Action successful and there were no errors and no purple tabs.		
Save profile	Save the profile, choose an appropriate name.	The new profile appears in the profiles list. Name Profiles Save Profile Apply to all Parameters Image: Trunk Profile MyTrunk Image: Save Apply to all Show		
Apply to All	Download this configuration easily to all trunks by using the apply to all	Open trunk#2 and verify the configuration is equal to trunk#1.		

Table B-5: Acceptance Test – Entity Profile: Digital CPE Devices



Note: Digital CPE devices include the following products: Mediant 500 MSBR, Mediant 500L MSBR, Mediant 600, Mediant 800B MSBR, Mediant 800B Gateway and E-SBC, Mediant 1000B MSBR, Mediant 1000B Gateway and E-SBC, Mediant 1000, Mediant 2000 and Mediant 3000.

B.2.6 Entity Profile – Analog CPE Devices

Step Name	Description	Expected Result		
Go to telephony frame	Click on the telephony button	Telephony configuration is displayed.		
Save profile	Save the profile, choose an appropriate name Name Profiles Save Profile Parameters MP Telephory Profile - Choose Profile	The new profile is displayed in the profiles list. Name Profiles Save Profile Parameters MP Telephony Profile MyTelephony Save Show		
Expose profile parameters	Press on the "show profile parameters" button	All profiles parameters are marked with the profile name. Polarity Reversal Type Fit123Hz Fit2 Metering Type Fit123Hz Fit2 Mn Flach Hook Time Soo Caller ID Timing Mode Disabled Fit2		
Detach profile	Change one of the profile parameters, and then press Apply .	A detach profile pop up message is displayed.		

Table B-6: Acceptance Test – Analog CPE Devices



Note: Analog CPE devices include the following products: MediaPack, Mediant 600, Mediant 500 MSBR, Mediant 500L MSBR, Mediant 800B MSBR, Mediant 800B Gateway and E-SBC, Mediant 1000B MSBR; Mediant 1000B Gateway and E-SBC and Mediant 1000.

B.3 Faults

B.3.1 Alarm Receiver

Figure B-1: Alarm Receiver

Alarm	Browser 208	View Level:	Node Level Alarms		_ 🕑 🐔 🗷 🗆 🖸 🗹 🖬 🖬 🔽 🔽 🗙
Ack	Severity Time	MG Name Sou	ce Alarm Na	ime	Description

Table B-7: Acceptance Test – Alarm Receiver

Step Name	Description	Expected Result		
Raise Alarm	Lock one of the elements in the MG, such as the trunk.	The alarm is received in the EMS.		
Clear Alarm	Unlock one of the elements in the media gateway, such as a trunk.	The clear alarm is received in the EMS.		

B.3.2 Delete Alarms

Table B-8: Acceptance Test – Delete Alarms

Step Name	Description	Expected Result
Delete Alarms	Right-click the alarms in the alarm browser and delete all the alarms	The alarm browser in empty.

B.3.3 Acknowledge Alarm

Table B-9: Acceptance Test – Acknowledge Alarm

Step Name	Description	Expected Result
Check Box	Click on the Acknowledge check box	The alarm is marked as acknowledge.

B.3.4 Forwarding Alarms

Destination Type	Email						Mail Host	40.7.5.22	
	Email							10.7.5.33	
Destination Rule Name	Johnyema	i					Mail Host Username	test	
Allow Forward	Prever	nt Forwa	rd				Mail Host Password	******	
Alarm Origin	A	EMS	SEM	MGW	IP Phon	e	From (no spaces allowed)	audiocodes_ems	
	Events	V	V	V	V		To (comma separated):		
Alarm Names	All Alarms					5	Johnsmith@gmail.com		
	 					6			
Alarm Types	communic	ationsAla	irm			8 .,			
Soverities									
Sevenues	E E								
Source									
Source MGW List	Select M	GW							
	Region	<u> </u>	MGW Name	IP A	ddress	_			
	Paris		10.33.70.8	10.3	3.70.8				
	Paris		10.15.7.95	10.1	5.7.95				

Figure B-2: Destination Rule Configuration

Table B-10: Acceptance Test – Forwarding Alarms

Step Name	Description	Expected Result
IP	Enable the Alarm Forwarding feature Tools >> trap configuration Add rule	Verify that you receive the Traps in the requested IP address on port 162.
Port	Change the Port number	Verify that you receive the Traps in the requested IP address on the new port.

B.4 Security

This section describes the EMS application security tests.

B.4.1 Users List

😵 Users List				
File View Ac	tions Help			
ሴ 🟠 👹 🕯	🗅 🌆 💧			
Basic Info	Regions Info			
User Name	Security Level	Login Type Full Nar	me Status	Valid IP
vladi	Administration	User/Password Lo	NOT ACTIVE	4
yaniv	Administration	User/Password Lo	ACTIVE	
shirlyg	Administration	User/Password Lo	NOT ACTIVE	
gena	Administration	User/Password Lo	NOT ACTIVE	
mon	Monitoring	User/Password Lo	NOT ACTIVE	
Brad	Administration	User/Password Lo	ACTIVE	
nachum	Administration	User/Password Lo	NOT ACTIVE	
eran	Administration	User/Password Lo	NOT ACTIVE	
yanive	Administration	User/Password Lo	NOT ACTIVE	
acladmin	Administrator Super User	User/Password Lo	NOT ACTIVE	
lilach	Administration	User/Password Lo	NOT ACTIVE	
Bahir	Administration	User/Password Lo	NOT ACTIVE	
•				7 F
			ок	Close

Figure B-3: Users List

 Table B-11: Acceptance Test – Add an Operator

Step Name	Description	Expected Result
Add	Add a new operator, and then press the OK key in the screen.	Verify the new operator was added to the operators table frame.

B.4.2 Non Repetitive Passwords

Table B-12: Acceptance Test – Non Repetitive Passwords

Step Name	Description	Expected Result
Change password	Change password and try to enter the old password.	The old password is not valid. The password has been used before, please choose another one."

B.4.3 Removing Operator

Step Name	Description	Expected Result
Remove	Remove a user from the operators table by selecting the remove button in the operators table.	A pop up window prompts you whether you wish to remove the user.
Verify	Select the OK button.	Verify that the user you selected was removed from the operators table.

B.4.4 Journal Activity

Figure B-4: Actions Journal

Actions	s Journal									X
File View	Help Nournal Entres I 0 Alarr	ns Entries out of 9	552		_		Advanced Eiter: 🕅	lournak 🐼	Abarman 70	2
Entres. 1150	o doumai Entres o Alam	From:	08-Dec-2009	E 1434 C	3	To: 15-Dec-200	Advanced Filler.			
Severity	Time	MG Name	Source	ActionAlarm	Name	Details		Region	Operator	
Journal	16:35:13 Dec 15 2009	10.77.10.130		Configuration	Update	Action UnLo	ck was performed	Mor	mor	
Journal	16:35:07 Dec 15 2009	10.77.10.130		Configuration	Update	Action Lock	was performed	Mor	mor	
Journal	16:35:06 Dec 15 2009	10.77.10.130		Configuration	Update	Update Para	meters: Field-tgMGInfoActi	o Mor	mor	Y

Table B-14: Acceptance Test – Journal Activity

Step Name	Description	Expected Result
Activity	Open the action journal.	Check that all actions that you performed until now are registered.
Filter	Use the filter: time, user and action.	Time, user, action filter are working OK.

B.5 Utilities

This section describes the EMS application utilities acceptance tests.

B.5.1 MG Search

Figure	B-5.	Media	Gateway	Search
Iguie	Б-Ј.	Meula	Galeway	Jearth

Search Media Gateway		×		
 Search By Product Information 	n:			
Product Type	Mediant 5000/Mediant 8000	•		
Software Version	All Versions	•		
Product Status	All	•		
 Search by IP address: Search by serial number: 				
Search by MG Name:				
Match case Match whole word only				
	ОК Са	ncel		

Step Name	Description	Expected Result
Search Box	Open the MG search dialog by choosing Tools >> Search MG in the EMS Main menu.	Search MG tool opens.
IP	Search /MG/Unknown machine by IP address.	Displays a dialog with a list of results according to selected criteria.
Serial Number	Search /MG/Unknown machine by serial number.	Displays a dialog with a list of results according to selected criteria.
MG Name	Search /MG/Unknown machine by MG Name.	Displays a dialog with a list of results according to selected criteria.
Additional Search Options	Search /MG/Unknown machine by matching case or by matching a whole word.	Displays a dialog with a list of results according to selected criteria.

Table B-15: Acceptance Test – MG Search

B.5.2 Online Help

Table B-16: Acceptance Test – Online Help

Step Name	Description	Expected Result
Alarms	Select one alarm and verify that the help opens in the correct context in the online help	Relevant information, clear and user friendly.
Status	Stand on one MG status screen and open the online help	Relevant information, clear and user friendly.
Provisioning	Stand on one tab in the provisioning windows and open the online help	Relevant information, clear and user friendly.

B.5.3 Backup and Recovery

Table B-17: Acceptance Test – Backup and Recovery

Step Name	Description	Expected Result
Backup	Create backup file in the EMS server according to the EMS Installation & Maintenance manual	A backup will be created in the same folder.
Recovery	Perform recovery on the new machine according to the EMS Installation & Maintenance manual	The new server is identical to the previous server.

C Configuring RAID-0 for AudioCodes EMS on HP ProLiant DL360p Gen8 Servers

This appendix describes the required equipment and the steps for configuring the HP ProLiant server to support RAID-0 Disk Array configuration for the EMS server installation.



Note: This procedure erases any residual data on the designated disk drives.

C.1 **Prerequisites**

This procedure requires the following:

- ProLiant DL360p Gen8 server pre-installed in a compatible rack and connected to power.
- Two 1.2TB SAS disk drives
- A VGA display, USB keyboard, and USB mouse must be connected to the server back I/O panel.

C.2 Hardware Preparation

Make sure that two 1.2TB SAS disk drives are installed on slot 1 and 2 of the server. If required, refer to the *HP Service Manual*.

Figure C-1: Hardware Preparation



C.3 Configuring RAID-0

This procedure describes how to configure RAID-0 using the HP Array Configuration Utility (ACU).

> To configure RAID-0:

- 1. Power up the server. If the server is already powered up and running, use the 'reboot' command (from system console as user root) to reboot the server.
- 2. While the server is powering up, monitor the server and wait for the following screen:

Figure C-2: HP Array Configuration Utility (ACU)



- 3. Press <F5> to run the HP Array Configuration Utility (ACU).
- 4. Wait for the ACU to finish loading.

When the ACU is ready, the following screen is displayed:

Figure C-3: RAID-Latest Firmware Versions

Administrator	• ? >
Available Device(s)	
Server	Welcome to HP Smart Storage
HP ProLiant ezsetupsystem2c44fd7c889c	Administrator
Array Controller(s)	The HP Smart Storage Administrator is an application that allows you to configure, diagnose and manage Smart Storage devices attached to your server.
Smart Array P4201 3	To begin, please select a device from the menu on the left
	What's New?
	The HP Smart Storage Administrator replaces the HP Array Configuration Utility for all configuration, diagnostics, and Smart SSD Wear Gauge functions
	Configuration and diagnostics support for the following Smart Array controllers: • HP Smart Array P212, P410, P410, P411, P711m, P712m, P812 • HP Smart Array P420, P222, P220, P420, P421, P721m, P822 • HP Smart Array P430, P431, P731m
	HP SSD Smart Path configuration.

5. In the left-hand pane, select **Smart Array P420i**; an Actions menu is displayed:

Figure C-4: Actions Menu



AudioCodes

6. Click **Configure**, and then click **Clear Configuration** to clear any previous configuration; the following confirmation is displayed:

Figure C-5: Clear Configuration



7. Click **Clear** to confirm; a summary display appears:





8. Click **Finish** to return to the main menu. The following screen is displayed:

Figure C-7: Main Screen

onfigure 📿 Refresh	😝 Unassigned Drives
Devices	
Smart Array P420i Embedded Slot	Group By Enclosure
Arrays	Select All (2)
Unassigned Drives (2)	1.2 TB

- 9. In the left-hand pane, select **Unassigned Drives (2)**; make sure that both the drives are selected, and then click **Create Array**.
- 10. Select RAID 0 for RAID Level.
- 11. Select the 'Custom Size' check box, and then enter 2000 GiB.
- **12.** At the bottom of the screen, click **Create Logical Drive**; the following screen is displayed:



Figure C-8: Logical Drive

After the array is created, a logical drive should be created.

13. Click Create Logical Drive.



A summary screen is displayed:

Figure C-9: Summary Screen

Logical Drive was successfully	created. Please choose one of the actions below.	
Array Details		
Status	OK	
Unused Space	235.5 GiB (10.5%)	
Used Space	2000.0 GiB (89.5%)	
Total Usable Space	2.1 TIB	
Logical Drives		
Logical Drive 1	1.95 TiB (2.15 TB)	
Physical Drives		
1 2 TB SAS HDD at Port 11 : Box 1	Bay 1	
1.2 TB SAS HDD at Port 11 : Box 1	Bay 2	
Device Path		
Smart Array P420i in Embedded S	llot	

14. Click Finish.

Figure C-10: Set Bootable Logical Drive/Volume



The new logical volume needs to be set as a bootable volume.

15. In the left-hand pane, select **Smart Array P420i**, and then click **Set Bootable Logical Drive/Volume**; the following screen is displayed:

	Embedded Slot	> Set Bootable	2 Logical Drive/volume
t	Logical Drive/Volu	me Precedence (what's this	.?)
Logi	al Drive/Volume	Primary Boot Logical Drive/Volume	Secondary Boot Logical Drive/Volume
	Local - Logical Drive 1	O	0

16. Select the "Local - Logical Drive 1" as **Primary Boot Logical Drive/Volume**, and then click **Save**.

A summary window is displayed.

- 17. Click Finish.
- **18.** Exit the ACU by clicking the **X** sign on the top right-hand side of the screen, and then confirm the following dialog:







19. Click **Exit ACU** at the bottom left-hand corner of the screen; the following screen is displayed:



Figure C-13: Power Button

20. Click the **Power** icon in the upper right-hand corner of the screen. The following screen is displayed:

Figure C-14: Reboot Button

21. Click **Reboot** to reboot the server.

The Disk Array configuration is now complete.

22. Install the EMS server installation (see Section 6.2 on page 37).

D Managing Clusters

This appendix describes how to manually migrate or move EMS VMs to another cluster node.

D.1 Migrating EMS Virtual Machines in a VMware Cluster

This section describes how to migrate your EMS Virtual Machine from one ESXi host to another.

To migrate your EMS VM:

1. Select the EMS VM that you wish to migrate and then choose the **Migrate** option:

Navigator	I I 7.2.2123 Actions -
Home 🕨 🕈	D Getting Started Summary Me
	<u>Q</u>
🗣 🕝 qaswvcenter01.corp.audioc	what is a Virtual Machine?
	A virtual machine is a software
🗕 🗊 qaswCluster01	like a physical computer, runs a
10.3.180.211	system and applications. An op system installed on a virtual ma
10.3.180.212	called a guest operating system
7.2.2123	>
🔂 CentOS7-en 🖽 A	Actions - 7.2.2123
EMS 203-7.2	Power >s
High-7.2.205	Guest OS
High217-LyI	Snapshots
🔓 Low-7.2.205 学 🤇	Open Console ii
🔂 SSBC_01 🔤	Migrate
SSBC_02	Clone
SSBC_03	Template •
rCenter ∎	Fault Tolerance
🗗 vEMS	· · · · · · · · · · · · · · · · · · ·
vEMS 7.2.10	VM Policies

Figure D-1: Migration

3. Change a cluster host for migration:

Figure D-2: Change Host



4. Choose the target host for migration:



🗗 7.2.2123 - Migrate		€ €
 1 Select the migration type 2 Select a compute resource 	Select a compute resource Select a cluster, host, vApp or resource pool to run the virtual machines.	
3 Select network	Filter	
4 Select Motion priority	Hosts Clusters Resource Pools vApps	
5 Ready to complete		ier 🔹
	Name 1 Cluster	
	0 📱 10.3.180.211	
	● [] 10.3.180.212 [] qaswCluster01 [] [] []	
	A4	2 Objects
	Compatibility:	
	Compatibility checks succeeded.	
	Back Next	Finish Cancel

Figure D-3: Target Host for Migration

The migration process commences:

Figure D-4: Migration Process Started

Navigator I	10.3.100.211 Actions +						·**
(Hosts and Clusters)	Summary Monitor Manag	e Related Objects					
	10.1.100. Type: Nocesso Logic and You Vision Logic and Some Lightme:	P11 E551 m9 ProLant DL36Qu Gent m9 ProLant DL36Qu Gent metrifly Xeon(R) CPU E5-2 cressors 20 4 chrines 6 Connected 30 days	000 v2 @ 2.600He			090 07820-168 346 07650-6650-08 07660-6650-08 07860-242 18	FREE 20.25 OH CAPACITY, 27.85 OH FREE 20.87 OB CAPACITY, 56.87 OB PREE 3.16 TB CAPACITY, 5.58 TB
තු 5565_01 ශූ 5565_02 ශූ 5565_03 ශූ vEMS ශූ vEMS ශූ vEMS 7.2.1000	Hardware Manufacturer Model CPU Memory CPU Memory Methoring Starsge Starsge	HP ProLiant DL360p Gen8 10 CPUs x 2.76 GHz 70.657 MB / 68.269 MB 0 000 8 / 000 8 localhost.corp audiocodes.com 3 Datastore(s)	Configuration ESX/ESX/Version Image Profile Vobnere HA State Fault Tolerance (Legacy) Fault Tolerance EVC Mode Related Objects Cluster Guster Guster	VMeare ESX, 6.0.0, 3620759 HPE-ESX, 6.0.0-Update2-Iso-600 @ Running (Mater) Unsupported Unsupported Intel® "Sandy Bridge" Generation	95048		
	Update Manager Compli	ance C		More Relate	d Objects		
Recent Tasks Task None Refocule virtual machine	Teget States () 7.2.3123	54 N O VI	ater Duqued Fo Wate	r Start Time 14 ms 10/5/2016 2:25:05 PM	Completion Time	Savar qasavcentari01 corp.audiocodes.com	

After the migration has completed, the EMS application will run seamlessly on the VM on the new cluster's host.

D.2 Moving EMS VMs in a Hyper-V Cluster

This section describes how to move a Virtual Machine to another host node in a Hyper-V cluster.

To move a Virtual Machine to another node of the cluster:

 Select the Virtual Machine, right-click and from the menu, choose Move > Live Migration > Select Node.

File Action View He	le .							
nie Action view rie	1P							_
Failover Cluster Manage	Roles (2)						Actions	
Roles	Search	Today Delawar			Quenes •		Roles	•
Modes	Name	Status	Type	Owner Node	Priority	Informa	Configur.	. ·
Disks Pools Networks Cluster Events	EMS_LC	Connect Start Save Shut Down	ual Mach	ine QAHyperV1	Medi	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	Virtual M. Create E View Refresh Help	
		Turn Off Settings Manage Replication	•				EMS_Low_1 Connect. Start Save	•
		Move	• 😹 G	ive Migration	• 😰	Best Possible N	Node w	m
	< <u>1</u>	Cancel Live Migration	20 候	uick Migration	• 蒙	Select Node		
	v 11. 10	Change Startup Priority	• 🕱 V	irtual Machine Storage	Dwne	rs: Any node	Settings	
	Virtual Mac	Information Details Show Critical Events	-			^	Manage Replicatio	n.►
	8	Add Storage Add Resource	Running 0%	Up Tim	B:	6:1:0	🔝 Move 👬 Cancel Li.	•
		More Actions	4096 ME	8 Availab 8 Integra	le Memory: ion Services	O ME	😒 Change S	
	< ×	Remove				>	🚯 Informati.	
III >	Summary	Properties					Show Cri.	

Figure D-5: Hyper-V Live Migration

The following screen is displayed:

Figure D-6: Move Virtual Machine

from 'QAHyperv2'.		
OOK TOP:		Clear
luster nodes:		-1 <u>2</u> 01
Name	Status	
QAHyperV1	🕑 Up	******

5. Select the relevant node and click **OK**.

The migration process starts.

			Failover Clu	ster Manager				
Action ⊻iew Hel	р							
) 🖄 🖬 🖬 🖬								
ilover Cluster Manage	Roles (2)						Actions	
QAHyperv-Cl.corp.a	Search					🔎 Queries 🔻 🔛 👻 😪	Roles	
Nodes	Name	Status	Туре	Owner Node	Priority	Information	Nonfigur	
🙇 Storage	EMS_High_1	Running	Virtual Machine	QAHyperV1	Medium		Virtual M	
🔠 Disks	EMS_Low_1	Live Migrating	Virtual Machine	QAHyperV1	Medium	Live Migrating, 3% completed	Create E	
Pools							View	
Cluster Events							D D C I	
<u>.</u>							G Refresh	
							Help	
							EMS_Low_1	
							onnect	
							Start	
							Save	
							Shut Down	i.
	<		ш			>	Turn Off	
	783						Sattings	
	Y EMS_Low_1					Preferred Owners: Any node	Secongs	
	~11						invianage	
	Virtual Machine EMS Lo	w 1				Ê	Replication	1
		Status:	Running			=	Move Move	
		CPU Usage:	0%	Up Tim	e:	0:00:06	Cancel Li	0
		Memory Demand:	4096 MB	Availab	le Memory:	0 MB	😼 Change S	
		Assigned Memory:	4096 MB	Integral	tion Services:		🚯 Informati	į.
		Heartbeat:	ок			×	Show Cri	
III >	Summary Resources						Add Stor	

Figure D-7: Hyper-V Migration Process Started

After the migration has completed, the EMS application will run seamlessly on the VM on the new cluster's node.



This page is intentionally left blank.

E Custom X.509 Certificates- Supplementary Procedures

The procedures in this appendix describe supplementary procedures for completing the setup of X.509 Custom certificates.

Note:

- For more information on the implementation of custom certificates, refer to the OVOC Security Guidelines document.
- If you wish to configure the JAWS page to use HTTPS (see Section 19.10.3).

This appendix describes the following procedures:

- Load KeyStore File to PC Client (see below)
- Cleanup Temporary Files on EMS Server (see Section E.2)
- Cleanup Java Web Start Client Files (see Section E.3)
- Update the Java Security Level on PC (see Section E.4)
- Download certificates to the AudioCodes device (see Section E.5).

E.1 Load KeyStore File to PC Client

If you are connecting to the EMS from your PC client, you need to manually copy the KeyStore.jks file to the EMS client installation directory on your PC.

To load the certificate file to the PC client:

- 1. Transfer the file /home/acems/client_certs/KeyStore.jks from the EMS server to the PC that runs the EMS client.
- 1. On the PC that runs EMS client, close EMS client application.
- Navigate to: C:\Program Files (x86)\AudioCodes\EMS Client <Client-Version>\externals\security\ssIDb.
- 3. Rename the existing KeyStore.jks file to KeyStore.jks.ORIG.

Figure E-8: Java KeyStore

> 🐌 → Computer → OSDisk (C:) → Program Fil	es (x86) 🕨 AudioCode	es 🕨 EMS Client 7.0.11	44 ► externals	s ▶ security ▶ sslDb
✓ Include in library ▼ Share with ▼	Burn New folder	r		
Name	Date modified	Туре	Size	
keystore.jks.ORIG	26-Jan-15 16:54	ORIG File	3 KB	
🏂 keystore.p12	04-May-15 14:04	Personal Informati	2 KB	



Note: If you have performed an EMS upgrade, copy the KeyStore file generated in this step to the new EMS client.

- 4. Copy the generated java KeyStore file 'KeyStore.jks' to the following directory: C:\Program Files (x86)\AudioCodes\EMS Client <Client-Version>\externals\security\sslDb
- 5. If you defined a custom (non-default) password, open the file C:\Program Files (x86)\AudioCodes\EMS Client <Client-Version>\externals\configurationProperties\sslConfig.propert ies

and update the following line with the new password:

sslPassword=password

E.2 Cleanup Temporary Files on EMS Server

It is highly recommended to cleanup temporary files on the EMS server after certificates have been successfully installed. This is necessary to prevent access to security-sensitive material (certificates and private keys) by malicious users.

To delete temporary certificate files:

- 1. Login to the EMS server as user root.
- 2. Remove the temporary directories:

```
rm -rf /home/acems/server_certs
```

```
rm -rf /home/acems/client_certs
```

E.3 Cleanup Java Web Start Client Files

This step describes how to update the Java Web Start Client Certificates.



Note: Apply this procedure if you are connecting to the EMS server using Java Web Start (i.e., <u>https://EMS-IP/jaws</u>). Skip this procedure when installing the EMS client using the AudioCodes-supplied Installation DVD.

Do the following:

- 1. Before connecting with JAWS:
 - JAWS for Chrome: Delete directory C:\Users\<pc user>\Downloads\JavaWebStart
 - JAWS for IE: Delete directory C:\Users\<pc user>\AppData\Local\Temp\JavaWebStart.

E.4 Update the Java Security Level on PC

For Java Versions 7 or 8 on your PC, you need to update the Java security level on your PC for your EMS client to function correctly.



Note: This procedure is relevant for both HTTP and HTTPS connections.

Java Version 7:

- Open the Java Control Panel (Start > Program Files > Control Panel > Java).
- 2. Click the **Security** tab, and then set the 'Security Level' to **Medium**.

Figure E-9: Java Control Panel (Version 7.2)

🛃 Java Control Panel							
General Update Java Security Advanced							
Enable Java content in the browser							
Security Level							
- Very High							
- High (minimum recommended)							
Medium							
Exception Site List							
Applications launched from the sites listed below will be allowed to run after the appropriate security prompts.							
Click Edit Site List to add items to this list. Edit Site List							
Restore Security Prompts Manage Certificates							
OK Cancel Apply							

3. Click OK.

Java Version 8:

- Open the Java Control Panel (Start > Program Files > Control Panel > Java).
- 2. Click the **Security** tab and set the Security Level to **High**.

Figure E-10: Java Control Panel (Version 8.0)

<u>د</u>	Java Control Panel 🛛 🗕 🗖 🗙
General Update Java Security Adv	anced
✔ Enable Java content in the browser	
Convrite lovel for poplications pat on th	- Evention Citaliat
Security level for applications not on th	e exception site list
 Very High - Only Java applications allowed to run, and only if the cer 	s identified by a certificate from a trusted authority are tificate can be verified as not revoked.
 High - Java applications identified run, even if the revocation status 	by a certificate from a trusted authority are allowed to of the certificate cannot be verified.
Exception Site List	
Applications launched from the sites	listed below will be allowed to run after the appropriate security
http://10.3.180.2	<u> </u>
http://10.3.180.6	✓ Edit Site List
	Restore Security Prompts Manage Certificates
	OK Cancel Apply
	OK Cancel Apply

3. Click Edit Site List..., and then click Add.

Figure E-11: Exception Site List

	Location		
1	http://10.3.180.10		
1	http://10.3.180.34		
1	http://10.3.180.4		
1	http://10.3.180.7		
1	http://10.3.180.8		
_			
		Add	Domaine

- 4. Enter the server IP address in *http://<server IP>* format.
- 5. Click **OK** to close the Exception Site List window.
- 6. Click **OK** to close the Java Control Panel.

E.5 Installing Custom Certificates on AudioCodes Devices

This section describes how to install Custom certificates on AudioCodes devices. These certificates will be used to secure the connection between the device and EMS / SEM server.

This procedure is performed using the device's embedded Web server. This section describes how to install certificates for the following devices:

- Enterprise gateways and SBC devices (see Section E.5.1).
- MP-1xx devices (see Section E.5.3 on page 253).



Note: When you wish to secure the connection with the device over HTTPS, the certificate loaded to the device must be signed by the same CA as the certificate loaded to the OVOC server.

E.5.1 Single-Sign On to the AudioCodes Device

The Single-Sign On mechanism is used to enable automatic login to the devices embedded Web server tool from the device's status screen in the EMS. When this connection is secured over HTTPS and you wish to replace the certificate, then it can be loaded using the "Certificate File" option in the EMS Software Manager (refer to the *EMS User's Manual*).

When this mechanism is used, the certificate loaded to the EMS must be signed by the same root CA used by the AudioCodes device. In this scenario, the EMS authenticates the device before establishing the connection.

E.5.2 Enterprise Gateways and SBC Devices

This section describes how to install custom certificates on Enterprise gateways and SBC devices.

The device uses TLS Context #0 to communicate with the EMS / SEM server. Therefore, the configuration described below should be performed for **TLS Context** #0.

E.5.2.1 Step 1: Generate a Certificate Signing Request (CSR)

This step describes how to generate a Certificate Signing Request (CSR).

To generate certificate signing request:

- 1. Login to the device's Web server.
- Open the TLS Contexts page (Setup menu > IP Network tab > Security folder > TLS Contexts).
- In the table, select the TLS Context #0, and then click the TLS Context Certificate button, located below the table; the Context Certificates page appears.

Figure E-12: Context Certificates

← TLS Context [#0] > Context Certificates

ERTIFICATE SIGNING REQUEST		
Subject Name [CN]		
Organizational Unit [OU] (optional)	Headquarters	
Company name [O] <i>(optional)</i>	Corporate	
Locality or city name [L] <i>(optional)</i>	Poughkeepsie	
State [ST] <i>(optional)</i>		New York
Country code [C] <i>(optional)</i>		US
Signature Algorithm		SHA-1
	Create CSR	

- 3. Under the Certificate Signing Request group, do the following:
 - a. In the 'Subject Name [CN]' field, enter the device's DNS name, if such exists, or device's IP address
 - **b.** Fill in the rest of the request fields according to your security provider's instructions.
 - c. Click the **Create CSR** button; a textual certificate signing request is displayed in the area below the button:

Å S AudioCodes	SETUP	MONITOR	TROUBLESHOOT	Save	Reset	Actions 🔻	۲ <mark>۲</mark>	Admin 🔻	?		
IP NETWORK SIGNALING & MEDIA	ADM	INISTRATION					р Entit	y, parameter, v	value		
SRD All V											
CORE ENTITIES	•	TLS Context [/	#0] > Context Cert	ificates							
IP Interfaces (1)	CERTIFICATE SIGNING REQUEST										
Ethernet Devices (2) Ethernet Groups (2)	Subject Name [CN]								в		
Physical Ports (2) Static Routes (0)	Or	ganizational Unit [OU] <i>(optional)</i>		Headquarters						
NAT Translation (0)	Co	mpany name [O] <i>(</i>	(optional)		Corporate						
▲ SECURITY	Loc	cality or city name	[L] <i>(optional)</i>		Poughkeepsie						
TLS Contexts (2)	Sta	ate [ST] <i>(optional)</i>			New York US SHA-1						
Firewall (0)	Co	untry code [C] <i>(op</i>	tional)								
Security Settings	Sig	gnature Algorithm									
▶ QUALITY				Create CSR							
RADIUS & LDAP	After crea	ating the CSR, cop	y the text below (includi	ng the BEGIN/END	lines) and se	nd it to your Cer	tification Au	thority for sign	ing.		
> ADVANCED	BE	GIN CERTIFICATI CAQsCAQAwYjEVM	E REQUEST BMGA1UECwwMSGVhZHF1	YXJ0ZXJzMRIwEA	DVQQKDA1D						
DNS •	b3Jwb3JhdGUxFTATBgNVBACMDFBvdMdoa2V1cHNpZTERMA8CA1UECAwITmV3IF1v cmsxCzAJBgNVBAYTA1VTMIGfMA0GCSqGSIb3DQEBAQUAA4GNADCB1QKBgDQDZ+oJ FVGMPLu+sibZ4cn2EAB448Ty2GHROfCN2pRqOdMduO1rkqJUejsFW7BkJUYH3B 3etIvXAssHTmNH+yn4XyId/sO22nBYfUI6rAM65FvL7hi8Z1Ks2Fn401LzOkdhaG IV1gPYjJnLLuce70RC55EdtLetj1Vc86TD20wIDAQABoAAxDQYXKoZIhvcHAQEF DDDTev12020FD8-0-01021-0020-00200-00200-00200-00200-00200-0000-0000-0000-0000-0000-0000-0000-0000										
WEB SERVICES											
HTTP PROXY	V9occI+ 7TXgr65	N/cd/jWdEwjWdM 2s52P7v2eMy+vD	wJxu0yCTRg0IdryXAki Fm80N6C171s780CYMnu	norGGhh8ZqtfdR+ 1AccCrT8Wj8=	XWRLrb07J						
TLS Contexts	EN	D CERTIFICATE I	REQUEST						-		

Figure E-13: Certificate Signing Request Group

4. Copy the text and send it to the certificate authority (CA) to sign this request.

E.5.2.2 Step 2: Receive the New Certificates from the CA

You will receive the following files from the Certificate Authority (CA):

- Your (device) certificate rename this file to "device.crt"
- Root certificate rename this file to "root.crt"
- Intermediate CA certificates (if such files exist) rename these files to "ca1.crt", "ca2.crt" etc.

Save the signed certificate to a file (e.g., device.crt). Make sure that all certificates are in PEM format and appear as follows:

```
----BEGIN CERTIFICATE----
MIIBuTCCASKgAwIBAgIFAKKlMbgwDQYJKoZIhvcNAQEFBQAwFzEVMBMGA1UEAxMM
RU1TIFJPT1QgQ0EyMB4XDTE1MDUwMzA4NTE0MFoXDTI1MDUwMzA4NTE0MFowKjET
...
Tl6vqn51270q/24KbY9q6EK2Yc3K2EAadL2IF1jnb+yvREuewpr0z6TEEuxNJol0
L6V8lzUY0fHrEiq/6g==
```

```
----END CERTIFICATE----
```

AudioCodes



Notes:

- The above files are required in the following steps. Make sure that you obtain these files before proceeding and save them to the desired location.
- Use the exact filenames as mentioned above.

E.5.2.3 Step 3: Update Device with New Certificate

This step describes how to update the device with the new certificate.

> To update device with new certificate:

- Open the TLS Contexts page (Configuration tab > System menu > TLS Contexts).
- In the table, select the TLS Context #0, and then click the TLS Context Certificate button, located below the table; the Context Certificates page appears.
- Under the Upload certificates files from your computer group, click the Browse button corresponding to the 'Send Device Certificate...' field, navigate to the device.crt file, and then click Send File.

Figure E-14: Upload Certificate Files from your Computer Group

UPLOAD CERTIFICATE FILES FROM	YOUR COMPUTER		
Private key pass-phrase <i>(optional)</i>			audc
Send Private Key file from your comp The file must be in either PEM or PFX (F	uter to the device. PKCS#12) format.		
Browse No file selected.	Send File		
Note: Replacing the private key is	not recommende	d but if it's don	e it should be over a physically-secure network link

lote: Replacing the private key is not recommended but if it's done, it should be over a physically-secure network lin

Send Device Certificate file from your computer to the device. The file must be in textual PEM format.

 Browse...
 No file selected.
 Send File

E.5.2.4 Step 4: Update Device's Trusted Certificate Store

This step describes how to update the device's Trusted Certificate Store.

> To update device's trusted certificate store:

- Open the TLS Contexts page (Configuration tab > System menu > TLS Contexts).
- In the table, select the TLS Context #0, and then click the TLS Context Trusted Root Certificates button, located below the table; the Trusted Certificates page appears.
- 3. Click the **Import** button, and then browse to the root.crt file. Click **OK** to import the root certificate.

Figure E-15: Importing Certificate into Trusted Certificates Store

Import New Certificate	×
Browse No file selected.	
	Ok Cancel

4. If you received intermediary CA certificates – ca1.crt, ca2.crt, etc. – import them in a similar way.

E.5.2.5 Step 5: Configure HTTPS Parameters on the Device

This section describes how to configure HTTPS related parameters on the device.

Note:

- You can optionally pre-stage the device with a pre-loaded ini file including this configuration (for more information, contact your AudioCodes representative).
- If you have enabled the Interoperability Automatic Provisioning feature, ensure that your template file is also configured as described in this procedure to maintain an active HTTPS connection after the template file has been loaded to the device.
 - When you setup an HTTPS connection on the device, you must also enable HTTPS ("Enable HTTPS Connection") when adding the device to the EMS (refer to the EMS User's manual).

To configure HTTPS parameters on the device:

- 1. Create a new text file using a text-based editor (e.g., Notepad).
- 2. Include the following ini file parameters for server-side authentication:
 - For Media Gateway and SBC devices:

AUPDVerifyCertificates=1

- For MP-1xx devices:
 - The ini file should include the following two lines:

AUPDVerifyCertificates=1

ServerRespondTimeout=10000

 When working with SEM TLS (see Section 19.10.5), add the following parameter.

QOEENABLETLS=1

- **3.** Save and close the file.
- Load the generated file as "Incremental INI file" (Maintenance menu > Software Update > Load Auxiliary Files > INI file (incremental).
- Open the TLS Contexts page (Setup menu > IP Network tab > Security folder > TLS Contexts).
- 6. In the table, select the **TLS Context #0**, and then click **Edit** button. The following screen is displayed:

Figure E-16: TLS Contexts: Edit Record

TLS Contexts [default]						– x				
GENERAL			OCSP							
Index	0		OCSP Server		Disable	•				
Name	• default		Primary OCSP Server							
TLS Version	Any - Including SSLv3	•	Secondary OCSP Server	۰						
DTLS Version	Any	•	OCSP Port		2560					
Cipher Server	RC4:AES128		OCSP Default Response		Reject	•				
Cipher Client	RC4:DEFAULT									
Strict Certificate Extension Validation	Disable	•								
DH key Size	1024	•								
		_	_							
	Cancel APPLY									

- 7. Set 'TLS Version' to 1 (TLS 1.0 only).
- 8. Set 'HTTPS Cipher Server' to ALL.
- 9. Set 'HTTPS Cipher Client' to ALL.

E.5.2.6 Step 6: Reset Device to Apply the New Configuration

This step describes how to reset the device to apply the new configuration.

To reset the device:

 In the top-level menu, click **Device Actions > Reset**. The following screen is displayed.

	SETUP	MONITOR	TROUBLES	ноот				
IP NETWORK SIGNALING & ME	DIA ADMINIST	RATION	Save	Reset	Actions 🔻	47	Admin 🔻	?
SRD All	 Entity, paramet 	ter, value						
☆ TIME & DATE WEB & CLI	Maintenance A	ctions						
▶ SNMP	RESET DEVICE			LC	DCK/UNLOCK			
▲ MAINTENANCE	Reset Device		Rese	Loc	ck		LOC	
Configuration File Auxiliary Files Maintenance Actions License Key Software Upgrade	Save To Flash Graceful Option For Reset Device all changes made sin For Save Configur in voice quality, then	If you choose nce the last tin ration: Saving efore, it is reco	e not to save the ne the configura g configuration t ommended to p	device's con tition was sav o flash mem erform this o	aceful Option teway Operation nfiguration to fla ved will be lost a nory may cause during low-traffi	nal State ash memor after the de some temp c periods	IO UNLOCKED y, vice is reset. orary degrada	tion

Figure E-17: Device Reset

2. From the Burn to FLASH drop-down list, select **Yes**, and then click **Reset** button. The device will save the new configuration to non-volatile memory and reset itself.
E.5.3 MP-1xx Devices

This section describes how to install Custom certificates on the MP 1xx devices.

E.5.3.1 Step 1: Generate a Certificate Signing Request (CSR)

This step describes how to generate a Certificate Signing Request (CSR).

To generate a CSR:

- 1. Your network administrator should allocate a unique DNS name for the device (e.g., dns_name.corp.customer.com). This DNS name is used to access the device and therefore, must be listed in the server certificate.
- 2. If the device is operating in HTTPS mode, then set the 'Secured Web Connection (HTTPS)' parameter (HTTPSOnly) to HTTP and HTTPS (refer to the *MP-11x and MP-124 User's Manual*). This ensures that you have a method for accessing the device in case the new certificate does not work. Restore the previous setting after testing the configuration.
- **3.** Login to the MP-1xx Web server.
- 4. Open the Certificates page (Configuration tab > System menu > Certificates).
- 5. Under the **Certificate Signing Request** group, do the following:
 - **a.** In the 'Subject Name [CN]' field, enter the DNS name.
 - **b.** Fill in the rest of the request fields according to your security provider's instructions.
 - c. Click the Create CSR button; a textual certificate signing request is displayed in the area below the button:

Figure E-18: Certificate Signing Request Group

 Certificate Signing Request 	
Subject Name [CN]	audio.com
Organizational Unit [OU] (optional)	Headquarters
Company name [O] (optional)	Corporate
Locality or city name [L] (optional)	Poughkeepsie
State [ST] (optional)	New York
Country code [C] (optional)	US
Certification Authonty for Signing.	
Certification Authority for signing.	······································
BEGIN CERTIFICATE REQUEST MIIBEJCCAR&CAQAwdjESMBAGA\UEAXMJYXVkai cXVhenRlenMxEjAQBgNVBAOTCUNvenBvemP021 c2llMREwDwVDVQQTEwh02XcgWW9yazELMAkGA AQEBBQADgY0AMIGJAOGBAPHpf2t4OLy3FRk5Bv xL70f8Y0L63eeIK2eDo8nmfrJ0f77z/AHWJmFV LLNUnoqjEsK7CF32uv0H//gFkhy5zleNv0bI4 gMBAAGgADANBgkqhkiG9w0BAQQFAA0BgQDih q7FGjFXAf2k1MmgnBMc/MYfSGTbawrQF7p6dNa	<pre>%8uy29tMRUwEwYDVQQLEwxIZWFk EVMBMGA1UEBxMMUG9122hrZWVw 1UEBhMCVVMwg28wDQYJKoZIhvcN %FlZFWCXQ7nvuocHtu7Nns071M 55paAt1cboIFgoZNS0g6+5JAmJAA +25pn38aJzEXc8DkGwz19rRogRZ igbc1zkAtLF+5FRuSekygUXEM6 f00ivmuCPf6gz25m2uqc6LqoIi</pre>
BEGIN CERTIFICATE REQUEST MIIBİJCCAR8CAQAWdjESMBAGAİUEANJYXVkai cXYhcnRlcnMxEjAQBgNVBAOTCUNvcnBvcmF0Z: c211MRewDwYDVQQTEwhOZXcgWW9yazELMAKGAİ AQEBBQADgY0AMIGJAoGBAPHpf2t4OLy3FRk5Bs x170f8y0L63eeIK2eDo8nn6rJ06772/JAHWJMEU LINUnoqjEsK7CF32uvolH//gFkhy5zleNv0bFi AgMBAAGADANBGKqhkiGW0BAQQFAAOBgQDih0 q7FGjFXAfzk1MmgnBMc/MYfSGTbawrQF7p6dN. nLnQpVCmbdva/BlQyEPbQAQULJ8CSe3rrY3 ZvnZ2e5M5CB3La=	<pre>w8uy29tMRUwEwYDVQQLEwxIZWFk TEVMBMGA1UEBxMMUG9122hr2WVW UEBhMCVMwg28wDQyJKoZ1hvcN w7F12FWCXQ7nvuoethu7Nns071M 55paR1cboIPg0ZN80g6+5JAmJAA +25pn38aJzEXc8DkGwZ19rR0qRZ igbc1zkHdLr+55RuScKygUXBM6 J60DivmuCPf6Gzz5m2uqC6Lq0Ii 5ru23AZeDUDYyh0901kRbAp//+3</pre>

6. Copy the text and send it to the certificate authority (CA) to sign this request.

E.5.3.2 Step 2: Receive the New Certificates from the CA

You will receive the following files from the Certificate Authority (CA):

- Your (device) certificate rename this file to "device.crt"
- Root certificate rename this file to "root.crt"
- Intermediate CA certificates (if such files exist) rename these files to "ca1.crt", "ca2.crt" etc.

Save the signed certificate to a file (e.g., device.crt). Make sure that all certificates are in PEM format and appear as follows:

----BEGIN CERTIFICATE-----

MIIDkzCCAnugAwIBAgIEAgAAADANBgkqhkiG9w0BAQQFADA/MQswCQYDVQQGEw JGUjETMBEGA1UEChMKQ2VydG1wb3N0ZTEbMBkGA1UEAxMSQ2VydG1wb3N0ZSBT ZXJ2ZXVyMB4XDTk4MDYyNDA4MDAwMFoXDTE4MDYyNDA4MDAwMFowPzELMAkGA1 UEBhMCR1IxEzARBgNVBAoTCkN1cnRpcG9zdGUxGzAZBgNVBAMTEkN1cnRpcG9z dGUgU2VydmV1cjCCASEwDQYJKoZIhvcNAQEBBQADggEOADCCAQkCggEAPqd4Mz iR4spWldGRx8bQrhZkonWnNm`+Yhb7+4Q67ecf1janH7GcN/SXsfx7jJpreWUL f7v7Cvpr4R7qIJcmdHIntmf7JPM5n6cDBv17uSW63er7NkVnMFHwK1QaGFLMyb FkzaeGrvFm4k3lRefiXDmuOe+FhJgHYezYHf44LvPRPwhSrzi9+Aq3o8pWDguJ uZDIUP1F1jMa+LPwvREXfFcUW+w==

----END CERTIFICATE-----



Notes:

- The above files are required in the following steps. Make sure that you obtain these files before proceeding.
- Use the exact filenames as mentioned above.

E.5.3.3 Step 3: Update Device with New Certificate

This step describes how to update the device with the new certificate.

To update the device with the new certificate:

- In the Certificates page, scroll down to the Upload certificates files from your computer group, click the Browse button corresponding to the 'Send Device Certificate...' field, navigate to the device.crt file, and then click Send File.
- 2. After the certificate successfully loads to the device, save the configuration with a device reset (see Section E.5.3.6 below).

E.5.3.4 Step 4: Update Device's Trusted Certificate Store

For the device to trust a whole chain of certificates you need to combine the contents of the root.crt and ca.crt certificates into a single text file (using a text editor).

To update the device with the new certificate:

- 1. Open the root.crt file (using a text-based editor, e.g., Notepad).
- 2. Open the ca.crt file (using a text-based editor, e.g., Notepad).
- **3.** Copy the content of the ca.crt file and paste it into the root.crt file above the existing content.

Below is an example of two certificate files combined (the file "ca2.crt" and the "root.crt") where the ca2.crt file contents are pasted above the root.crt file contents:

----BEGIN CERTIFICATE-----

MIIDNjCCAh6gAwIBAgIBBDANBgkqhkiG9w0BAQUFADAhMQwwCgYDVQQKEwNBQ0wx ETAPBqNVBAMUCEVNU19ST09UMB4XDTEwMDEwMTAwMDAwMFoXDTIwMDEwMTAwMDAw MFowIDEMMAoGA1UEChMDQUNMMRAwDgYDVQQDFAdFTVNfQ0EyMIIBIjANBgkqhkiG 9w0BAQEFAAOCAQ8AMIIBCgKCAQEA4CmsdZNpWo6Gg5UgxflPjJeNggwnlQiUYhOK kPEvS6yWH7tr8+TwnIzjT58kuuy+fFVLDyZzp117J53FIsgnCSxpVqcYfMoBbCL/ 0fmXKHWlPIIbovWpZddqz8U1pEzD+5eGMUwCnqw99rbUseAHdwkxsXtOquwqE4yk ihiWesMp54LwX5dUB46GWKUfT/pdQYqAuunM76ttLpUBc6yFYeqpLqj90gKkR4cu 5B6wYNPoTjJX50Xgd9Yf+0IQYB2EiP06uzLtlyWL3AENGwDVeOvlfZgppLEZPBKI hfULeMjay4fzE4XnS9LDxZGjJ+nV9ojA7WaRB5tl6nEJQ/7sLQIDAQABo3oweDAM BgNVHRMEBTADAQH/MB0GA1UdDgQWBBRy2JQ1yZrvN4GifsXUB7AvctWvrTBJBgNV HSMEQjBAgBThf6GbMQbO5b0CkLV8kW+Rg0AAhqElpCMwITEMMAoGA1UEChMDQUNM $\label{eq:mrewdwydvqdfahftvnfuk9pviibatanbgkqhkiG9w0BAQUFAAOCAQEAdasYyfcg$ TdkF/uDxlOGk0ygXrRAXHG2WFOS6afrcJHoZCCH3PNsvftRrEAwroGwx7tsn1/o+ CNV5YalstIz7BDIEIjTzCDrp09sUsiHqxGuOnNhjLDUoLre1GDC00yiKb4BOhlCq hiemkXRe+eN7xcg0IfUo78VLTPuFMUhz0Bdn7TuE7QbiSayq2fY2ktHHOyDEKJGO RUosIqgVwSZIsCnRZFumkKJtrT4PtnNYluYJHej/SHcsOWtgtCQ8cPdNJCZAWZ+V XoAhN6pH17PMXLPclm9L/MlkVkmf0tp1bPmefrEBl0+np/08F+P551uH0i0YA6Cc Cj6oHGLq8RIndA==

----END CERTIFICATE----

----BEGIN CERTIFICATE----

MIIDNzCCAh+gAwIBAgIBATANBgkqhkiG9w0BAQUFADAhMQwwCgYDVQQKEwNBQ0wx ETAPBgNVBAMUCEVNU19ST09UMB4XDTEwMDEwMTAwMDAwMFoXDTIwMDEwMTAwMDAw MFowITEMMAoGA1UEChMDQUNMMREwDwYDVQQDFAhFTVNfUk9PVDCCASIwDQYJKoZI hvcNAQEBBQADggEPADCCAQoCggEBANCsaGivTMMcSv57+j5Hya3t6A6FSFhnUQrS 667hVpbQ1Eaj02jaMh8hNv9x8SFDT52hvgVXMnLBmpZwy+To1VR4kqbAEoIs+7/q ebESJyW8pTLTszGQns2E214+U18sKHItpUZvs1dVUIX6xQiSYFDG1CDIPR5/70pq zwtdbIipSsKgYijos0yRV3roVqNi4e+hmLVZA9rOIp6LR72Ta9HMJFJ4gyxJPUQA jV3Led2Y4JObvBTN1ka18WI7KORJigMMp7T8ewRkBQ1JM7nmeGDPUf1wRjDWg14G BRw2MACYsu/M9z/H821UOICtsZ4oKUJMqbwjQ91XI/HQkKRSTf8CAwEAAaN6MHgw DAYDVR0TBAUwAwEB/zAdBgNVHQ4EFgQU4X+hmzEGzuW9ApC1fJFvkYNAAIYwSQYD VR0jBEIwQIAU4X+hmzEGzuW9ApC1fJFvkYNAAIahJaQjMCExDDAKBgNVBAoTA0FD TDERMA8GA1UEAxQIRU1TX1JPT1SCAQEwDQYJKoZIhvcNAQEFBQADggEBAHqkg4F6 wYiHMAjjH3bqxUPHt2rrrALaXA9eYWFCz1q4QVpQNYAwdBdEAKENznZttoP3aPZE 3EOx1C8Mw2wU4pOxD7B6pH0XO+oJ4LrxLB3SAJd5hW495X1RDF99BBA9eGUZ2nXJ 9pin4PWbnfc8eppq8Tp18jJMW0Z13prfPt012q93iEalkDEZX+wxkHGZEqS4ayBn

AudioCodes

8bU3NHt5qh0Egpai8hB/nth1xnA1m841wxCbJW86AMRs2NznROyG695InAYaNlIo HU9zBRdRRASV5vmBN/q5JnDhshZhL1Bm+M6QxOyGoNjL1DqE+aWZkmsw2k9STOpN itSUgGYwEagnsMU=

----END CERTIFICATE----



Notes: The maximum supported size of the combined file of trusted chain of certificates is 100,000 bytes (including the certificate's headers).

- 4. Save the combined content to a file named "chain.pem" and close the file.
- 5. Open the Certificates page and upload chain.pem file using the 'Trusted Root Certificate Store' field.

E.5.3.5 Step 5: Configure HTTPS Parameters on Device

Configure HTTPS Parameters on the device (see Section E.5.2.5 above).

E.5.3.6 Step 6: Reset Device to Apply the New Configuration

This section describes how to apply the new configuration.

To save the changes and reset the device:

- **1.** Do one of the following:
 - On the toolbar, click the **Device Actions** button, and then from the drop-down menu, choose **Reset**.
 - On the Navigation bar, click the **Maintenance** tab, and then in the Navigation tree, select the **Maintenance** menu and choose **Maintenance Actions**.

Reset Board	Reset
Burn To FLASH	Yes 💌
Graceful Option	No 💌
 LOCK / UNLOCK 	
Lock	LOCK
Graceful Option	No 💌
Current Admin State	UNLOCKED
Burn To FLASH	BURN

Figure E-19: Maintenance Actions Page

- 2. Click the **Reset** button; a confirmation message box appears, requesting you to confirm.
- Click OK to confirm device reset; when the device begins to reset, a notification message is displayed.



This page is intentionally left blank.

F Transferring Files

This appendix describes how to transfer files to and from the EMS server using any SFTP/SCP file transfer application.



Note: .FTP by default is disabled in the EMS server.

To transfer files to and from the EMS server:

- 1. Open your SFTP/SCP application, such as WinSCP or FileZilla.
- 2. Login with the acems/acems credential (all files transferred to the EMS server host machine are then by default saved to /home/acems directory).
- 3. Copy the relevant file(s) from your PC to the host machine (or vice-versa). For example using the FileZilla program, you drag the relevant file from the left pane i.e. in your PC directory to the right pane i.e. the /home/acems directory on the EMS server host machine.



This page is intentionally left blank.

G Verifying and Converting Certificates

This appendix describes how to verify that certificates are in PEM format and describes how to convert them from DER to PEM if necessary.

To verify and convert certificates:

- 1. Login to the EMS server as user root.
- 2. Transfer the generated certificate to the EMS server.
- **3.** Execute the following command on the same directory that you transfer the certificate to verify that the certificate file is in PEM format:

```
Openssl x509 -in certfilename.crt -text -noout
```

- 4. Do one of the following:
 - **a.** If the certificate is displayed in text format, then this implies that the file is in PEM format, and therefore you can skip the steps below.
 - **b.** If you receive an error similar to the one displayed below, this implies that you are trying to view a DER encoded certificate and therefore need to convert it to the PEM format.

```
unable to load certificate
12626:error:0906D06C:PEM routines:PEM_read_bio:no start
line:pem_lib.c:647:Expecting: TRUSTED CERTIFICATE
```

5. Convert the DER certificate to PEM format:

```
openssl x509 -inform der -in certfilename.crt -out certfilename.crt
```



This page is intentionally left blank.

H Self-Signed Certificates

When using self-signed certificates, use the following instructions for recognizing the secure connection with the EMS server from your IP Phone Manager and SEM Web client browsers.

H.1 Internet Explorer

When the following screen is displayed, select the "Continue to website (not recommended)" option.

		- • •
← 🕀	t͡3 https://10.1.8.23:9400/EMS-VQ/Main.html# ♀ ♂ ♂ ♂ t͡3 Certificate Error: Navigatio ×	☆ 🕸
File Edit	t View Favorites Tools Help	
8	There is a problem with this website's security certificate.	
	The security certificate presented by this website was not issued by a trusted certificate authority. The security certificate presented by this website was issued for a different website's address.	
	Security certificate problems may indicate an attempt to fool you or intercept any data you send to the server.	
	We recommend that you close this webpage and do not continue to this website.	
	Click here to close this webpage.	
	Sontinue to this website (not recommended).	
	More information	

Figure H-1: Continue to Website

H.2 Using Mozilla Firefox

Do the following:

- 1. When the following screen is displayed, click the "I Understand the Risks" option.
- 2. Click the Add Exception button, and then click the Confirm Security Exception button.

Figure H-2: Mozilla Firefox Settings

🚹 Untrusted Connection 🛛 🗙 🕂						
A https://10.4.2.60:9400/EMS-VQ/Main.html#		$\forall C' $ $\bigcirc Q_s$ Search	☆ 自	ŧ	^	9
	This Connection is Untrusted You have asked Firefox to connect securely to 10.4.2.60 connection is secure. Normally, when you try to connect securely, sites will p are going to the right place. However, this site's identity What Should I Do? If you usually connect to this site without problems, th impersonate the site, and you shouldn't continue.	2:9400, but we can't confirm that your resent trusted identification to prove that you can't be verified. Add Security Exception You are about to override how Firefox identifi Legitimate banks, stores, and other public	fies this site. sites will not a	sk you	to do t	his.
	Get me out of here! Technical Details I Understand the Risks If you understand what's going on, you can tell Firefox you trust the site, this error could mean that someor Don't add an exception unless you know there's a good identification. Add Exception.	Server Location: https://10.4.2.60.9400/EMS-VQ/Main.htm Certificate Status This site attempts to identify itself with invalid inform Wrong Site The certificate belongs to a different site, which coul impersonate this site. Unknown Identity The certificate is not trusted because it hasn't been v authority using a secure signature.	nt= nation. Id mean that so rerified as issued	Ge omeone d by a t	<u>V</u> iew is tryin	g to
		Permanently store this exception <u>Confirm</u>	Security Except	ion	Can	cel

H.3 Chrome

When the following screen is displayed, click **Advanced** and then click the "Proceed to <Server IP> (unsafe)" link.



Figure H-3: Chrome Browser Settings

International Headquarters

1 Hayarden Street, Airport City Lod 7019900, Israel Tel: +972-3-976-4000 Fax: +972-3-976-4040

AudioCodes Inc.

27 World's Fair Drive, Somerset, NJ 08873 Tel: +1-732-469-0880 Fax: +1-732-469-2298

Contact us: <u>www.audiocodes.com/contact</u> Website: <u>www.audiocodes.com</u>

©2017 AudioCodes Ltd. All rights reserved. AudioCodes, AC, HD VoIP, HD VoIP Sounds Better, IPmedia, Mediant, MediaPack, What's Inside Matters, OSN, SmartTAP, User Management Pack, VMAS, VoIPerfect, VoIPerfectHD, Your Gateway To VoIP, 3GX, VocaNom, AudioCodes One Voice and CloudBond are trademarks or registered trademarks of AudioCodes Limited. All other products or trademarks are property of their respective owners. Product specifications are subject to change without notice.

Document #: LTRT-94152



